Overconfidence in Elimination Contests

Yuxi Chen^{*} Luis Santos-Pinto[†]

March 13, 2024

Abstract

This paper analyzes how overconfidence affects behavior in elimination contests. We focus on two-stage elimination contests with four players. In the first-stage, the players are matched pairwise, and each pair competes in one semifinal. In the second-stage, the first-stage winners compete in the final. An overconfident player overestimates his winning probability in each pairwise interaction. Our findings reveal a nuanced interplay between overconfidence and effort exertion. An overconfident player expends less effort in the final stage than a rational rival. However, this pattern can be inverted in the semifinal, where an overconfident player can exert more effort than a rational rival. We also uncover that an overconfident player can have the highest equilibrium probability of winning an elimination contest. Our results offer a novel perspective on the promotion of overconfident individuals to CEO positions. They also highlight that large increases in executive compensation can render the pursuit of CEO positions exceptionally appealing to overconfident individuals.

Keywords: Overconfidence, Elimination Contest, Encouragement Effect, Complacency Effect.

^{*}Department of Economics, University of Lausanne, Quartier UNIL-Dorigny, 1015 Lausanne, Switzerland. Email: Yuxi.Chen@unil.ch.

[†]Department of Economics, University of Lausanne, Quartier UNIL-Dorigny, 1015 Lausanne, Switzerland. Email: luispedro.santospinto@unil.ch.

1 Introduction

Elimination contests are a common feature in politics, organizations, sports, and academia. In politics, politicians compete for top positions in the party, and those who reach prominent positions in the party compete to become high-level government officials. In organizations, employees compete for promotion to manager, and managers compete for promotion to a chief executive officer position. In academia, PhDs compete to be hired as assistant professors, and assistant professors compete for tenure. In tennis and many other sports, players compete in elimination contests.

Overconfidence is one of the most widely documented biases in judgment and has been detected both in the laboratory and in the field.¹ Overconfidence has consequences for economic behavior in labor markets (Spinnewijn 2013, Spinnewijn 2015, Köszegi 2014, Santos-Pinto and de la Rosa 2020). A large proportion of CEOs is overconfident and CEO overconfidence affects corporate decisions (Malmendier and Tate, 2005, 2008, 2015). It remains an open research question why do these overconfident CEOs obtain their jobs in the first place.

In this paper we analyze how overconfidence, conceptualized as overestimation of ability, affects behavior in elimination contests. We are interested in finding answers to the following questions. How does overconfidence affect effort provision in the different stages of an elimination contest? Is an overconfident player more or less likely to win an elimination contest than a rational player? What are the welfare implications of overconfidence for the players and for the contest designer? How does overconfidence alter the optimal prize structure chosen by the contest designer?

To address these questions we consider a two-stage elimination contest with four players. In the first stage, the players are matched pairwise, and each pair competes in one semifinal. The first-stage winners go on to the second stage of the contest and compete against each other in the final. The winner receives prize w_1 , the runner-up prize w_2 , and the first-stage losers receive nothing, with $w_1 > w_2 \ge 0$.

In each pairwise interaction the players choose their efforts simultaneously to maximize their expected utilities. The probability of winning a pairwise interaction depends on the efforts of both players through Alcalde and Dahm's (2007) contest success function (CSF).² Player are homogeneous, except for their confidence levels. This allow us to zero in on the impact of overconfidence on players' incentives to exert effort. Finally, we assume an overconfident player has a correct perception of the prizes and cost of effort but overestimates the impact of his effort on his winning probability in each pairwise interaction. Furthermore, an overconfident player's bias is observable by his rivals.

Section 3 analyzes an elimination contest with four rational players. Proposition 1 characterizes its equilibrium which serves as a benchmark to which we compare all our results. When the four players are rational the elimination contest is symmetric and hence each player has 1/4 probability of being the winner.

Section 4 considers an elimination contest with one overconfident player and three rational players. This configuration describes scenarios where a minority of players is overconfident and allows us to study the effects of overconfidence in the simplest possible way. Proposition 2 demonstrates that an overconfident player always exerts less effort in a final than a rational rival. Intuitively, the (mis)perceived advantage of the overconfident

 $^{^{1}}$ Moore and Healy (2008) distinguish between three types of overconfidence: overestimation of one's skill (absolute overconfidence), overplacement (relative overconfidence), and excessive precision in one's beliefs (miscalibration or overprecision).

 $^{^{2}}$ This CSF allows us to derive closed form solutions for the players' efforts, winning probabilities, continuation values, and expected utilities at each stage of the contest. Section 2 explains Alcalde and Dahm's (2007) CSF in detail. Section 6 shows that our main results extend to Tullock's (1980) CSF.

player leads him to lower his effort. The rational player, anticipating the overconfident player will lower his effort, also lowers her effort but not as much. Hence, both players exert less effort in the final than if both were rational. Proposition 2 also shows that both players' perceived expected utility of the final increases in the bias of the overconfident player. Hence, the bias makes reaching the final more attractive for an overconfident player and for a rational rival.

Proposition 3 shows that an overconfident player can exert higher effort in a semifinal than a rational rival. In a semifinal, a player chooses the level of effort at which the perceived marginal benefit equals the marginal cost. The perceived marginal benefit is the product of the perceived marginal probability of winning the semifinal and the perceived continuation value (a player's perceived expected utility of the final before knowing the identity of his rival in the final). As Rosen (1986) first hinted at, overconfidence has two effects on a player's incentives to exert effort in a semifinal. On the one hand, the bias lowers the perceived marginal probability of winning the semifinal which motivates an overconfident player to lower effort. We label this the complacency effect of overconfidence. On the other hand, the bias raises the perceived continuation value which motivates an overconfident player to raise effort. We label this the encouragement effect of overconfidence. When the encouragement (complacency) effect dominates, the overconfident player exerts higher (lower) effort at equilibrium in the semifinal. Proposition 3 also shows that two conditions have to be met for the encouragement effect to dominate. First, the prize spread needs to be sufficiently large. Second, the overconfident player's bias cannot be too high. Propositions 2 and 3 generate the model's main testable prediction: an overconfident player always exerts less effort in a final than a rational rival but can exert higher effort in a semifinal than a rational rival.

Proposition 4 reveals that the presence of an overconfident player in one of the semifinals of an elimination contest has a spillover effect on the efforts in the other semifinal. In the semifinal with two rational players, each player has an incentive to exert higher effort than if all players in the contest were rational. This happens because the winner of this semifinal will face an overconfident rival in the final with positive probability. This raises the rational players' continuation value due to the higher expected utility of the final.

Proposition 5 shows that the overconfident player can be the one with the highest equilibrium probability of winning the elimination contest. Three conditions have to be met for this to be the case. First, the role that effort plays in determining the winning probabilities must be sufficiently high. Second, the prize spread needs to be sufficiently large. Third, the overconfident player's bias needs to be small.

Section 5 studies an elimination contest with two overconfident players and two rational players. We assume the overconfident players have different biases and consider two types of seedings: (i) the overconfident players are seeded in the same semifinal, and (ii) the overconfident players are seeded in different semifinals. Proposition 6 demonstrates that in a final featuring two overconfident players, the more confident player always exerts lower effort, and both players exert lower efforts than if both were rational. Proposition 7 reveals that when two overconfident players are seeded in the same semifinal, both can exert higher efforts than if both were rational. Proposition 8 shows that when two overconfident players are seeded in different semifinals, both can exert higher efforts than if both were rational. Proposition 8 shows that when two overconfident players are seeded in different semifinals, both can exert higher efforts than if both were rational. Proposition 8 shows that when two overconfident players are seeded in different semifinals, both can exert higher efforts than if both were rational players efforts than their rational rivals. Propositions 6, 7, and 8 collectively underscore that the findings derived for an elimination contest with one overconfident and two rational players, irrespective of the seeding.

Section 6 analyzes the welfare implications of overconfidence for the players and the

contest designer. To do that we consider an elimination contest with one overconfident player and three rational players. Proposition 9 shows the overconfident player can gain from his bias. The rational player seeded in the same semifinal as the overconfident player losses (gains) when the overconfident player exerts higher (lower) effort. The rational players seeded in the same semifinal always gain. Proposition 10 shows that while overconfidence always lowers aggregate effort in the final stage it can either raise or lower aggregate effort in the semifinals stage. This result implies that we are unable to state general conditions under which overconfidence lowers or raises the contest designer's welfare measured as the players' aggregate effort in the two stages.

Section 7 considers how the contest designer should set prizes when players are overconfident and her goal is to maximize players' aggregate efforts in the two stages. To answer this question we assume players are homogeneous and the contest designer has a fixed prize budget she can allocate to either the winner or the runner-up. Proposition 11 shows that the contest designer chooses a winner-take-all prize structure when players are risk neutral. In contrast, when players are risk-averse, the optimal prize structure involves multiple prizes with the winner receiving most of the prize money and a smaller part being assigned to the runner-up. Finally, Proposition 11 shows that an increase in players' overconfidence, leads the contest designer to allocate an increasingly higher share of the prize budget to the winner.

Section 8 discusses three extensions of the model which demonstrate the robustness of our findings. It starts by showing that our main results extend to a two-stage elimination Tullock contest. Subsequently, it shows our main results also hold when the overconfident player's bias is not observable by the rational players. Next, it shows our main results extend to a three-stage elimination contest with eight players. Finally, it discusses the impact of underconfidence on a two-stage elimination contest with one underconfident and three rational players. It uncovers that the underconfident player exerts less effort than his rational rivals during the final and the semifinal stages of an elimination contest. Hence, the underconfident player is the one with the lowest equilibrium probability of winning the elimination contest.

Our study relates to four strands of literature. First, it contributes to the literature on CEO overconfidence. Empirical evidence documents that a substantial share of CEOs are overconfident (for a review see Malmendier and Tate, 2015). The seminal contribution to this literature is Malmendier and Tate (2005, 2008) who measure CEO overconfidence as the tendency to hold stock options longer before exercise. Malmendier and Tate (2015) use this measure together with additional controls and find that approximately 40 percent of CEOs of companies listed in the Standard & Poor's 1500 index are overconfident.

The literature on CEO overconfidence offers two main explanations for why overconfident managers are promoted to CEO positions. Goel and Thakor (2008) study tournaments where risk-averse managers compete for promotion to become CEO by choosing the level of risk of their projects. Some managers are rational while others are overconfident. An overconfident manager underestimates project risk which increases the propensity to take risky projects (e.g. R&D activities). Some of the more risky projects will be successful and hence, the higher risk taking of overconfident managers will improve their chances of promotion to CEO. According to Van den Steen (2005), CEO overconfidence can serve as a commitment device that helps attract and retain employees that share the same values as the CEO.

Our results provide a new explanation for why overconfident managers are promoted to CEO positions. One can think of reaching a CEO position as winning a series of labor promotions in which managers compete based on performance to move up the corporate ladder. Our model predicts that if the prize spread over the corporate ladder is large, then moderately overconfident managers can be more likely to be promoted to a CEO position than rational ones. This happens because moderately overconfident managers will exert more effort than rational ones (e.g., by working longer hours) in the early stages of their careers due to the encouragement effect of overconfidence. The larger the prize spread between the compensation of a low level manager and that of a CEO, the greater the encouragement effect. Hence, our results highlight the role that large increases in executive compensation (Murphy 2013) can have in rendering the pursuit of a CEO position extremely attractive to overconfident individuals.

Second, our study contributes to the large literature on gender gaps in the labor market. Empirical evidence documents gender gaps in wages and in top business positions. For instance, in 2022 women in the US earn 82 percent of their male counterparts (Kochhar 2023) and women represent only 6 percent of top business executives in the US (Keller et al. 2022). The wage gender gap is larger in high skilled work, and much of it seems to be caused by gaps in promotions (Blau and DeVaro 2007, Blau and Kahn 2017, Bronson and Thoursie 2020). Laboratory experiments show that gender differences in confidence and risk attitudes can account for gender gaps in behavior in tournaments and contests (Niederle and Vesterlund 2007, Kamas and Preston 2012, Gillen et al. 2019, Price 2020, Buser et al. 2021, van Veldhuizen 2022).

Our findings show that the large executive compensation spreads coupled with higher male confidence can make competing for a top business position much more attractive for male candidates. We also predict that much of the gender gap in promotions will take place early in workers' careers. This could place women at a further disadvantage besides the negative effects of childbirth and child-rearing (Bertrand et al. 2010, Goldin and Katz 2011, Goldin 2014).³

Third, our study also contributes to the theoretical literature on overconfidence, tournaments, and contests. Santos-Pinto (2010) shows how firms can optimally set tournament prizes to exploit workers' overconfidence, defined as overestimation of ability. Ludwig et al. (2011) show that an overconfident player, defined as someone who underestimates the cost of effort, exerts more effort than a rational player in a Tullock contest. Santos-Pinto and Sekeris (2023) study how confidence gaps affect effort provision and entry in Lazer-Rosen tournaments and Tullock contests. They find, among other things, that the more confident player exerts lower effort in a Tullock contest. All of these studies focus on one-shot tournaments and contests. To the best of our knowledge, ours is the first theoretical study on the impact of overconfidence on a two-stage elimination contest.

Finally, our study contributes to the literature on elimination contests. The seminal contribution is Rosen (1986) who shows how to optimally set prizes in a multiple stage elimination contest. There are many studies on elimination contests regarding different aspects, such as the discouragement effects in multi-stage contests (Konrad, 2012), optimal prize setting (Mago et al, 2013; Cheng et al, 2019; Coehn et al, 2018; Moldovanu and Sela, 2006), optimal contest structure (Gradstein and Konrad, 1999; Moldovanu and Sela, 2006; Fu and Lu, 2018; Hou and Zhang, 2021), heterogeneity in abilities (Rosen, 1986; Brown and Minor, 2014) and seeding (Groh et al, 2012). Our paper expands this strand of literature by considering a new dimension: heterogeneity in confidence levels.

The paper is organized as follows. Section 2 sets-up the model. Sections 3, 4, and 5 study an elimination contest with four rational players, one overconfident and three rational players, and two overconfident and two rational players, respectively. Section 6 considers the welfare implications of overconfidence. Section 7 shows how the con-

 $^{^{3}}$ Many studies suggest that gender gap varies with culture (Gneezy et al, 2003 and 2009; Booth and Nolen, 2009 and 2014). In societies where gender equality is more promoted, gender gaps become less significant in many areas, including entry and performance in a competitive environments. Differences in work environment, characteristics of professions, and education also affect the magnitude of gender gaps.

test designer should set prizes when players are overconfident. Section 8 discusses the extensions. Section 9 concludes the paper. All proofs are in the Appendix.

2 Set-up

Consider a two-stage elimination contest with four players. In the first stage, the players are matched pairwise, and each pair competes in one semifinal. The first-stage winners go on to the second stage of the contest and compete against each other in the final. The winner receives prize w_1 , the runner-up receives prize w_2 , and the first-stage losers receive nothing, with $w_1 > w_2 \ge 0$.

The players choose their efforts simultaneously to maximize their expected utilities in each pairwise interaction. The effort of player *i* is denoted by e_i . Players derive utility u(w) from prize $w \ge 0$, where u'(w) > 0, $u''(w) \le 0$, and u(0) = 0. Players have linear cost of effort, i.e., $c(e_i) = ce_i$, with $c \ge 1$, and $e_i \ge 0$. The probability of winning a pairwise interaction depends on the efforts of the two players through Alcalde and Dahm's (2007) contest success function. Letting p_{ij} denote player *i*'s winning probability when paired with *j*, we have

$$p_{ij} = \begin{cases} 1 - \frac{1}{2} \left(\frac{e_j}{e_i}\right)^{\alpha} & \text{if} \quad e_i \ge e_j \\ \frac{1}{2} \left(\frac{e_i}{e_j}\right)^{\alpha} & \text{if} \quad e_i \le e_j \end{cases}$$

Under this CSF a player's winning probability is increasing in effort. When players exert the same effort, each has a 50 percent chance of winning. The player who exerts more effort is the favorite, having a winning probability higher than 1/2 and his rival is the underdog. The parameter α determines how sensitive the CSF is to effort. When $\alpha = 0$ the CSF is completely insensitive to effort and we obtain the extreme case of a (fair) lottery. As α increases, the CSF becomes more sensitive to effort, and the contest becomes more deterministic until the extreme case of an all-pay auction is reached when $\alpha \to \infty$. We assume $0 < \alpha \leq 1$ which implies that each stage of the elimination contest has a unique pure strategy Nash equilibrium.

This CSF has several desirable properties. First, it is homogeneous of degree zero in players' efforts. Second, it is piecewise continuous which makes the first-order conditions solvable. Third, as mentioned by Alcalde and Dahm (2007), it yields a very tractable model for multi-stage games as the equilibrium efforts and payoffs of the subgames can be easily computed and plugged into earlier stages of the game. These properties, together with the linear cost function, allow us to derive close-form solutions for the equilibrium efforts and payoffs.

We assume an overconfident player has a correct perception of the prizes and cost of effort but overestimates the impact of his effort on his winning probability in each pairwise interaction. This definition of overconfidence is in line with Santos-Pinto (2008, 2010) and Santos-Pinto and Sekeris (2023). Letting λ_i denote player *i*'s level of confidence and \tilde{p}_{ij} player *i*'s perceived winning probability when paired with *j* we have:

α

$$\widetilde{p}_{ij} = \begin{cases} 1 - \frac{1}{2} \frac{e_j}{\lambda_i e_i^{\alpha}} & \text{if} \quad \lambda_i e_i^{\alpha} \geqslant e_j^{\alpha} \\ \frac{1}{2} \frac{\lambda_i e_i^{\alpha}}{e_i^{\alpha}} & \text{if} \quad \lambda_i e_i^{\alpha} \leqslant e_j^{\alpha} \end{cases}$$
(1)

We assume $\lambda_i \ge 1$, meaning that a player can be either rational ($\lambda_i = 1$) or overconfident ($\lambda_i > 1$). We see from (1) that an overconfident player holds a higher perceived winning probability than his true winning probability for any given efforts. On Figure 1, we depict the true (solid blue curve) and the perceived (solid red curve) winning probabilities of an

overconfident player with $\lambda_i = 1.5$. We set $\alpha = 0.9$ and $e_j = 1$. We see that the true and perceived winning probabilities are increasing in effort. Note that when player *i*'s effort is identical to *j*'s, i.e., $e_i = e_j = 1$, player *i*'s true winning probability is 1/2.



Figure 1: True and Perceived Winning Probabilities

The bias also affects an overconfident player's perceived marginal winning probability. Letting $mg\tilde{p}_{ij} = \partial \tilde{p}_{ij}/\partial e_i$ denote player *i*'s perceived marginal winning probability when paired with *j*, it follows from (1) that

$$mg\widetilde{p}_{ij} = \begin{cases} \frac{\alpha}{2\lambda_i} \frac{e_j^{\alpha}}{e_i^{\alpha+1}} & \text{if} \quad \lambda_i e_i^{\alpha} \ge e_j^{\alpha} \\ \frac{\alpha}{2}\lambda_i \frac{e_i^{\alpha-1}}{e_j^{\alpha}} & \text{if} \quad \lambda_i e_i^{\alpha} \le e_j^{\alpha} \end{cases}$$
(2)

On Figure 2, we depict the true (solid blue curve) and the perceived (solid red curve) marginal winning probabilities of an overconfident player with $\lambda_i = 1.5$. As before, we set $\alpha = 0.9$ and $e_j = 1$. We see that the true and perceived marginal winning probabilities are decreasing in effort. Furthermore, the perceived marginal winning probability of an overconfident player is less (greater) than his true marginal winning probability when his effort is high (low).



Figure 2: True and Perceived Marginal Winning Probabilities

The solution concept is Subgame Perfect Nash Equilibrium. We solve the elimination contest via backwards induction and determine the equilibrium of the second-stage (the final) before we determine equilibrium in the first-stage (the semifinals). To be able to compute the equilibrium taking into account that players can hold mistaken beliefs we assume: (i) a player who faces a biased opponent is aware that the latter's perception (and probability of winning) is mistaken, (ii) each player thinks that his own perception (and probability of winning) is correct, and (iii) both players have a common understanding of each other's beliefs, despite their disagreement on the accuracy of their opponent's beliefs. Hence, players agree to disagree about their perceptions (and winning probabilities). This approach follows Heifetz et al. (2007a, 2007b) for games with complete information, and Squintani (2006) for games with incomplete information.⁴ Finally, we assume that each player not only knows the confidence level of his direct rival in the semifinal but also the confidence levels of the other two potential rivals in the other semifinal.⁵

In a final between players i and j, player i chooses the level of effort e_i that maximizes his perceived expected utility:

$$\widetilde{E}^f(U_{ij}) = \widetilde{p}_{ij}^f u(w_1) + (1 - \widetilde{p}_{ij}^f)u(w_2) - ce_i$$

subject to the participation constraint $\widetilde{E}^{f}(U_{ij}) \geq 0$, where \widetilde{p}_{ij}^{f} is player *i*'s perceived winning probability in a final against *j*. Hence, the first-order condition of player *i* in a final against *j* is given by:

$$mg\widetilde{p}_{ij}^f[u(w_1) - u(w_2)] = c.$$
(3)

Since

$$\frac{\partial^2 \widetilde{p}_{ij}^f}{\partial e_i^2} < 0,$$

 $^{^{4}}$ These assumptions are consistent with the psychology literature on the "Blind Spot Bias" according to which individuals believe that others are more susceptible to behavioral biases than themselves (Pronin et al. 2002, Pronin and Kugler 2007). ⁵In Section 8 we discuss what happens when overconfidence is unobservable.

we have

$$\frac{\partial^2 \widetilde{E}^f(U_{ij})}{\partial e_i^2} < 0$$

and the second-order condition is satisfied.

Let $R_i^f(e_j)$ denote player *i*'s best response in the final obtained from (3). Lemma 1 describes the shape of player *i*'s best response in the final.

Lemma 1 $R_i^f(e_i)$ is quasi-concave in e_i and reaches a maximum at $\lambda_i e_i^{\alpha} = e_i^{\alpha}$.

Lemma 1 tells us that the players' best responses in the final are non-monotonic. Given high effort of the rival, a player reacts to an increase in effort of the rival by decreasing effort; given low effort of the rival, a player reacts to an increase in effort of the rival by increasing effort.

Lemma 2 describes how player *i*'s best response in the final changes with his overconfidence parameter λ_i .

Lemma 2 An increase in player i's overconfidence λ_i leads to a contraction of his best response in the final, $\partial R_i^f / \partial \lambda_i < 0$, for $e_j^{\alpha} < \lambda_i e_i^{\alpha}$, and to an expansion of his best response in the final, $\partial R_i^f / \partial \lambda_i > 0$, for $e_j^{\alpha} > \lambda_i e_i^{\alpha}$. Moreover, the maximum value of player i's best response in the final is independent of player i's overconfidence.

Lemma 2 characterizes how overconfidence shifts a player's best response in the final. For a high effort of the rival, an increase in confidence raises player i's effort level, while for low effort of the rival, an increase in confidence lowers player i's effort level. Moreover, the maximal value taken by player i's best response in the final is independent of his overconfidence bias.

Now consider the semifinals stage. Let players i and h be paired up in one semifinal and players j and k be paired up in the other semifinal. If player i wins his semifinal, then i faces j in the final with probability p_{jk}^s and k in the final with probability $p_{kj}^s = 1 - p_{jk}^s$. Hence, player i's perceived expected utility of the final before knowing the identity of his rival in the final is $p_{jk}^s \tilde{E}^f(U_{ij}) + p_{kj}^s \tilde{E}^f(U_{ik})$. Therefore, player i's perceived benefit of winning his semifinal, or player i's perceived continuation value \tilde{v}_i , is:

$$\widetilde{v}_i = p^s_{ik} \widetilde{E}^f(U_{ij}) + p^s_{kj} \widetilde{E}^f(U_{ik}).$$

In the semifinal between players i and h, player i chooses the level of effort e_i that maximizes his perceived expected utility:

$$E^s(U_{ih}) = \widetilde{p}^s_{ih}\widetilde{v}_i - ce_i.$$

subject to the participation constraint $\widetilde{E}^{s}(U_{ih}) \geq 0$, where \widetilde{p}_{ih}^{s} is player *i*'s perceived winning probability in a semifinal against *h*. Hence, the first-order condition of player *i* in a semifinal against *h* is given by:

$$mg\widetilde{p}_{ih}^{s}\widetilde{v}_{i} = c. \tag{4}$$

Since

$$\frac{\partial^2 \widetilde{p}^s_{ih}}{\partial e_i^2} < 0,$$

we have

$$\frac{\partial^2 \widetilde{E}^s(U_{ih})}{\partial e_i^2} < 0,$$

and the second-order condition is satisfied.

Let $R_i^s(e_h)$ denote player *i*'s best response in the semifinal obtained from (4). It can be inferred from equations (3) and (4) that the shape of player *i*'s best response in the semifinal closely mirrors that of his best response in the final. However, Lemma 3 shows that the impact of overconfidence on player *i*'s best response in the semifinal diverges from the final.

Lemma 3 If player i's perceived continuation value \tilde{v}_i increases in his overconfidence $\lambda_i, \ \partial \tilde{v}_i/\partial \lambda_i > 0$, then an increase in player i's overconfidence leads to a contraction of his best response in the semifinal, $\partial R_i^s/\partial \lambda_i < 0$, for $e_j^{\alpha} < \lambda_i e_i^{\alpha}$ and $\partial \tilde{v}_i/\partial \lambda_i < \tilde{v}_i/\lambda_i$, otherwise, it leads to an expansion of his best response in the semifinal, $\partial R_i^s/\partial \lambda_i > 0$. Moreover, the maximum value of player i's best response in the semifinal increases in player i's overconfidence.

Lemma 3 characterizes how overconfidence shifts a player's best response in the semifinal. It shows that if player i's perceived continuation value increases in overconfidence, then an increase in player i's overconfidence contracts his best response in the semifinal for low effort of the rival when the elasticity of the perceived continuation value with respect to bias is smaller than 1. Otherwise, an increase in player i's overconfidence expands his best response in the semifinal. Moreover, if player i's perceived continuation value increases in his overconfidence, then the maximal value taken by player i's best response in the semifinal is increasing in his overconfidence bias.

Let P_i denote player *i*'s probability of winning the elimination contest. P_i is the product of player *i*'s probability of winning his semifinal and winning the final before knowing the identity of his rival in the final:

$$P_i = p_{ih}^s (p_{jk}^s p_{ij}^f + p_{kj}^s p_{ik}^f)$$

We denote the four players as 1, 2, 3, and 4 from now on. Players 1 and 2 are paired in one semifinal and players 3 and 4 are paired in the other semifinal.

3 Four Rational Players

This section characterizes the equilibrium of an elimination contest with four rational players. This serves as a benchmark to which we compare all our results.

Proposition 1 In a final with two rational players, the equilibrium effort is

$$\bar{e}^f = \frac{\alpha}{2c} [u(w_1) - u(w_2)],$$

the equilibrium winning probability is

$$\overline{p}^f = \frac{1}{2},$$

and the equilibrium expected utility is

$$\overline{E}^f(U) = \frac{1-\alpha}{2}u(w_1) + \frac{1+\alpha}{2}u(w_2).$$

In a semifinal of a two-stage elimination contest with four rational players, the equilibrium effort is

$$\overline{e}^s = \frac{\alpha}{2c} \left[\frac{1-\alpha}{2} u(w_1) + \frac{1+\alpha}{2} u(w_2) \right],$$

the equilibrium winning probability is

$$\overline{p}^s = \frac{1}{2},$$

and the equilibrium expected utility is

$$\overline{E}^{s}(U) = \frac{1-\alpha}{2} \left[\frac{1-\alpha}{2} u(w_1) + \frac{1+\alpha}{2} u(w_2) \right].$$

In a final featuring two rational players, equilibrium is reached when both exert the same effort, resulting in an equal winning probability for each player. The equilibrium effort in the final increases in $u(w_1) - u(w_2)$, in α , and decreases in c. Similarly, in a semifinal featuring two rational players, equilibrium is reached when both exert the same effort, resulting in an equal winning probability for each player. The equilibrium effort in the semifinal is smaller than the equilibrium effort in the final for all $\alpha \in (0, 1]$ when $u(w_1) > 3u(w_2)$. Finally, we have

$$\overline{P} = \overline{p}^s \overline{p}^f = \frac{1}{4}$$

When all players are rational the elimination contest is symmetric and hence each has 1/4 probability of being the winner.

4 One Overconfident Player and Three Rational Players

This section characterizes the equilibrium of an elimination contest with one overconfident player and three rational players. Throughout we assume player 1 is overconfident with $\lambda_1 > 1$ and players 2, 3, and 4 are rational with $\lambda_2 = \lambda_3 = \lambda_4 = 1$.

4.1 Final

We start by analyzing the impact of overconfidence on the final. Since players 3 and 4 are identical, we consider a final with an overconfident player 1 and a rational player 3 without loss of generality.

Proposition 2 In a final between an overconfident player and a rational player, the equilibrium effort of the overconfident player is

$$e_1^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} [u(w_1) - u(w_2)],$$

and the equilibrium effort of the rational player is

$$e_3^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}} [u(w_1) - u(w_2)]$$

with $e_1^f < e_3^f < \overline{e}^f$. The perceived equilibrium winning probability of the overconfident player is

$$\widetilde{p}_{13}^f = 1 - \frac{1}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}},$$

and the true equilibrium winning probabilities are

$$p_{13}^f = \frac{1}{2}\lambda_1^{-\frac{\alpha}{2\alpha+1}}$$

$$p_{31}^f = 1 - \frac{1}{2}\lambda_1^{-\frac{\alpha}{2\alpha+1}}$$

with $\tilde{p}_{13}^f > p_{31}^f > 1/2 > p_{13}^f$. The perceived equilibrium expected utility of the overconfident player is

$$\widetilde{E}^{f}(U_{13}) = u(w_1) - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}[u(w_1) - u(w_2)]$$

and the equilibrium expected utility of the rational player is

$$E^{f}(U_{31}) = u(w_{1}) - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})],$$

with $\widetilde{E}^f(U_{13}) > E^f(U_{31}) > \overline{E}^f(U)$.

Proposition 2 shows that an overconfident player always exerts less effort in a final than a rational rival. This result is intuitive. In the final, the overconfident player chooses the level of effort at which the perceived marginal benefit of effort equals the marginal cost and the rational player chooses the level of effort at which the marginal benefit of effort equals the marginal cost. Hence, in the final, the players' equilibrium efforts satisfy

$$mg\widetilde{p}_{13}^{f}[u(w_1) - u(w_2)] = c,$$

and

$$mgp_{31}^f[u(w_1) - u(w_2)] = c.$$

An increase in confidence leads to a drop in the overconfident player's perceived marginal probability of winning the final $mg\tilde{p}_{13}^f$. As a result, he lowers effort to save on costs of effort. The rational player, anticipating the overconfident player will lower his effort, also lowers her effort but not as much. Hence, both players exert lower effort than if both were rational. At equilibrium, the perceived winning probability of the overconfident player is greater than 1/2 whereas his true winning probability is less than 1/2 given the lower equilibrium effort. Intuitively, the overconfident player, given his (mis)perceived advantage, thinks, mistakenly, he can reduce his effort without endangering his prospects of success.

Proposition 2 also shows that the overconfident player's perceived equilibrium expected utility of the final increases in his bias. This happens due to two channels. First, an increase in the bias raises the overconfident player's perceived probability of winning the final. Second, an increase in the bias lowers the overconfident player's cost of effort. Furthermore, Proposition 2 shows that the rational player's equilibrium expected utility of the final increases in the overconfident player's bias. An increase in the overconfident player's bias raises the rational player's probability of winning the final and lowers his cost of effort. Hence, an increase in an overconfident player's bias makes reaching the final more attractive not only for the overconfident player but also for a rational rival. Note also that, at equilibrium, the overconfident player's perceived expected utility of the final is greater than that of the rational player.⁶

Figure 3 illustrates Proposition 2. It depicts the best responses and equilibrium efforts in a final where $\alpha = 0.9$ and $[u(w_1) - u(w_2)]/c = 2$. The best response of a rational player 1 is depicted in solid red and the best response of a rational player 3 in solid blue. The equilibrium when players 1 and 3 are rational is depicted by point E at the 45 degree

⁶As the overconfident player's bias converges to infinity, the efforts of both players converge to zero, the overconfident player's perceived probability of winning the final converges to 1, his true probability of winning the final converges to zero, his perceived equilibrium expected utility of the final converges to the winner's prize $u(w_1)$ and so does the rational player's equilibrium expected utility of the final.

line. The best response of an overconfident player 1 with $\lambda_1 = 1.2$ is depicted in dotted red. The equilibrium when player 1 is overconfident and player 3 is rational is depicted by point E' above the 45 degree line. In the final, overconfidence shifts the best response of player 1 inwards for low values of effort of player 3 and outwards for high values of effort of player 3.



Figure 3: Best Responses and Equilibrium Efforts in the Final

4.2 Semifinals

We now analyze the impact of overconfidence on the two semifinals. We start with the semifinal with overconfident player 1 and rational player 2. Next, we consider the semifinal with rational players 3 and 4.

Proposition 3 Consider a semifinal between an overconfident player and a rational player of a two-stage elimination contest where player 1 is overconfident and players 2, 3, and 4 are rational.

(i) If $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 < \hat{\lambda}$ where $\hat{\lambda}$ is given by $\frac{1+\alpha}{2} \frac{u(w_1)-u(w_2)}{u(w_1)} = \frac{\hat{\lambda}-1}{\hat{\lambda}-\hat{\lambda}^{-\frac{\alpha+1}{2\alpha+1}}}$, then the equilibrium efforts and winning probabilities satisfy $e_1^s > \overline{e}^s > e_2^s$ and $\widetilde{p}_{12}^s > p_{12}^s > \frac{1/2}{2} > p_{21}^s$. (ii) If either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leqslant \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \ge \hat{\lambda}$, then the equilibrium efforts and winning probabilities satisfy $e_1^s < e_2^s \leqslant \overline{e}^s$ and $\widetilde{p}_{12}^s > p_{21}^s \ge p_{12}^s$.

Proposition 3 shows that an overconfident player exerts higher effort in a semifinal than a rational rival when the prize spread is sufficiently large, $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$, and overconfidence is not too extreme, $\lambda_1 < \hat{\lambda}$. When either of these two conditions is not met, the overconfident player exerts less effort in the semifinal than a rational rival. This result, together with Proposition 2, provides the model's main testable implication, namely, that an overconfident player always exerts less effort in a final than a rational rival but can exert higher effort in semifinal than a rational rival.

In a semifinal, the overconfident player chooses the level of effort at which the perceived marginal benefit $mg\widetilde{p}_{12}^s\widetilde{v}_1$ equals the marginal cost c, and the rational player chooses the level of effort at which the marginal benefit $mgp_{21}^sv_2$ equals the marginal cost c. Hence, in a semifinal, the players' equilibrium efforts satisfy

$$mg\widetilde{p}_{12}^s\widetilde{v}_1 = c,\tag{5}$$

and

$$mgp_{21}^s v_2 = c. ag{6}$$

Differentiating (5) and (6) and solving for $\partial e_1^s / \partial \lambda_1$ we find how the overconfident player's equilibrium effort changes in his bias

$$\frac{\partial e_1^s}{\partial \lambda_1} = \frac{mg\widetilde{p}_{12}^s}{\lambda_1} \frac{\frac{\partial mg\widetilde{p}_{12}^s}{\partial \lambda_1} \frac{\lambda_1}{mg\widetilde{p}_{12}^s} + \frac{\partial \widetilde{v}_1}{\partial \lambda_1} \frac{\lambda_1}{\widetilde{v}_1}}{-\frac{\partial mg\widetilde{p}_{12}^s}{\partial e_1} + \frac{\partial mg\widetilde{p}_{12}^s}{\partial e_2} \frac{\frac{\partial mgp\widetilde{p}_{21}^s}{\partial e_1}}{\frac{\partial mgp\widetilde{p}_{21}^s}{\partial e_2}},\tag{7}$$

where the sign of the denominator in equation (7) is positive.⁷ In equation (7), the term

$$\varepsilon_{mg\tilde{p}_{12}^s,\lambda_1} = \frac{\partial mg\tilde{p}_{12}^s}{\partial\lambda_1} \frac{\lambda_1}{mg\tilde{p}_{12}^s},\tag{8}$$

represents the elasticity of the perceived marginal probability of winning the semifinal with respect to the bias. Meanwhile, the term

$$\varepsilon_{\widetilde{v}_1,\lambda_1} = \frac{\partial \widetilde{v}_1}{\partial \lambda_1} \frac{\lambda_1}{\widetilde{v}_1},\tag{9}$$

represents the elasticity of the perceived continuation value with respect to the bias. Hence, the sign of $\partial e_1^s / \partial \lambda_1$ depends on the signs of these two elasticities.

The proof of Proposition 3 shows that at equilibrium $\lambda_1 e_1^{\alpha} \ge e_3^{\alpha}$ which, together with (2), implies that the elasticity of the perceived marginal probability of winning the semifinal with respect to the bias is

$$\varepsilon_{mg\tilde{p}_{12}^s,\lambda_1} = -\frac{\alpha}{2\lambda_1^2} \frac{e_2^{\alpha}}{e_1^{\alpha+1}} \frac{\lambda_1}{\frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}}} = -1.$$
(10)

The overconfident player's perceived continuation value is

$$\widetilde{v}_1 = u(w_1) - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}[u(w_1) - u(w_2)]$$

Hence, the elasticity of the perceived continuation value with respect to the bias is

$$\varepsilon_{\tilde{v}_1,\lambda_1} = \frac{(\alpha+1)^2}{2(2\alpha+1)} \frac{1}{\frac{u(w_1)}{u(w_1) - u(w_2)} \lambda_1^{\frac{\alpha+1}{2\alpha+1}} - \frac{1+\alpha}{2}}.$$
(11)

We see from (10) and (11) that the bias has two effects on the overconfident player's incentives to exert effort in a semifinal. On the one hand, an increase in the bias lowers the overconfident player's perceived marginal probability of winning the semifinal which motivates him to lower effort. We label this the complacency effect of overconfidence.

 $^{^{7}}$ In the Appendix we derive equation (7) and show that its denominator is positive.

On the other hand, an increase in the bias raises the overconfident player's perceived continuation value which motivates him to raise effort. We label this the encouragement effect of overconfidence. When $\varepsilon_{\tilde{v}_1,\lambda_1} > (<)1$ the encouragement (complacency) effect dominates, and the overconfident player exerts higher (lower) effort in the semifinal than the rational player.

Equation (10) shows that the size of the complacency effect is -1 and equation (11) shows that the size of the encouragement effect decreases in the bias and converges to zero when the bias converges to infinity (as $\lambda_1 \to \infty$ we have $\tilde{v}_1 \to u(w_1)$). Hence, when the bias is small, i.e., λ_1 is close to 1, a necessary condition for the encouragement effect to dominate is that $\varepsilon_{\tilde{v}_1,\lambda_1} > 1$ at $\lambda_1 = 1$. Setting $\lambda_1 = 1$ in (11) this is equivalent to $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$. Thus, the encouragement effect dominates when the prize spread is sufficiently large and the bias is close to 1. As the bias increases, the size of the encouragement effect decreases and converges to zero while the size of the complacency effect is fixed at -1. Hence, there exists an upper bound for the bias above which the complacency effect dominates. Figure 4 depicts in light green the values of $\left(\alpha, \frac{u(w_1)-u(w_2)}{u(w_2)}\right)$ that satisfy $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$.



Figure 4: Values of $\left(\alpha, \frac{u(w_1)-u(w_2)}{u(w_2)}\right)$ where $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$

1

Figure 5 illustrates result (i) in Proposition 3. It depicts the best responses and equilibrium efforts in a semifinal of an elimination contest where $u(w_1) = 11.25$, $u(w_2) = 1.25$, $\alpha = 0.9$, and c = 1. Note that these parameter values imply $\hat{\lambda} = 4.4606$ and satisfy $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$. The best response of a rational player 1 is depicted in solid red and that of a rational player 2 in solid blue. Point E at the 45 degree line depicts the equilibrium when players 1 and 2 are rational. The best response of an overconfident player 1 with $\lambda_1 = 1.1$ is depicted in dashed red. Point point E' below the 45 degree line depicts the equilibrium when player 1 is overconfident with $\lambda_1 = 1.1$ and player 1 is overconfident with $\lambda_1 = 1.1$ is overconfident with $\lambda_1 = 1.1$ is ove

2 is rational. Since the encouragement effect dominates, overconfidence shifts the best response of player 1 outwards for all values of effort of player 2.



Figure 5: Best Responses and Equilibrium Efforts in a Semifinal where $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 < \hat{\lambda}$

Figure 6 illustrates result (ii) in Proposition 3. It depicts the best responses and equilibrium efforts in a semifinal of an elimination contest where $u(w_1) = 11.25$, $u(w_2) = 1.25$, $\alpha = 0.9$, and c = 1. The best response of an overconfident player 1 with $\lambda_1 = 7$ is depicted in dashed red. Point point E' above the 45 degree line depicts the equilibrium when player 1 is overconfident with $\lambda_1 = 7$ and player 2 is rational. Since the complacency effect dominates, overconfidence shifts the best response of player 1 inwards for low values of effort of player 2 and outwards for high values of effort of player 2.



Figure 6: Best Responses and Equilibrium Efforts in a Semifinal where either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \geq \hat{\lambda}$

Next, we characterize the equilibrium of the semifinal with rational players 3 and 4.

Proposition 4 In a semifinal between two rational players of a two-stage elimination contest where player 1 is overconfident and players 2, 3, and 4 are rational, the equilibrium efforts and winning probabilities satisfy $e_3^s = e_4^s > \overline{e}^s$ and $p_{34}^s = p_{43}^s = 1/2$.

Proposition 4 shows that the presence of an overconfident player in one of the semifinals of an elimination contest has a spillover effect on the equilibrium effort in the other semifinal. In the semifinal with two rational players, both players has an incentive to exert higher effort when player 1 is overconfident than if player 1 were rational. This happens because the winner of the semifinal with two rational players will face the overconfident player 1 in the final with probability p_{12}^s and this leads to a higher continuation value than if player 1 were rational.

4.3 Equilibrium Winning Probabilities

We now consider how overconfidence affects each player's equilibrium probability of winning the elimination contest. We are interested in knowing whether the overconfident player 1 can have the brighter prospects throughout the elimination contest. The winning probability of overconfident player 1 is

$$P_1 = p_{12}^s (p_{34}^s p_{13}^f + p_{43}^s p_{14}^f) = p_{12}^s p_{13}^f,$$

where the second equality follows from $p_{13}^f = p_{14}^f$. The winning probability of rational player 2 is

$$P_2 = p_{21}^s (p_{34}^s p_{23}^f + p_{43}^s p_{24}^f) = p_{21}^s \frac{1}{2},$$

where the second equality follow from $p_{34}^s + p_{43}^s = 1$ and $p_{23}^f = p_{24}^f = 1/2$. The winning probabilities of rational players 3 and 4 are

$$P_3 = P_4 = p_{34}^s (p_{12}^s p_{31}^f + p_{21}^s p_{32}^f) = \frac{1}{2} \left[p_{12}^s (1 - p_{13}^f) + (1 - p_{12}^s) \frac{1}{2} \right]$$

Since the overconfident player 1 has an equilibrium probability of winning the final p_{13}^f that is less than 1/2, a necessary condition for him to have the highest equilibrium winning probability is that his equilibrium probability of winning the semifinal p_{12}^s is greater than 1/2. In other words, the overconfident player must exert higher effort in his semifinal than the rational player 2. When this is the case, rational player 2 has an equilibrium probability of winning which is less than 1/4 since $p_{21}^s = 1 - p_{12}^s < 1/2$. Regardless of the identity of the winner of the semifinal between players 1 and 2, rational players 3 and 4 have an equilibrium winning probability greater than 1/4 since they have a positive probability of facing the overconfident player 1 in the final. Hence, in equilibrium, the overconfident player 1 has the highest winning probability when $P_1 > P_3 = P_4$ which, from the equations above, is equivalent to $6p_{12}^s p_{13}^f - p_{12}^s - 1 > 0$.

Proposition 5 In a two-stage elimination contest where player 1 is overconfident and players 2, 3, and 4 are rational, if $\alpha > \frac{\sqrt{97-5}}{12}$ and $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(4\alpha+5)}{6\alpha^2+5\alpha-3}$, then there exist $\lambda_1 \in (1, \hat{\lambda})$ for which the overconfident player has the highest equilibrium winning probability, i.e., $P_1 > P_3 = P_4 > 1/4 > P_2$.

Proposition 5 shows that, in equilibrium, the overconfident player can have the highest probability of winning the elimination contest. Moreover, for this to be the case three conditions need to be met. First, the CSF's effort sensitivity parameter α must be greater than $\frac{\sqrt{97-5}}{12} \approx 0.404$. Hence, the role that effort plays in determining the winning probabilities must be sufficiently high. Second, the utility prize spread needs to be sufficiently large, $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(4\alpha+5)}{6\alpha^2+5\alpha-3}$. Third, the overconfidence bias must be low. Figure 7 depicts in light green the values of $\left(\alpha, \frac{u(w_1)-u(w_2)}{u(w_2)}\right)$ that satisfy the two inequalities in Proposition 5.



Figure 7: Values of $\left(\alpha, \frac{u(w_1) - u(w_2)}{u(w_2)}\right)$ where $\alpha > \frac{\sqrt{97} - 5}{12}$ and $\frac{u(w_1) - u(w_2)}{u(w_2)} > \frac{2(4\alpha + 5)}{6\alpha^2 + 5\alpha - 3}$

5 Two Overconfident Players and Two Rational Players

This section studies the equilibrium of an elimination contest with two overconfident players and two rational players. We assume the two overconfident players differ in their confidence levels. This setup enables us to assess if our prior findings remain applicable when two overconfident players encounter each other, either in the final or the semifinal.

Now, there are two possible seedings: (i) the overconfident players are seeded in the same semifinal, and (ii) the overconfident players are seeded in different semifinals. These two types of seeding induce different results and hence we study them separately.

5.1 Final

When the overconfident players are seeded in the same semifinal, the final will be played between an overconfident and a rational player and we can apply Proposition 2. In contrast, when the overconfident players are seeded in different semifinals, the final can have two overconfident players. Hence, we start by characterizing the equilibrium of a final with two overconfident players. Without loss of generality we consider a final between players 1 and 3 with $\lambda_1 > \lambda_3 > 1$.

Proposition 6 In a final between two overconfident players, the equilibrium effort of the more overconfident player is

$$e_1^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \lambda_3^{-\frac{\alpha}{2\alpha+1}} [u(w_1) - u(w_2)]$$

and the equilibrium effort of the less overconfident player is

$$e_3^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}} \lambda_3^{-\frac{\alpha+1}{2\alpha+1}} [u(w_1) - u(w_2)]$$

with $e_1^f < e_3^f < \overline{e}^f$. The perceived equilibrium winning probabilities are

$$\widetilde{p}_{13}^{f} = 1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha}{2\alpha+1}}$$
$$\widetilde{p}_{31}^{f} = 1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}}$$

with $\tilde{p}_{13}^f > \tilde{p}_{31}^f > 1/2$. The equilibrium winning probabilities are

$$p_{13}^{f} = \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{\frac{\alpha}{2\alpha+1}}$$
$$p_{31}^{f} = 1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{\frac{\alpha}{2\alpha+1}}$$

with $p_{13}^f < 1/2 < p_{31}^f$. The perceived equilibrium expected utilities are

$$\widetilde{E}^{f}(U_{13}) = u(w_{1}) - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})],$$

$$\widetilde{E}^{f}(U_{31}) = u(w_{1}) - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}}[u(w_{1}) - u(w_{2})],$$

$$\widetilde{E}^{f}(U_{13}) \geq \overline{E}^{f}(U_{13})$$

with $\widetilde{E}^f(U_{13}) > \widetilde{E}^f(U_{31}) > \overline{E}^f(U)$.

Proposition 6 shows that in a final between two overconfident players, the more overconfident player exerts lower effort at equilibrium. As we have seen, the bias lowers an overconfident player's perceived marginal probability of winning the final. The more overconfident a player is, the higher is the drop in his perceived marginal probability of winning the final. Hence, the more overconfident player exerts lower effort at equilibrium. Both players exert lower effort than if both were rational. Each player perceives he has a winning probability greater than 1/2 but, in fact, only the less overconfident player has a true winning probability greater than 1/2. The perceived expected utility of each player is increasing in his own bias as well as in the rival's bias.

5.2Overconfident players seeded in the same semifinal

Assume players 1 and 2, seeded in one semifinal, are overconfident with $\lambda_1 > \lambda_2 > 1$ and players 3 and 4, seeded in the other semifinal, are rational with $\lambda_3 = \lambda_4 = 1$. Note that, under this seeding, the final will involve an overconfident and a rational player and hence we can apply Proposition 2. Note also that since the two rational players are identical, they exert equal efforts in the semifinal and hence, each has an equal probability of winning it. This means that the identity of winner of the semifinal between two rational players does not affect the overconfident players' behavior in their semifinal. However, since the overconfident players' biases differ, the identity of winner of the semifinal between two overconfident players matters for the effort choices of the rational players in their semifinal. Taking this into account, we start by solving the equilibrium of the semifinal with two overconfident players.

Proposition 7 Consider the semifinal between two overconfident players of a two-stage elimination contest where the overconfident players 1 and 2 are seeded in one semifinal,

the rational players 3 and 4 are seeded in the other semifinal, and $\lambda_1 > \lambda_2 > 1 = \lambda_3 = \lambda_4$. (i) If $\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} \leq \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}}$, then the equilibrium efforts and winning probabil-ities satisfy $e_1^s > e_2^s$, $e_1^s > \overline{e}^s$, and $\widetilde{p}_{12}^s > p_{12}^s > 1/2 \geq \widetilde{p}_{21}^s > p_{21}^s$.

 $\begin{array}{ll} \text{(ii)} & If \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}} \leqslant \frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1+\alpha} < \frac{\frac{\lambda_1}{\alpha+1}}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{\alpha+1}}{\lambda_1 - \lambda_2}, \text{ then the equilibrium efforts and} \\ \text{winning probabilities satisfy } e_1^s > e_2^s > \overline{e}^s \text{ and } \widetilde{p}_{12}^s > p_{12}^s > 1/2 > p_{21}^s. \\ & \left[\frac{\lambda_1}{\alpha+1} - \frac{\lambda_2}{\alpha+1} - \frac{\lambda_2}{\alpha+1} - \frac{(\alpha+1)^2}{\alpha+1} - \frac{\alpha+2}{\alpha+1} - \frac{(\alpha+1)^2}{\alpha+1} - \frac{\alpha+2}{\alpha+1} - \frac$

$$(iii) If \frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} \ge \max\left[\frac{\frac{\alpha+1}{\lambda_2^{2\alpha+1}} - \frac{\alpha+1}{\lambda_2^{2\alpha+1}}}{\lambda_1 - \lambda_2}, \frac{\lambda_2^{\frac{(\alpha+1)}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha(2\alpha+1)}} - \lambda_1^{-1}}\right], then the equilibrium efforts$$

and winning probabilities satisfy $e_1^s \leq e_2^s$ and $\tilde{p}_{12}^s > \tilde{p}_{21}^s > p_{21}^s \geq 1/2 \geq p_{12}^s$.

Proposition 7 reveals that in a semifinal with two overconfident players, both players can exert higher efforts than if both were rational. It also shows that the identity of the player who exerts the highest effort depends on the prize spread, on the confidence gap, $\lambda_1 - \lambda_2$, and the bias of the less overconfident player 2.

Part (i) tells us that the more overconfident player 1 exerts higher effort at equilibrium when the prize spread is large and the confidence gap is moderate.⁸ Part (ii) tells us that the more overconfident player 1 exerts higher effort at equilibrium when the prize spread is moderate, the confidence gap is small, and the bias of the less overconfident player 2 is low. In this case both players exert higher effort than if both were rational.⁹ Finally, part (iii) tells us that the less overconfident player 2 exerts higher effort at equilibrium when either the prize spread is small, or the confidence gap is large, or the confidence gap is small and the bias of the less overconfident player 2 is large.

Figure 8 illustrates result (ii) in Proposition 7. It depicts the best responses and equilibrium efforts in a semifinal of an elimination contest where $u(w_1) = 11$, $u(w_2) = 1$, c = 1, and $\alpha = 0.9$. Point E depicts the equilibrium when both players are rational. Point E' below the 45 degree line depicts the equilibrium when player 1 is overconfident with $\lambda_1 = 1.18$, and player 2 is overconfident with $\lambda_2 = 1.07$. These parameter values satisfy the two inequalities in (ii) and hence the more overconfident player 1 exerts higher effort at equilibrium than the less overconfident player 2.

⁸When the confidence gap becomes increasingly large, i.e., $\lambda_1 \to \infty$, the right hand side of the inequality in part (i) converges to $\lambda_2^{-\frac{\alpha+1}{2\alpha+1}} < 1$. When the confidence gap becomes increasingly small, i.e., $\lambda_1 \to \lambda_2$, the right hand side of the inequality in part (i) also converges to $\lambda_2^{-\frac{\alpha+1}{2\alpha+1}}$ which is less than 1. Hence, since the left hand side of the inequality in part (i) is greater than 1, the inequality cannot be satisfied when the confidence gap is either too large or too small. These two limits are computed in Lemma 4 in the Appendix.

⁹When $\lambda_1 \to \infty$, the right hand side of the second inequality in part (ii) converges to $\lambda_2^{-\frac{\alpha+1}{2\alpha+1}}$ which is less than 1. Hence, since the left hand side of the second inequality in part (ii) is greater than 1, the second inequality in part (ii) cannot be satisfied when the confidence gap is large. When $\lambda_1 \to \lambda_2$, the left hand side of the first inequality in (ii) converges $\alpha + 1$ (iii) converges $\alpha +$ to $\lambda_2^{-\frac{\alpha+1}{2\alpha+1}}$ which is less than 1 and the right hand side of the second inequality converges to $\frac{3\alpha+2}{2\alpha+1}\lambda_2^{-\frac{\alpha+1}{2\alpha+1}}$. Hence, the two inequalities in (ii) can be satisfied when the confidence gap becomes increasingly small as long as the bias of the less

overconfident player 2 is low. These two limits are computed in Lemma 4 in the Appendix.



Figure 8: Best Responses and Equilibrium Efforts in a Semifinal with Two Overconfident Players

The equilibrium of the semifinal with two rational players is similar to that derived in Proposition 4 and is relegated to Lemma 5 in the Appendix. The main difference is that now the continuation value of the rational players takes into account the fact that the overconfident players exert different efforts and hence have different winning probabilities. Still, regardless of the identity of the winner of the semifinal between the two overconfident players, the rational players will have a higher continuation value than if all players were rational. Each rational player knows she will meet an overconfident player in the final which makes reaching the final more attractive. Hence, in the semifinal with two rational players, the equilibrium effort is higher than if all players were rational.

Proposition 7 also shows that, except for a knife-hedge parameter configuration, one of the two overconfident players has a probability of winning his semifinal that is greater than 1/2. Moreover, some confidence gaps will generate quite large gaps between p_{12}^s and p_{21}^s . We also know that given the equal equilibrium effort, each rational player has an equal probability of winning his semifinal. This means that there will exist parameter configurations where an overconfident player is the one with the highest equilibrium probability of winning the elimination contest.

Hence, the findings obtained for an elimination contest with one overconfident and three rational players extend to an elimination contest where two overconfident players are seeded in one semifinal and two rational players are seeded in the other semifinal.

5.3 Overconfident players seeded in different semifinals

We continue to assume players 1 and 2 are seeded in one semifinal and players 3 and 4 are seeded in the other semifinal. However, we now assume players 1 and 3 are overconfident with $\lambda_1 > \lambda_3 > 1$, and players 2 and 4 are rational with $\lambda_2 = \lambda_4 = 1$.

In the semifinal between players 1 and 2, the continuation values are

$$\widetilde{v}_{1} = p_{34}^{s} \widetilde{E}^{f}(U_{13}) + (1 - p_{34}^{s}) \widetilde{E}^{f}(U_{14}) = u(w_{1}) - \frac{1 + \alpha}{2} \left(1 - p_{34}^{s} + p_{34}^{s} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} \right) \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2}]],$$

and

$$v_{2} = p_{34}^{s} E^{f}(U_{23}) + (1 - p_{34}^{s}) E^{f}(U_{24})$$

= $u(w_{1}) - \frac{1 + \alpha}{2} \left(1 - p_{34}^{s} + p_{34}^{s} \lambda_{3}^{-\frac{\alpha}{2\alpha + 1}}\right) [u(w_{1}) - u(w_{2}].$

In the semifinal between players 3 and 4, the continuation values are

$$\widetilde{v_3} = p_{12}^s \widetilde{E}^f(U_{31}) + (1 - p_{12}^s) \widetilde{E}^f(U_{32}) = u(w_1) - \frac{1 + \alpha}{2} \left(1 - p_{12}^s + p_{12}^s \lambda_1^{-\frac{\alpha}{2\alpha+1}} \right) \lambda_3^{-\frac{\alpha+1}{2\alpha+1}} [u(w_1) - u(w_2],$$

and

$$v_{4} = p_{12}^{s} E^{f}(U_{41}) + (1 - p_{12}^{s}) E^{f}(U_{42})$$

= $u(w_{1}) - \frac{1 + \alpha}{2} \left(1 - p_{12}^{s} + p_{12}^{s} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\right) [u(w_{1}) - u(w_{2}].$

The four expressions above shows us that the continuation values of players seeded in one semifinal depend on the equilibrium winning probabilities of players seeded in the other semifinal. As the equilibrium efforts of one semifinal cannot be solved separately from those of the other semifinal, the equilibrium efforts in the semifinals are jointly determined by the four first-order conditions $mg\tilde{p}_{12}^s\tilde{v}_1 = c, \ mgp_{21}^sv_2 = c, \ mg\tilde{p}_{34}^s\tilde{v}_3 = c,$ and $mgp_{43}^sv_4 = c.$

Still, the findings in Proposition 3 can be applied to both semifinals. In other words, we know that in both semifinals there exist parameter configurations where the overconfident player exerts higher effort than the rational player. Our next result shows that this is indeed the case.

Proposition 8 Consider the semifinals of a two-stage elimination contest where overconfident player 1 and rational player 2 are seeded in one semifinal, overconfident player 3 and rational player 4 are seeded in the other semifinal, and $\lambda_1 > \lambda_3 > 1 = \lambda_2 = \lambda_4$. If $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} < \frac{1}{\lambda_1-1} \left(\lambda_1 - \lambda_1^{-\frac{\alpha+1}{2\alpha+1}}\right) \lambda_3^{-\frac{\alpha}{\alpha+1}}$ and $\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} < \frac{1}{\lambda_3-1} \left(\lambda_3 - \lambda_3^{-\frac{\alpha+1}{2\alpha+1}}\right) \lambda_1^{-\frac{\alpha}{\alpha+1}}$, then the equilibrium efforts and winning probabilities satisfy $e_1^s > e_2^s, e_3^s > e_4^s, p_{12}^s > 1/2 > p_{21}^s$, and $p_{34}^s > 1/2 > p_{43}^s$.

Proposition 8 shows that in an elimination contest where two overconfident players are seeded in different semifinals, the overconfident players can exert higher effort at equilibrium than their rational rivals. This happens when the prize spread is sufficiently large and the overconfident players are not too confident. In this case, each overconfident player has a higher probability of winning his semifinal than his rational rival.

Hence, the results found for an elimination contest with one overconfident player and three rational players also extend to an elimination contest where one overconfident and one rational player are seeded in each semifinal.

6 Welfare

This section studies the effects of overconfidence on the welfare of the players and of the contest designer. The analysis focuses on a two-stage elimination contest with one overconfident and three rational players.

6.1 Players

To evaluate the impact of overconfidence on player i's welfare we consider how overconfidence changes player i's equilibrium expected utility in the semifinal with players i and h which is given by

$$E^{s}(U_{ih}) = p_{ih}^{s} \left[p_{ij}^{f} E^{f}(U_{ij}) + p_{ik}^{f} E^{f}(U_{ik}) \right] - ce_{i}^{s},$$

where e_i^s is player *i*'s equilibrium effort in the semifinal with *h*, p_{ih}^s is player *i*'s equilibrium winning probability in the semifinal with *h*, and the term inside parenthesis is player *i*'s equilibrium continuation value.

Proposition 9 Consider a two-stage pairwise elimination contest with one overconfident player and three rational players.

(i) If $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$, then there exist $\lambda_1 \in (1, \hat{\lambda})$ for which $E^s(U_{12}) > \overline{E}^s(U)$. (ii) If $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$ and $\lambda_1 < \hat{\lambda}$, then $E^s(U_{21}) < \overline{E}^s(U)$, otherwise, $E^s(U_{21}) \ge \overline{E}^s(U)$. (iii) $E^s(U_{34}) = E^s(U_{43}) > \overline{E}^s(U)$.

Part (i) shows that if the prize spread is large enough, then there exist (small) overconfidence levels where the equilibrium expected utility of the overconfident player 1 in the semifinal with rational player 2 is higher than if all players were rational. Part (ii) shows that if the prize spread is large enough and overconfidence is not too extreme, then the equilibrium expected utility of the rational player 2 in the semifinal with the overconfident player 1 is lower than if all players were rational. Finally, part (iii) shows that the equilibrium expected utility of each of the rational players seeded in the same semifinal is higher than if all players are rational.

6.2 Contest Designer

We assume the welfare of the contest designer is increasing in the players' aggregate effort, the sum of the efforts in the two-stages of the contest.

Proposition 10 Consider a two-stage pairwise elimination contest with one overconfident player and three rational players.

(i) The equilibrium aggregate effort in a final with players i and j satisfies $e_i^f + e_j^f \leq 2\overline{e}^f$. (ii) If $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$ and $\alpha < \frac{1}{2}$, then there exist $\lambda_1 \in (1, \hat{\lambda})$ such that the equilibrium aggregate effort in the semifinals stage satisfies $\sum_{i=1}^4 e_i^s > 4\overline{e}^s$.

Part (i) shows that the equilibrium aggregate effort in the final stage is less than or equal to the equilibrium aggregate effort in a final with two rational players. This result follows directly from Propositions 1 and 2. The impact of overconfidence on aggregate effort in the semifinals stage is harder to characterize. We know from Proposition 3 that in the semifinal with one overconfident and one rational player two situations can emerge. If the prize spread is high enough, $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$, and overconfidence is not too extreme, $\lambda_1 < \hat{\lambda}$, the overconfident player's equilibrium effort is higher than if he were rational and the rational player's equilibrium effort is lower than if she were facing a rational rival. In all other cases, both players' equilibrium efforts are less than if all players were rational. We also know from Proposition 4 that in the semifinal with two rational players equilibrium efforts go up since both players have a higher continuation value of winning their semifinal. Still, part (ii) shows that overconfidence raises aggregate effort in the semifinals stage when the prize spread is high enough, $\alpha < 1/2$, and the overconfident player's bias is small. This result implies that it is unclear whether overconfidence lowers or raises players' aggregate efforts in the two stages. As a consequence, we are unable to provide general conditions under which overconfidence unambiguously lowers or raises the contest designer's welfare.

7 Optimal Prize Structure

This section analyzes how the contest designer should optimally choose the prize structure of the contest when players are overconfident. We assume the overall prize money is fixed and consider that the goal of the contest designer is the maximization of aggregate incentives, defined as the sum of efforts provided by all players across the two stages of the contest (this is a natural objective of the contest designer, see Sisak 2009). We also assume the four players are homogeneous, that is, all hold the same overconfidence bias. Finally, we assume W units are available as the total prize money, hence $W = w_1 + w_2$.

Proposition 11 Consider a two-stage elimination contest with four identical players with an overconfidence bias equal to $\lambda \ge 1$.

(i) If players are risk-neutral, then the optimal prize structure is the winner-take-all, i.e., $w_1 = W$ and $w_2 = 0$.

(ii) If players are risk-averse, then the optimal prize structure solves $W = w_1 + w_2$ and

$$\frac{u'(w_1)}{u'(w_2)} = \frac{\frac{\alpha}{\lambda}}{2 - \frac{\alpha}{\lambda}}.$$
(12)

Proposition 11 shows how the optimal prize structure depends on the players' risk attitudes and overconfidence. Part (i) shows that the contest designer allocates all of the prize money to the winner of the final when the players are rational and risk-neutral. This result is in line with Fu and Lu (2012) and Stracke et al. (2014) who show, in the context of a pyramid Tullock contest and of a two-stage elimination Tullock contest, respectively, that a winner-take-all prize structure maximizes aggregate effort when players are rational and risk-neutral. The novel result in part (i) is that the optimal prize structure is also the winner-take-all when players are overconfident and risk-neutral.

Part (ii) shows that when the players are rational and risk-averse, the optimal prize structure involves multiple prizes with the winner of the final receiving most of the prize money and a smaller part being assigned to the runner-up. This result is in line with Krishna and Morgan (1998) findings in the context of two-stage elimination Lazer-Rosen tournament with rational and risk-averse players. The novel result in part (ii) is that when players are overconfident and risk-averse, an increase in overconfidence leads the contest designer to allocate an increasingly larger share of the prize money to the winner of the final. The contest designer, exploits the players' overconfidence by shifting pay from w_2 to w_1 given that the bias leads the players to overestimate the probability of gaining prize w_1 and underestimate the probability of gaining prize w_2 . This result is in line with Santos-Pinto (2010) who finds, in the context of a one-shot Lazear-Rosen tournament, that the tournament designer can exploit players' overconfidence by raising the prize spread.

8 Extensions

This section discusses four extensions of the model. It starts by showing that most of the results extend to a two-stage elimination Tullock contest. Next, it shows all of our results hold when the overconfident player's rivals cannot observe his bias. After that, it shows our results also extend to a three-stage elimination contest. Finally, it describes the impact of underconfidence on a two-stage elimination contest.

8.1 Two-Stage Elimination Contest with Tullock CSF

In a two-stage elimination contest where the players' winning probabilities are determined by Tullock's (1980) CSF, player *i*'s probability of winning when paired with j is

$$p_{ij} = \begin{cases} \frac{q(e_i)}{q(e_i) + q(e_j)} & \text{if } q(e_i) + q(e_j) > 0\\ \frac{1}{2} & \text{if } q(e_i) + q(e_j) = 0 \end{cases}$$

where the function $q(e_i)$, often referred to as the impact function (Ewerhart 2015), satisfies $q(0) \ge 0$, $q'(e_i) > 0$, and $q''(e_i) \le 0$.¹⁰ Following Santos-Pinto and Sekeris (2023), an overconfident player *i*'s perceived probability of winning when paired with *j* is

$$\widetilde{p}_{ij} = \begin{cases} \frac{\lambda_i q(e_i)}{\lambda_i q(e_i) + q(e_j)} & \text{if} \quad \lambda_i q(e_i) + q(e_j) > 0\\ \frac{1}{2} & \text{if} \quad \lambda_i q(e_i) + q(e_j) = 0 \end{cases}$$

where $\lambda_i > 1$. Hence, in a final between an overconfident player 1 and a rational player 3, the equilibrium efforts (e_1^f, e_3^f) satisfy

$$\frac{\lambda_1 q'(e_1^f) q(e_3^f)}{[\lambda_1 q(e_1^f) + q(e_3^f)]^2} [u(w_1) - u(w_2)] = c,$$
(13)

and

$$\frac{q'(e_3^f)q(e_1^f)}{[q(e_1^f) + q(e_3^f)]^2}[u(w_1) - u(w_2)] = c.$$
(14)

It is not possible to solve (13) and (14) explicitly for the equilibrium efforts in the final. Nevertheless, Santos-Pinto and Sekeris (2023) characterize the equilibrium of a one-shot Tullock contest between two overconfident players. In the Online Appendix we show that the results of Santos-Pinto and Sekeris (2023) also apply to a one-shot Tullock contest between one overconfident player and one rational player. Namely, the overconfident player exerts less effort than the rational player, and both players exert less effort than if both were rational. Furthermore, the equilibrium efforts of both players decrease in the overconfident player's bias. Hence, the equilibrium efforts in the final of a two-stage elimination Tullock contest satisfy the same qualitative properties as those in the final of a two-stage elimination Alcalde and Dham (2007) contest.

Let us now turn our attention to the semifinals of a two-stage elimination Tullock contest. In the semifinal between an overconfident player 1 and a rational player 2, the

¹⁰When $q(e_i) = e_i^{\alpha}$, where $0 < \alpha \leq 1$, the parameter α determines how sensitive the CSF is to effort.

equilibrium efforts (e_1^s, e_2^s) satisfy

$$\frac{\lambda_1 q'(e_1^s) q(e_2^s)}{[\lambda_1 q(e_1^s) + q(e_2^s)]^2} \widetilde{v}_1 = c,$$

and

$$\frac{q'(e_2^s)q(e_1^s)}{[q(e_1^s) + q(e_2^s)]^2}v_2 = c,$$

The Online Appendix shows that the overconfident player's perceived continuation value \tilde{v}_1 increases in his bias. Hence, overconfidence has an encouragement effect in the semifinal of a two-stage elimination Tullock contest. The complacency effect is given by

$$\frac{\partial mg\widetilde{p}_{12}^s}{\partial\lambda_1} \frac{\lambda_1}{mg\widetilde{p}_{12}^s} = -q(e_2^s)q'(e_1^s) \frac{\lambda_1 q(e_1^s) - q(e_2^s)}{(\lambda_1 q(e_1^s) + q(e_2^s))^3} \frac{\lambda_1}{\frac{\lambda_1 q(e_2^s)q'(e_1^s)}{(\lambda_1 q(e_1^s) + q(e_2^s))^2}} = -\frac{\lambda_1 q(e_1^s) - q(e_2^s)}{\lambda_1 q(e_1^s) + q(e_2^s)}.$$
(15)

It follows from (15) that there is a complacency effect when $\lambda_1 q(e_1^s) > q(e_2^s)$. Hence, the effect of overconfidence on the equilibrium efforts in a semifinal of a two-stage elimination Tullock contest will depend on the sizes of the encouragement and complacency effects.

8.2 Unobservability of Overconfidence

Our results also extend to an elimination contest where the overconfident player's rivals cannot observe his bias. In the Online Appendix we characterize the equilibrium of a final between an overconfident player and a rational player who is unaware of the overconfident player's bias. In this case, the rational player exerts the benchmark equilibrium effort \overline{e}^{f} . The overconfident player chooses a best response to \overline{e}^{f} given his mistaken beliefs, i.e., he chooses the level of effort e that solves $mq\widetilde{p}^f(e,\overline{e}^f)[u(w_2) - u(w_1)] = c$. In equilibrium, the overconfident player exerts less effort than the rational player in the final. Note that since the rational player does not lower her effort, the overconfident player lowers his effort by less than when the rational player is aware of the overconfident player's bias. An increase in the bias raises the overconfident player's perceived probability of winning the final and lowers his cost of effort. Hence, an increase in the bias raises the overconfident player's perceived equilibrium expected utility of the final. We also characterize the equilibrium of a semifinal between an overconfident player and a rational rival who is unaware of the overconfident player's bias. Once again, we show that if the prize spread is sufficiently large and the bias is not too high, the overconfident player exerts more effort than the rational player in the semifinal. In the semifinal between two rational players, overconfidence no longer leads to an increase in equilibrium effort due to the unobservability of bias. Finally, we show that there exists conditions under which the overconfident player has the highest equilibrium probability of winning the elimination contest.

8.3 Elimination Contest with Three Stages

Our results also hold in a three-stage elimination contest. In the Online Appendix we study a three-stage elimination contest with eight players. In the third-stage, the eight players are matched pairwise and each pair competes in one of the four quarterfinals. The third-stage winners move on to compete in the second-stage. The winner of the contest receives prize w_1 , the runner-up prize w_2 , the second-stage losers receive prize w_3 , and the third-stage losers receive nothing, with $w_1 > w_2 > w_3 \ge 0$.

We fully characterize the equilibrium of the quarterfinal between an overconfident and a rational player in a three-stage elimination contest with one overconfident player and seven rational players. We find that, regardless of the overconfident player exerting more or less effort than a rational rival in the semifinals stage, an overconfident player's perceived expected utility of the semifinal $\tilde{E}^s(U)$ is greater than the benchmark $\overline{E}^s(U)$. Hence, in a quarterfinal, an overconfident player has a higher perceived continuation value \tilde{v}^q than the benchmark \overline{v}^q . This implies that overconfidence always has an encouragement effect in the quarterfinals stage. Moreover, depending on the parameters of the model, the encouragement effect can dominate the complacency effect.¹¹

8.4 Underconfidence

In the Online Appendix we characterize the equilibrium of a two-stage elimination contest with one underconfident player and three rational players. The underconfident player underestimates the impact of his effort on his winning probability in each pairwise interaction, i.e., he has a bias $\lambda \in (0,1)$. We find that an underconfident player exerts less effort than a rational rival in the final. An increase in the underconfident player's bias (a decrease in λ) has two opposite effects on his perceived equilibrium expected utility of the final. On the one hand, it lowers the underconfident player's perceived probability of winning the final. On the other hand, it lowers the underconfident player's cost of effort. The former effect dominates and an underconfident player's perceived equilibrium expected utility of the final decreases in his bias. This, in turn, implies that underconfidence has a double negative effect in the semifinal. First, it lowers the underconfident player's perceived continuation value as it makes reaching the final less attractive. Second, it lowers the underconfident player's perceived marginal probability of winning the semifinal. Hence, an underconfident player exerts less effort than a rational rival in a semifinal. The presence of an underconfident player has a spillover effect on effort provision in the semifinal between the two rational players since the probability of facing an underconfident player in the final raises their continuation values. These results imply that the underconfident player has the lowest equilibrium probability of winning the elimination contest.

9 Conclusion

Our findings reveal a nuanced interplay between overconfidence and effort exertion in a two-stage elimination contest. An overconfident player expends less effort in the final stage than a rational rival. However, this pattern can be inverted in the semifinals stage, where an overconfident can exert more effort than a rational rival.

In the final stage, an overconfident player always exerts lower effort at equilibrium than a rational player. The (mis)perceived advantage of the overconfident player leads him to think, mistakenly, he can reduce his effort without endangering his prospects of success. The rational player, aware of the rival's bias, also lowers her effort but not as much. Hence, the bias unambiguously lowers the overconfident player's probability of winning the final.

Overconfidence also changes how the players' assess the attractiveness of the final. The overconfident player's perceived expected utility of the final increases in his bias. This

 $^{^{11}}$ The spillover effect is always present in the quarterfinals among the rational players who won't meet the overconfident player until the final. However, this is not true for the quarterfinal among the two rational players who have the chance of meeting the overconfident player in the semifinal. The continuation values of these two players compared to the benchmark depends on whether the overconfident player or the rational player exerts higher effort in the semifinal. The spillover effect only exists if the rational player has a higher equilibrium winning probability than the overconfident player in the semifinal.

happens due to two channels. First, the overconfident player overestimates his probability of winning the final. Second, the overconfident player anticipates, correctly, he needs to exert lower effort in the final. The rational player's expected utility of the final increases with the overconfident players' bias. She has a higher probability of winning the final and anticipates, correctly, she needs to exert lower effort in the final. Hence, overconfidence makes reaching the final more attractive not only for an overconfident player but also for a rational player.

In the semifinals stage, the bias has two opposite effects on an overconfident player's incentives to exert effort. On the one hand, it leads to a decrease in the overconfident player's perceived marginal probability of winning the semifinal. On the other hand, it raises the overconfident player's perceived continuation value due to the higher expected utility of moving on to the final. The first effect discourages an overconfident player from exerting effort whereas the second effect encourages him to exert effort. If the encouragement (complacency) effect dominates, an overconfident player exert higher (lower) effort at equilibrium in a semifinal than a rational player.

The encouragement effect dominates when two conditions are met. First, the prize spread needs to be sufficiently large. Second, the overconfident player's bias cannot be too high. The intuition behind these two conditions is as follows. The higher is the prize spread, the higher is the continuation value, and the higher is the encouragement effect. As the overconfident player's bias increases, the increase in the encouragement effect gets smaller whereas the increase in the complacency effect gets larger. Hence, there exists an upper bound for the bias above which the complacency effect dominates.

Our results also reveal that the presence of an overconfident player in an elimination contest has a spillover effect on the effort provision of the semifinal with rational players. This result is straightforward. The presence of an overconfident player in the contest makes reaching the final more attractive for rational players and hence they have incentives to exert higher effort in their semifinal.

Next, we consider how the overconfident player's bias affects his equilibrium probability of winning the elimination contest. We find that the overconfident player can be the one with the highest probability of winning the elimination contest. For this to be the case three conditions have to be met. First, the role that effort plays in determining the winning probabilities must be sufficiently high. Second, the prize spread needs to be sufficiently large. Third, the overconfident player's bias needs to be small.

Our results contribute to the literature on labor market promotions and overconfidence. More particularly, they contribute to the literature on CEO overconfidence. They provide a novel explanation for why overconfident individuals are promoted to CEO positions, namely, exerting higher efforts at the early stages of their careers. In addition, our results highlight the role that increases in executive compensation (interpreted as increases in the prize spread) can have in making elimination contests more attractive to overconfident individuals.

Future work might study elimination contests where players are not only heterogeneous in their confidence levels but also in terms of their abilities. When an overconfident player has a lower ability than a rational player, overconfidence might not only increase a player's perceived continuation value but also his perceived marginal probability of winning a final and a semifinal. Future work can also study how the contest designer should optimally set prizes in the presence of overconfident players.

References

- Alcalde, J. and Dahm, M. (2007). Tullock and hirshleifer: a meeting of the minds. *Review* of *Economic Design*, 11:101–124.
- Altmann, S., Falk, A., and Wibral, M. (2012). Promotions and incentives: The case of multistage elimination tournaments. *Journal of Labor Economics*, 30(1):149–174.
- Baik, K. H. (1994). Effort levels in contests with two asymmetric players. Southern Economic Journal, 61(2):367–378.
- Bertrand, M., Goldin, C., and Katz, L. F. (2010). Dynamics of the gender gap for young professionals in the financial and corporate sectors. *American Economic Journal: Applied Economics*, 2(3):228–255.
- Beviá, C. and Corchón, L. C. (2015). Relative difference contest success function. Theory and Decision, 78(3):377–398.
- Blau, F. D. and DeVaro, J. (2007). New evidence on gender differences in promotion rates: An empirical analysis of a sample of new hires. *Industrial Relations: A Journal of Economy and Society*, 46(3):511–550.
- Blau, F. D. and Kahn, L. M. (2017). The gender wage gap: Extent, trends, and explanations. *Journal of Economic Literature*, 55(3):789–865.
- Bronson, M. A. and Thoursie, P. S. (2019). The wage growth and within-firm mobility of men and women: New evidence and theory. Unpublished.
- Brown, J. and Minor, D. B. (2014). Selecting the best? spillover and shadows in elimination tournaments. *Management Science*, 60(12):3087–3102.
- Buser, T., Niederle, M., and Oosterbeek, H. (2021a). Can competitiveness predict education and labor market outcomes? evidence from incentivized choice and survey measures. Technical Report w28916, National Bureau of Economic Research.
- Buser, T., Ranehill, E., and Van Veldhuizen, R. (2021b). Gender differences in willingness to compete: The role of public observability. *Journal of Economic Psychology*, 83:102366.
- Cheng, X., Gou, Q., Yue, J., and Zhang, Y. (2019). Equilibrium decisions for an innovation crowdsourcing platform. Transportation Research Part E: Logistics and Transportation Review, 125:241–260.
- Cohen, N., Maor, G., and Sela, A. (2018). Two-stage elimination contests with optimal head starts. *Review of Economic Design*, 22:177–192.
- Czibor, E. and Dominguez Martinez, S. (2019). Never too late: Gender quotas in the final round of a multistage tournament. *The Journal of Law, Economics, and Organization*, 35(2):319–363.
- Deaner, R. O. (2006). More males run relatively fast in us road races: Further evidence of a sex difference in competitiveness. *Evolutionary Psychology*, 4(1):147470490600400126.
- Deutscher, C., Sahm, M., Schneemann, S., and Sonnabend, H. (2022). Strategic investment decisions in multi-stage contests with heterogeneous players. *Theory and Decision*, 93(2):281–317.

- Ewerhart, C. (2015). Contest success functions: The common-pool perspective. Working Paper Series, University of Zurich.
- Frick, B. (2011). Gender differences in competitiveness: Empirical evidence from professional distance running. *Labour Economics*, 18(3):389–398.
- Fu, Q. and Lu, J. (2012). The optimal multi-stage contest. *Economic Theory*, 51:351–382.
- Fu, Q. and Wu, Z. (2018). Feedback and favoritism in sequential elimination contests. Working Paper.
- Gillen, B., Snowberg, E., and Yariv, L. (2019). Experimenting with measurement error: Techniques with applications to the caltech cohort study. *Journal of Political Economy*, 127(4):1826–1863.
- Goel, A. M. and Thakor, A. V. (2008). Overconfidence, ceo selection, and corporate governance. the Journal of Finance, 63(6):2737–2784.
- Goldin, C. (2014). A grand gender convergence: Its last chapter. American Economic Review, 104(4):1091–1119.
- Goldin, C. and Katz, L. F. (2011). The cost of workplace flexibility for high-powered professionals. *The Annals of the American Academy of Political and Social Science*, 638(1):45–67.
- Gradstein, M. and Konrad, K. A. (1999). Orchestrating rent seeking contests. *The Economic Journal*, 109(458):536–545.
- Groh, C., Moldovanu, B., Sela, A., and Sunde, U. (2012). Optimal seedings in elimination tournaments. *Economic Theory*, 49:59–80.
- Heifetz, A., Shannon, C., and Spiegel, Y. (2007a). The dynamic evolution of preferences. *Economic Theory*, 32(2):251–286.
- Heifetz, A., Shannon, C., and Spiegel, Y. (2007b). What to maximize if you must. *Journal* of *Economic Theory*, 133:31–57.
- Hirshleifer, J. (1989). Conflict and rent-seeking success functions: Ratio vs. difference models of relative success. *Public Choice*, 63(2):101–112.
- Hou, T. and Zhang, W. (2021). Optimal two-stage elimination contests for crowdsourcing. Transportation Research Part E: Logistics and Transportation Review, 145:102156.
- Iriberri, N. and Rey-Biel, P. (2019). Competitive pressure widens the gender gap in performance: Evidence from a two-stage competition in mathematics. *The Economic Journal*, 129(620):1863–1893.
- Jia, H., Skaperdas, S., and Vaidya, S. (2013). Contest functions: Theoretical foundations and issues in estimation. *International Journal of Industrial Organization*, 31(3):211– 222.
- Kamas, L. and Preston, A. (2012). The importance of being confident: Gender, career choice, and willingness to compete. *Journal of Economic Behavior & Organization*, 83(1):82–97.
- Keller, W., Molina, T., and Olney, W. W. (2023). The gender gap among top business executives. Journal of Economic Behavior & Organization, 211:270–286.

- Kochhar, R. (2023). The enduring grip of the gender pay gap. Pew Research Center's Social Demographic Trends Project. Retrieved from https://policycommons.net/artifacts/3456461/the-enduring-grip-of-the-gendergap/4256832/ on 15 Mar 2023. CID: 20.500.12592/23j05s.
- Konrad, K. A. (2012). Dynamic contests and the discouragement effect. *Revue* d'Économie Politique, (2):233-256.
- Koszegi, B. (2014). Behavioral contract theory. *Journal of Economic Literature*, 52(4):1075–1118.
- Krishna, V. and Morgan, J. (1998). The winner-take-all principle in small tournaments. Advances in applied microeconomics, 7:61–74.
- Ludwig, S., Wichardt, P. C., and Wickhorst, H. (2011). Overconfidence can improve an agent's relative and absolute performance in contests. *Economics Letters*, 110(3):193–196.
- Mago, S. D. and Razzolini, L. (2019). Best-of-five contest: An experiment on gender differences. *Journal of Economic Behavior & Organization*, 162:164–187.
- Mago, S. D., Sheremeta, R. M., and Yates, A. (2013). Best-of-three contest experiments: Strategic versus psychological momentum. *International Journal of Industrial Organi*zation, 31(3):287–296.
- Malmendier, U. and Tate, G. (2005). Ceo overconfidence and corporate investment. *The Journal of Finance*, 60(6):2661–2700.
- Malmendier, U. and Tate, G. (2008). Who makes acquisitions? ceo overconfidence and the market's reaction. *Journal of Financial Economics*, 89(1):20–43.
- Malmendier, U. and Tate, G. (2015). Behavioral ceos: The role of managerial overconfidence. *Journal of Economic Perspectives*, 29(4):37–60.
- Moldovanu, B. and Sela, A. (2006). Contest architecture. *Journal of Economic Theory*, 126(1):70–96.
- Moore, D. A. and Healy, P. (2008). The trouble with overconfidence. *Psychological Review*, 115(2):502–517.
- Murphy, K. (2013). Executive compensation: Where we are, and how we got there. In Constantinides, G. M., Harris, M., and Stulz, R. M., editors, *Handbook of the Economics* of Finance, volume 2A, pages 211–356. North Holland: Elsevier.
- Niederle, M. and Vesterlund, L. (2007). Do women shy away from competition? do men compete too much? *The Quarterly Journal of Economics*, 122(3):1067–1101.
- Rosen, S. (1986). Prizes and incentives in elimination tournaments. *The American Economic Review*, 76(4):701–715.
- Santos-Pinto, L. (2008). Positive self-image and incentives in organisations. The Economic Journal, 118(531):1315–1332.
- Santos-Pinto, L. (2010). Positive self-image in tournaments. International Economic Review, 51(2):475–496.

- Santos-Pinto, L. and de la Rosa, L. E. (2020). Overconfidence in labor markets. In Zimmermann, K., editor, Handbook of Labor, Human Resources and Population Economics, pages 1–42. Springer.
- Santos-Pinto, L. and Sekeris, P. (2023). Overconfidence in tullock contests. Working Paper, University of Lausanne.
- Schmitt, P., Shupp, R., Swope, K., and Cadigan, J. (2004). Multi-period rent-seeking contests with carryover: Theory and experimental evidence. *Economics of Governance*, 5:187–211.
- Sisak, D. (2009). Multiple-prize contests-the optimal allocation of prizes. Journal of Economic Surveys, 23(1):82–114.
- Spinnewijn, J. (2013). Insurance and perceptions: How to screen optimists and pessimists. *The Economic Journal*, 123(569):606–633.
- Spinnewijn, J. (2015). Unemployed but optimistic: Optimal insurance design with biased beliefs. Journal of the European Economic Association, 13(1):130–167.
- Squintani, F. (2006). Mistaken self-perception and equilibrium. *Economic Theory*, 27:615–641.
- Stracke, R., Höchtl, W., Kerschbamer, R., and Sunde, U. (2014). Optimal prizes in dynamic elimination contests: Theory and experimental evidence. *Journal of Economic Behavior & Organization*, 102:43–58.
- Tullock, G. (1980). Efficient rent seeking. In Buchanan, J., Tollison, R., and Tullock, G., editors, *Toward a Theory of Rent Seeking Society*, pages 97–112. Texas A&M University Press, College Station.
- Van den Steen, E. (2005). Organizational beliefs and managerial vision. Journal of Law, Economics, & Organization, 21:256–283.
- Van Veldhuizen, R. (2022). Gender differences in tournament choices: Risk preferences, overconfidence, or competitiveness? Journal of the European Economic Association, 20(4):595–1618.

Appendix

Proof of Lemma 1

The best response of player i in the final, $R_i^f(e_j)$, is defined by (3). Hence, the slope of the best response of player i in the final is given by

$$-\frac{\frac{\partial R_i^f}{\partial e_j}}{\frac{\partial R_i^f}{\partial e_i}} = -\frac{\frac{\partial^2 \tilde{E}^f(U_{ij})}{\partial e_i \partial e_j}}{\frac{\partial^2 \tilde{E}^f(U_{ij})}{\partial e_i^2}} = -\begin{cases} \frac{\frac{\alpha^2}{2\lambda_i} \frac{e_j^{\alpha-1}}{e_i^{\alpha+1}} [u(w_1) - u(w_2)]}{-(1+\alpha)\frac{\alpha}{2\lambda_i} \frac{e_j^{\alpha}}{e_i^{\alpha+2}} [u(w_1) - u(w_2)]} = -\frac{\alpha}{1+\alpha} \frac{e_i}{e_j} & \text{if } \lambda_i e_i^{\alpha} > e_j^{\alpha} \\ \frac{-\frac{\alpha^2}{2}\lambda_i \frac{e_i^{\alpha-1}}{e_i^{\alpha+1}} [u(w_1) - u(w_2)]}{-(1-\alpha)\frac{\alpha}{2}\lambda_i \frac{e_i^{\alpha-2}}{e_j^{\alpha}} [u(w_1) - u(w_2)]} = \frac{\alpha}{1-\alpha} \frac{e_i}{e_j} & \text{if } \lambda_i e_i^{\alpha} < e_j^{\alpha} \end{cases}$$

Therefore, the sign of the slope of player *i*'s best response in the final is positive for $\lambda_i e_i^{\alpha} > e_j^{\alpha}$ and negative for $\lambda_i e_i^{\alpha} < e_j^{\alpha}$. This implies that $R_i^f(e_j)$ increases in e_j for $\lambda_i e_i^{\alpha} > e_j^{\alpha}$, reaches the maximum at $\lambda_i e_i^{\alpha} = e_j^{\alpha}$, and decreases in e_j for $\lambda_i e_i^{\alpha} < e_j^{\alpha}$.

Proof of Lemma 2

Player *i*'s best response in the final is defined by (3). Hence, we have

$$\frac{\partial R_i^f}{\partial \lambda_i} = \begin{cases} -\frac{\alpha}{2\lambda_i^2} \frac{e_j^{\alpha}}{e_i^{\alpha+1}} [u(w_1) - u(w_2)] & \text{if } \lambda_i e_i^{\alpha} > e_j^{\alpha} \\ \\ \frac{\alpha}{2} \frac{e_i^{\alpha-1}}{e_j^{\alpha}} [u(w_1) - u(w_2)] & \text{if } \lambda_i e_i^{\alpha} < e_j^{\alpha} \end{cases}$$

We see that $\partial R_i^f / \partial \lambda_i < 0$ for $\lambda_i e_i^{\alpha} > e_j^{\alpha}$ and $\partial R_i^f / \partial \lambda_i > 0$ for $\lambda_i e_i^{\alpha} < e_j^{\alpha}$. Substituting $e_j^{\alpha} = \lambda_i e_i^{\alpha}$ into equation (3) and denoting the maximal effort that *i* is willing to invest in the final by e_i^{fmax} we obtain

$$\frac{\alpha}{2\lambda_i} \frac{\lambda_i (e_i^{fmax})^{\alpha}}{(e_i^{fmax})^{\alpha+1}} [u(w_1) - u(w_2)] = \alpha$$
$$e_i^{fmax} = \frac{\alpha}{2c} [u(w_1) - u(w_2)].$$

or

This implies that the value of
$$e_i$$
 corresponding to the maximum value of player *i*'s best response in the final, e_i^{fmax} , does not depend on λ_i .

Proof of Lemma 3

Player *i*'s best response in the semifinal is defined by (4). Hence, we have

$$\frac{\partial R_i^s}{\partial \lambda_i} = \begin{cases} -\frac{\alpha}{2\lambda_i^2} \frac{e_j^{\alpha}}{e_i^{\alpha+1}} \widetilde{v}_i + \frac{\alpha}{2\lambda_i} \frac{e_j^{\alpha}}{e_i^{\alpha+1}} \frac{\partial \widetilde{v}_i}{\partial \lambda_i} = \frac{\alpha}{2\lambda_i^2} \frac{e_j^{\alpha}}{e_i^{\alpha+1}} \left(-\widetilde{v}_i + \lambda_i \frac{\partial \widetilde{v}_i}{\partial \lambda_i} \right) & \text{if } \lambda_i e_i^{\alpha} > e_j^{\alpha} \\ \frac{\alpha}{2} \frac{e_i^{\alpha-1}}{e_j^{\alpha}} \widetilde{v}_i + \frac{\alpha}{2} \lambda_i \frac{e_i^{\alpha-1}}{e_j^{\alpha}} \frac{\partial \widetilde{v}_i}{\partial \lambda_i} = \frac{\alpha}{2} \frac{e_i^{\alpha-1}}{e_j^{\alpha}} \left(\widetilde{v}_i + \lambda_i \frac{\partial \widetilde{v}_i}{\partial \lambda_i} \right) & \text{if } \lambda_i e_i^{\alpha} < e_j^{\alpha} \end{cases}$$
(16)

It follows from (16) that if $\frac{\partial \tilde{v}_i}{\partial \lambda_i} > 0$, then $\frac{\partial R_i^s}{\partial \lambda_i} > 0$ for $\lambda_i e_i^{\alpha} < e_j^{\alpha}$. That is, if the overconfident player's perceived continuation value increases in overconfidence, then an increase in overconfidence expands the overconfident player's best response in the semifinal for high effort of the rival.

It also follows from (16) that if $\frac{\partial \tilde{v}_i}{\partial \lambda_i} > 0$ and $\frac{\partial \tilde{v}_i}{\partial \lambda_i} < 1(>1)$, then $\frac{\partial R_i^s}{\partial \lambda_i} < 0(>0)$ for $\lambda_i e_i^{\alpha} > e_j^{\alpha}$. That is, if the overconfident player's perceived continuation value increases in overconfidence and the elasticity of the perceived continuation value with respect to the bias is less (greater) than 1, then an increase in overconfidence contracts (expands) the overconfident player's best response in the semifinal for low effort of the rival.

Substituting $e_j^{\alpha} = \lambda_i e_i^{\alpha}$ into equation (4) and denoting the maximal effort that *i* is willing to invest in the semifinal by e_i^{smax} we obtain

$$\frac{\alpha}{2\lambda_i} \frac{\lambda_i (e_i^{smax})^{\alpha}}{(e_i^{smax})^{\alpha+1}} \widetilde{v}_i = c$$

or

$$e_i^{smax} = \frac{\alpha}{2c} \widetilde{v}_i.$$

Hence, if $\frac{\partial \tilde{v}_i}{\partial \lambda_i} > 0$, then e_i^{fmax} increases in λ_i .

Proof of Proposition 1

The final stage

$$p_{13}^{f} = \begin{cases} 1 - \frac{1}{2} \frac{e_{3}^{\alpha}}{e_{1}^{\alpha}} & \text{if } e_{1}^{\alpha} \ge e_{3}^{\alpha} \\ \frac{1}{2} \frac{e_{1}^{\alpha}}{e_{3}^{\alpha}} & \text{if } e_{1}^{\alpha} \leqslant e_{3}^{\alpha} \end{cases}$$
$$p_{31}^{f} = \begin{cases} 1 - \frac{1}{2} \frac{e_{1}^{\alpha}}{e_{3}^{\alpha}} & \text{if } e_{3}^{\alpha} \ge e_{1}^{\alpha} \\ \frac{1}{2} \frac{e_{3}^{\alpha}}{e_{1}^{\alpha}} & \text{if } e_{3}^{\alpha} \leqslant e_{1}^{\alpha} \end{cases}$$

Rational player 1 max $E^{f}(U_{13}) = p_{13}^{f}[u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{1}$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_3^{\alpha}}{e_1^{\alpha}}\right) \left[u(w_1) - u(w_2)\right] + u(w_2) - ce_1 & \text{if} \quad e_1^{\alpha} \ge e_3^{\alpha} \\ \frac{1}{2} \frac{e_1^{\alpha}}{e_3^{\alpha}} \left[u(w_1) - u(w_2)\right] + u(w_2) - ce_1 & \text{if} \quad e_1^{\alpha} \le e_3^{\alpha} \end{cases}$$

Rational player 3 max $E^{f}(U_{31}) = p_{31}^{f}[u(w_1) - u(w_2)] + u(w_2) - ce_3$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_1^{\alpha}}{e_3^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_3 & \text{if } e_3^{\alpha} \ge e_1^{\alpha} \\ \frac{1}{2} \frac{e_3^{\alpha}}{e_1^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_3 & \text{if } e_3^{\alpha} \le e_1^{\alpha} \end{cases}$$

There are 2 cases.

$$\begin{cases} e_1^{\alpha} \geqslant e_3^{\alpha} \\ e_1^{\alpha} \leqslant e_3^{\alpha} \end{cases}$$

1. equilibrium efforts

(1) case 1: $e_1^{\alpha} \ge e_3^{\alpha}$

Player 1 max $E^{f}(U_{13}) = \left(1 - \frac{1}{2}\frac{e_{3}^{\alpha}}{e_{1}^{\alpha}}\right) [u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{1}$ Player 3 max $E^{f}(U_{31}) = \frac{1}{2}\frac{e_{3}^{\alpha}}{e_{1}^{\alpha}} [u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{3}$

F.o.c

$$[e_1] \quad \frac{\alpha}{2} \frac{e_3^{\alpha}}{e_1^{\alpha+1}} [u(w_1) - u(w_2)] - c = 0$$

$$[e_3] \quad \frac{\alpha}{2} \frac{e_3^{\alpha-1}}{e_1^{\alpha}} [u(w_1) - u(w_2)] - c = 0$$

S.o.c

$$[e_1] \quad \frac{\alpha}{2}(-\alpha - 1)\frac{e_3^{\alpha}}{e_1^{\alpha+2}}[u(w_1) - u(w_2)] < 0$$
$$[e_3] \quad \frac{\alpha}{2}(\alpha - 1)\frac{e_3^{\alpha-2}}{e_1^{\alpha}}[u(w_1) - u(w_2)] < 0$$

Solve F.O.C , we get $e_1 = e_3 = \frac{\alpha}{2c} [u(w_1) - u(w_2)]$

(2) case 2: $e_1^\alpha \leqslant e_3^\alpha$

The same as the previous case.

Thus the unique equilibrium is $\overline{e}^f = e_1^f = e_3^f = \frac{\alpha}{2c}[u(w_1) - u(w_2)],$

2. winning probabilities

The true winning probabilities are

$$\overline{p}^f = p_{13}^f = p_{31}^f = \frac{1}{2}$$

3. expected utilities of final

$$\overline{E}^{f}(U) = E^{f}(U_{13}) = E^{f}(U_{31}) = \frac{1}{2}[u(w_{1}) + u(w_{2})] - c\frac{\alpha}{2c}[u(w_{1}) - u(w_{2})]$$
$$= \frac{1 - \alpha}{2}u(w_{1}) + \frac{1 + \alpha}{2}u(w_{2})$$

Since $0 < \alpha \leq 1$, we have $\overline{E}^f(U) \ge 0$. The participation constraints are satisfied.

The semifinals stage

1. continuation values

Using the expected utility of the final, we can get the continuation values of each player in the semifinal.

The continuation value of player 1 is given by

$$v_{1} = p_{34}^{s} E^{f}(U_{13}) + p_{43}^{s} E^{f}(U_{14})$$

= $\overline{E}^{f}(U)$
= $\frac{1-\alpha}{2}u(w_{1}) + \frac{1+\alpha}{2}u(w_{2})$
Since all 4 players are identical,

$$\overline{v} = v_1 = v_2 = v_3 = v_4 = \frac{1-\alpha}{2}u(w_1) + \frac{1+\alpha}{2}u(w_2).$$

2. equilibrium efforts in the semifinal

Using the extension of the equilibrium result in the final, we can get that

$$\overline{e}^s = e_1^s = e_2^s = e_3^s = e_4^s = \frac{\alpha}{2c}\overline{v} = \frac{\alpha}{2c}\left[\frac{1-\alpha}{2}u(w_1) + \frac{1+\alpha}{2}u(w_2)\right]$$

3. true winning probabilities

$$\overline{p}^s = p_{12}^s = p_{21}^s = p_{34}^s = p_{43}^s = \frac{1}{2}$$

4. expected utilities of semifinal

$$\overline{E}^{s}(U) = \frac{1}{2}\overline{v} - c\frac{\alpha}{2c}\overline{v} = \frac{1-\alpha}{2}\overline{v} = \frac{1-\alpha}{2}\left[\frac{1-\alpha}{2}u(w_{1}) + \frac{1+\alpha}{2}u(w_{2})\right]$$

Since $0 < \alpha \leq 1$, we have $\overline{E}^s(U) \ge 0$. The participation constraints are satisfied. 5. the prize spread that satisfies $\overline{e}^s < \overline{e}^f$

$$\overline{e}^{s} < \overline{e}^{f} \iff \frac{\alpha}{2c} \left[\frac{1-\alpha}{2} u(w_{1}) + \frac{1+\alpha}{2} u(w_{2}) \right] < \frac{\alpha}{2c} [u(w_{1}) - u(w_{2})]$$

$$\iff (1-\alpha)u(w_{1}) + (1+\alpha)u(w_{2}) < 2[u(w_{1}) - u(w_{2})]$$

$$\iff (3+\alpha)u(w_{2}) < (1+\alpha)u(w_{1})$$

$$\iff \frac{u(w_{1})}{u(w_{2})} > \frac{3+\alpha}{1+\alpha}$$

Since $\alpha \in (0, 1]$ this inequality is satisfied for all α when $u(w_1) > 3u(w_2)$.

Proof of Proposition 2

The perceived winning probabilities of the players are:

$$\widetilde{p}_{13}^{f} = \begin{cases} 1 - \frac{1}{2} \frac{e_{3}^{\alpha}}{\lambda_{1} e_{1}^{\alpha}} & \text{if} \quad \lambda_{1} e_{1}^{\alpha} \geqslant e_{3}^{\alpha} \\ \frac{1}{2} \frac{\lambda_{1} e_{1}^{\alpha}}{e_{3}^{\alpha}} & \text{if} \quad \lambda_{1} e_{1}^{\alpha} \leqslant e_{3}^{\alpha} \end{cases}$$
$$p_{31}^{f} = \begin{cases} 1 - \frac{1}{2} \frac{e_{1}^{\alpha}}{e_{3}^{\alpha}} & \text{if} \quad e_{3}^{\alpha} \geqslant e_{1}^{\alpha} \\ \frac{1}{2} \frac{e_{3}^{\alpha}}{e_{1}^{\alpha}} & \text{if} \quad e_{3}^{\alpha} \leqslant e_{1}^{\alpha} \end{cases}$$

Overconfident player 1 max $\widetilde{E}^f(U_{13}) = \widetilde{p}_{13}^f[u(w_1) - u(w_2)] + u(w_2) - ce_1$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_3^{\alpha}}{\lambda_1 e_1^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_1 & \text{if} \quad \lambda_1 e_1^{\alpha} \ge e_3^{\alpha} \\ \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_3^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_1 & \text{if} \quad \lambda_1 e_1^{\alpha} \le e_3^{\alpha} \end{cases}$$

Rational player 3 max $E^{f}(U_{31}) = p_{31}^{f}[u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{3}$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_1^{\alpha}}{e_3^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_3 & \text{if} \quad e_3^{\alpha} \ge e_1^{\alpha} \\ \frac{1}{2} \frac{e_3^{\alpha}}{e_1^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_3 & \text{if} \quad e_3^{\alpha} \le e_1^{\alpha} \end{cases}$$

There are 4 cases.

 $\begin{cases} \lambda_1 e_1^{\alpha} \ge e_3^{\alpha} & and \quad e_3 \ge e_1 \\ \lambda_1 e_1^{\alpha} \ge e_3^{\alpha} & and \quad e_3 \leqslant e_1 \\ \lambda_1 e_1^{\alpha} \leqslant e_3^{\alpha} & and \quad e_3 \ge e_1 \\ \lambda_1 e_1^{\alpha} \leqslant e_3^{\alpha} & and \quad e_3 \leqslant e_1 \end{cases}$

Since $\lambda_1 > 1$, the fourth case is impossible.

1. equilibrium efforts

(1) case 1:
$$\lambda_1 e_1^{\alpha} \ge e_3^{\alpha}$$
 and $e_3 \ge e_1$
Player 1 $max \left(1 - \frac{1}{2} \frac{e_3^{\alpha}}{\lambda_1 e_1^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_1$
Player 3 $max \left(1 - \frac{1}{2} \frac{e_1^{\alpha}}{e_3^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_3$
F.o.c
 $[e_1] \quad \frac{\alpha}{2\lambda_1} \frac{e_3^{\alpha}}{e_1^{\alpha+1}} [u(w_1) - u(w_2)] - c = 0$
 $[e_3] \quad \frac{\alpha}{2} \frac{e_1^{\alpha}}{e_3^{\alpha+1}} [u(w_1) - u(w_2)] - c = 0$
S a c

S.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1}(-\alpha - 1)\frac{e_3^{\alpha}}{e_1^{\alpha+2}}[u(w_1) - u(w_2)] < 0$$
$$[e_3] \quad \frac{\alpha}{2}(-\alpha - 1)\frac{e_1^{\alpha}}{e_3^{\alpha+2}}[u(w_1) - u(w_2)] < 0$$

Solve F.O.C , we get

$$e_{1} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})]$$
$$e_{3} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})]$$

Check the conditions $\lambda_1 e_1^{\alpha} \ge e_3^{\alpha}$ and $e_3 \ge e_1$:

$$\begin{split} \lambda_1 e_1^{\alpha} \geqslant e_3^{\alpha} & \Longleftrightarrow \lambda_1 \left(\frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\alpha} \geqslant \left(\frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}}\right)^{\alpha} \\ & \Longleftrightarrow \left(\frac{\alpha}{2c}\right)^{\alpha} \lambda_1^{-\frac{(\alpha+1)\alpha}{2\alpha+1}+1} \geqslant \left(\frac{\alpha}{2c}\right)^{\alpha} \lambda_1^{-\frac{\alpha^2}{2\alpha+1}} \\ & \Longleftrightarrow \lambda_1^{-\frac{(\alpha+1)\alpha}{2\alpha+1}+1+\frac{\alpha^2}{2\alpha+1}} \geqslant 1 \\ & \Longleftrightarrow \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \geqslant 1 \\ & \Longleftrightarrow \frac{\alpha+1}{2\alpha+1} \geqslant 0 \end{split}$$

$$e_{3} \ge e_{1} \iff \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \ge \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}$$
$$\iff \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \ge \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}$$
$$\iff \lambda_{1}^{\frac{1}{2\alpha+1}} \ge 1$$
$$\iff \frac{1}{2\alpha+1} \ge 0$$

The conditions are always satisfied.

(2) case 2: $\lambda_1 e_1^{\alpha} \ge e_3^{\alpha}$ and $e_3 \le e_1$ Player 1 $max \quad \left(1 - \frac{1}{2} \frac{e_3^{\alpha}}{\lambda_1 e_1^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_1$ Player 3 $max \quad \frac{1}{2} \frac{e_3^{\alpha}}{e_1^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_3$ F.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1} \frac{e_3^{\alpha}}{e_1^{\alpha+1}} [u(w_1) - u(w_2)] - c = 0$$

$$[e_3] \quad \frac{\alpha}{2} \frac{e_3^{\alpha-1}}{e_1^{\alpha}} [u(w_1) - u(w_2)] - c = 0$$

divide the two F.O.C , we get

$$\frac{e_3}{e_1} = \lambda_1 > 1$$

which contradicts the condition that $e_3 \leq e_1$

(3) case 3: $\lambda_1 e_1^{\alpha} \leq e_3^{\alpha}$ and $e_3 \geq e_1$

Player 1
$$max \quad \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_3^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_1$$

Player 3 $max \quad \left(1 - \frac{1}{2} \frac{e_1^{\alpha}}{e_3^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_3$

F.o.c

$$[e_1] \quad \frac{\alpha\lambda_1}{2} \frac{e_1^{\alpha-1}}{e_3^{\alpha}} [u(w_1) - u(w_2)] - c = 0$$

$$[e_3] \quad \frac{\alpha}{2} \frac{e_1^{\alpha}}{e_3^{\alpha+1}} [u(w_1) - u(w_2)] - c = 0$$

divide the two F.O.C , we get

$$\frac{e_3}{e_1} = \frac{1}{\lambda_1} < 1$$

which contradicts the condition that $e_3 \ge e_1$

Thus the unique equilibrium is

$$e_1^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} [u(w_1) - u(w_2)]$$

$$e_3^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}} [u(w_1) - u(w_2)]$$

where $\lambda_1 e_1^{\alpha} > e_3^{\alpha}$ and $e_3 > e_1$.

We show that $e_1^f < \overline{e}^f$ and $e_3^f < \overline{e}^f$:

$$e_{1}^{f} < \overline{e}^{f} \iff \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})] < \frac{\alpha}{2c} [u(w_{1}) - u(w_{2})]$$
$$\iff \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} < 1$$
$$e_{3}^{f} < \overline{e}^{f} \iff \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})] < \frac{\alpha}{2c} [u(w_{1}) - u(w_{2})]$$
$$\iff \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} < 1$$

2. equilibrium winning probabilities

The true winning probabilities are

$$p_{13}^{f} = \frac{1}{2} \left(\frac{e_1^{f}}{e_3^{f}} \right)^{\alpha}$$
$$= \frac{1}{2} \left(\lambda_1^{-\frac{1}{2\alpha+1}} \right)^{\alpha}$$
$$= \frac{1}{2} \lambda_1^{-\frac{\alpha}{2\alpha+1}}$$

$$p_{31}^f = 1 - p_{13}^f = 1 - \frac{1}{2}\lambda_1^{-\frac{\alpha}{2\alpha+1}}$$

The overconfident player 1's perceived winning probabilities are

$$\begin{split} \widetilde{p}_{13}^{f} &= 1 - \frac{1}{2} \frac{(e_{3}^{f})^{\alpha}}{\lambda_{1}(e_{1}^{f})^{\alpha}} \\ &= 1 - \frac{1}{2} \frac{\left(\frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})]\right)^{\alpha}}{\lambda_{1} \left(\frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})]\right)^{\alpha}} \\ &= 1 - \frac{1}{2} \frac{\left(\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\right)^{\alpha}}{\lambda_{1} \left(\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\alpha}} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \end{split}$$

We show that $\widetilde{p}_{13}^f > p_{31}^f > \frac{1}{2} > p_{13}^f$:

$$\begin{split} \widetilde{p}_{13}^{f} > p_{31}^{f} & \Longleftrightarrow 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} > 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \\ & \Leftrightarrow \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} > \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \\ & \Leftrightarrow \lambda_{1}^{\frac{1}{2\alpha+1}} > 1 \\ p_{31}^{f} > \frac{1}{2} & \Longleftrightarrow 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} > \frac{1}{2} \\ & \Leftrightarrow \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} < 1 \\ p_{13}^{f} < \frac{1}{2} & \Longleftrightarrow 1 - p_{31}^{f} < \frac{1}{2} \\ & \Leftrightarrow p_{31}^{f} > \frac{1}{2} \end{split}$$

3. expected utilities of final

$$\widetilde{E}^{f}(U_{13}) = \widetilde{p}_{13}^{f} u(w_{1}) + (1 - \widetilde{p}_{13}^{f}) u(w_{2}) - ce_{1}^{f}$$

$$= \left(1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right) u(w_{1}) + \frac{1}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} u(w_{2}) - \frac{\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})]$$

$$= u(w_{1}) - \frac{1 + \alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})]$$

$$E^{f}(U_{31}) = p_{31}^{f}u(w_{1}) + (1 - p_{31}^{f})u(w_{2}) - ce_{3}^{f}$$

= $\left(1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\right)u(w_{1}) + \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}u(w_{2}) - \frac{\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})]$
= $u(w_{1}) - \frac{1 + \alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})]$

We show that $\widetilde{E}^{f}(U_{13}) > E^{f}(U_{31}) > \overline{E}^{f}(U)$:

$$\begin{split} \widetilde{E}^{f}(U_{13}) &> E^{f}(U_{31}) \\ \iff u(w_{1}) - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}[u(w_{1}) - u(w_{2})] > u(w_{1}) - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})] \\ \iff -\frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}[u(w_{1}) - u(w_{2})] > -\frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})] \\ \iff \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} > \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \\ E^{f}(U_{31}) > \overline{E}^{f}(U) \\ \iff u(w_{1}) - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})] > u(w_{1}) - \frac{1+\alpha}{2}[u(w_{1}) - u(w_{2})] \\ \iff -\frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}[u(w_{1}) - u(w_{2})] > -\frac{1+\alpha}{2}[u(w_{1}) - u(w_{2})] \\ \iff 1 > \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \end{split}$$

Since $\overline{E}^{f}(U) \ge 0$, the participation constraints of both players are satisfied.

Proof of Proposition 3

1. Continuation values

Using Proposition 2, we can get the continuation values of each player.

Overconfident player 1:

$$\widetilde{v}_1 = p_{34}^s \widetilde{E}^f(U_{13}) + p_{43}^s \widetilde{E}^f(U_{14})$$

Since player 3 and 4 are identical, $\widetilde{E}^{f}(U_{13}) = \widetilde{E}^{f}(U_{14})$

$$\widetilde{v}_1 = u(w_1) - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}[u(w_1) - u(w_2)]$$

Since $\widetilde{E}^f(U_{13}) > \overline{E}^f(U)$, we can get $\widetilde{v}_1 > \overline{v}$.

Rational player 2:

$$v_2 = p_{34}^s E^f(U_{23}) + p_{43}^s E^f(U_{24})$$

Since players 3 and 4 are identical, $E^{f}(U_{23}) = E^{f}(U_{24})$

$$v_2 = \frac{1-\alpha}{2}u(w_1) + \frac{1+\alpha}{2}u(w_2) = \overline{v}$$

2. The equilibrium efforts and winning probabilities

Player 1 max
$$\widetilde{E}^s(U_{12}) = \widetilde{p}_{12}^s \widetilde{v}_1 - ce_1$$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right) \widetilde{v}_1 - ce_1 & \text{if } \lambda_1 e_1^{\alpha} \ge e_2^{\alpha} \\ \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_2^{\alpha}} \widetilde{v}_1 - ce_1 & \text{if } \lambda_1 e_1^{\alpha} \le e_2^{\alpha} \end{cases}$$

Player 2 max
$$E^{s}(U_{21}) = p_{21}^{s}v_{2} - ce_{2}$$

$$= \begin{cases} \left(1 - \frac{1}{2}\frac{e_{1}^{\alpha}}{e_{2}^{\alpha}}\right)v_{2} - ce_{2} & \text{if } e_{2} \ge e_{1} \\ \frac{1}{2}\frac{e_{2}^{\alpha}}{e_{1}^{\alpha}}v_{2} - ce_{2} & \text{if } e_{2} \le e_{1} \end{cases}$$

There are 4 cases.

 $\begin{cases} \lambda_1 e_1^{\alpha} \geqslant e_2^{\alpha} & and \quad e_2 \leqslant e_1 \\ \lambda_1 e_1^{\alpha} \geqslant e_2^{\alpha} & and \quad e_2 \geqslant e_1 \\ \lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha} & and \quad e_2 \geqslant e_1 \\ \lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha} & and \quad e_2 \leqslant e_1 \end{cases}$

Since $\lambda_1 > 1$, the fourth case is impossible.

(1) case 1: $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_2 \le e_1$, which corresponds to proposition 3 (i).

Player 1
$$max \left(1 - \frac{1}{2}\frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right)\widetilde{v}_1 - ce_1$$

Player 2 $max \quad \frac{1}{2}\frac{e_2^{\alpha}}{e_1^{\alpha}}v_2 - ce_2$

F.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}} \widetilde{v}_1 - c = 0$$
$$[e_2] \quad \frac{\alpha}{2} \frac{e_2^{\alpha-1}}{e_1^{\alpha}} v_2 - c = 0$$

S.o.c

 $[e_1] \quad \frac{\alpha}{2\lambda_1}(-\alpha-1)\frac{e_2^{\alpha}}{e_1^{\alpha+2}}\widetilde{v}_1 < 0$ $[e_2] \quad \frac{\alpha}{2}(\alpha-1)\frac{e_2^{\alpha-2}}{e_1^{\alpha}}v_2 < 0$

Solve the two F.O.C , we get

$$e_1 = \frac{\alpha}{2c} \lambda_1^{\alpha - 1} (\widetilde{v}_1)^{1 - \alpha} (v_2)^{\alpha}$$
$$e_2 = \frac{\alpha}{2c} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{\alpha + 1}$$
$$\frac{e_2}{e_1} = \lambda_1 \frac{v_2}{\widetilde{v}_1}$$

Check the conditions $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_2 \le e_1$: (1) $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$

As long as $e_1 \ge e_2$ is satisfied, $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ is satisfied.

(2) $e_2 \leqslant e_1$

$$e_{1} \ge e_{2} \iff \frac{e_{1}}{e_{2}} \ge 1$$

$$\iff \lambda_{1}^{-\frac{1}{2\alpha+1}} (\tilde{v}_{1})^{\frac{1}{2\alpha+1}} (v_{2})^{-\frac{1}{2\alpha+1}} \ge 1$$

$$\iff \left(\frac{\tilde{v}_{1}}{\lambda_{1}v_{2}}\right)^{\frac{1}{2\alpha+1}} \ge 1$$

$$\iff \frac{\tilde{v}_{1}}{\lambda_{1}v_{2}} \ge 1$$

$$\iff \frac{\left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right) [u(w_{1}) - u(w_{2})]}{\lambda_{1} \left[\left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\right) [u(w_{1}) - u(w_{2})] \right]} \ge 1$$

$$\iff \frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \ge \lambda_{1} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) = \lambda_{1} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) \le \lambda_{1} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{u(w_{1})}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) \le \lambda_{1} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{u(w_{1})}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) \le \lambda_{1} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{u(w_{1})}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)$$

Let

$$f(\lambda_1) = \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}\right) - \lambda_1\left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\right)$$

we can easily get that $f(\lambda_1 = 1) = 0$ and $f(\lambda_1 \to \infty) < 0$.

$$f'(\lambda_1) = \frac{(1+\alpha)^2}{2(2\alpha+1)} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}-1} - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\right)$$

$$f'(\lambda_{1}) \stackrel{\leq}{=} 0 \iff \frac{(1+\alpha)^{2}}{2(2\alpha+1)} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}-1} \stackrel{\leq}{\leq} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\right)$$
$$\iff \frac{(1+\alpha)^{2}}{2\alpha+1} \frac{u(w_{1})-u(w_{2})}{(1-\alpha)u(w_{1})+(1+\alpha)u(w_{2})} \stackrel{\leq}{\leq} \lambda_{1}^{\frac{\alpha+1}{2\alpha+1}+1}$$
$$\iff \left[\frac{(1+\alpha)^{2}}{2\alpha+1} \frac{u(w_{1})-u(w_{2})}{(1-\alpha)u(w_{1})+(1+\alpha)u(w_{2})}\right]^{\frac{1}{2\alpha+1}+1} \stackrel{\leq}{\leq} \lambda_{1}$$
Let $g(\alpha) = \left[\frac{(1+\alpha)^{2}}{2\alpha+1} \frac{u(w_{1})-u(w_{2})}{(1-\alpha)u(w_{1})+(1+\alpha)u(w_{2})}\right]^{\frac{1}{2\alpha+1}+1}$ a) $g(\alpha) \leqslant 1$

if $g(\alpha) \leq 1$, then $f'(\lambda_1) < 0$ always holds. Which means $f(\lambda_1) < 0$ always holds, thus $\frac{e_1}{e_2} < 1$ always holds.

$$\begin{split} g(\alpha) &\leqslant 1 \Longleftrightarrow \frac{(1+\alpha)^2}{2\alpha+1} \frac{u(w_1) - u(w_2)}{(1-\alpha)u(w_1) + (1+\alpha)u(w_2)} \leqslant 1 \\ &\iff \frac{(1+\alpha)^2}{2\alpha+1} \leqslant \frac{(1-\alpha)u(w_1) + (1+\alpha)u(w_2)}{u(w_1) - u(w_2)} \\ &\iff (1+\alpha)^2 [u(w_1) - u(w_2)] \leqslant (2\alpha+1)[(1-\alpha)u(w_1) + (1+\alpha)u(w_2)] \\ &\iff (\alpha+3\alpha^2)u(w_1) \leqslant (2+5\alpha+3\alpha^2)u(w_2) \\ &\iff \frac{u(w_1)}{u(w_2)} \leqslant \frac{2+5\alpha+3\alpha^2}{\alpha(1+3\alpha)} \\ &\iff \frac{u(w_1)}{u(w_2)} - 1 \leqslant \frac{2+5\alpha+3\alpha^2}{\alpha(1+3\alpha)} - 1 \\ &\iff \frac{u(w_1) - u(w_2)}{u(w_2)} \leqslant \frac{2(1+2\alpha)}{\alpha(1+3\alpha)} \end{split}$$

When $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$, the condition $e_1 \geq e_2$ is never satisfied given that $\lambda_1 > 1$.

b) $g(\alpha) > 1$

if
$$g(\alpha) > 1$$
, then

$$f'(\lambda_1) \begin{cases} > 0 \quad \text{when } \lambda_1 < \left[\frac{(1+\alpha)^2}{2\alpha+1} \frac{u(w_1) - u(w_2)}{(1-\alpha)u(w_1) + (1+\alpha)u(w_2)}\right]^{\frac{1}{\frac{\alpha+1}{2\alpha+1}+1}} \\ < 0 \quad \text{when } \lambda_1 > \left[\frac{(1+\alpha)^2}{2\alpha+1} \frac{u(w_1) - u(w_2)}{(1-\alpha)u(w_1) + (1+\alpha)u(w_2)}\right]^{\frac{\alpha}{\alpha+1}+1} \end{cases}$$

We now show that if $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$, then there exists a unique threshold $\hat{\lambda} > 1$ where $f(\lambda_1) = 0$, that is,

$$\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} = \lambda_1 \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)$$

which is equivalent to

$$u(w_1) - \frac{1+\alpha}{2}\hat{\lambda}^{-\frac{\alpha+1}{2\alpha+1}}[u(w_1) - u(w_2)] = \hat{\lambda}\left[\left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\right)[u(w_1) - u(w_2)]\right]$$

To see this is the case, we rearrange the equality as

$$u(w_1) - \frac{1+\alpha}{2}\hat{\lambda}^{-\frac{\alpha+1}{2\alpha+1}}[u(w_1) - u(w_2)] = \hat{\lambda}\left[u(w_1) - \frac{1+\alpha}{2}[u(w_1) - u(w_2)]\right],$$

or

or

$$\frac{1+\alpha}{2} [\hat{\lambda} - \hat{\lambda}^{-\frac{\alpha+1}{2\alpha+1}}][u(w_1) - u(w_2)] = (\hat{\lambda} - 1)u(w_1),$$
$$\frac{1+\alpha}{2} \frac{u(w_1) - u(w_2)}{u(w_1)} = \frac{\hat{\lambda} - 1}{\hat{\lambda} - \hat{\lambda}^{-\frac{\alpha+1}{2\alpha+1}}}.$$
(17)

Since $\alpha \in (0, 1]$ and $u(w_1) > u(w_2)$, the left-hand side of (17) takes a value in the interval (0, 1). The right-hand side of (17) is increasing in $\hat{\lambda}$ for $\lambda > 1$, its limit when $\hat{\lambda} \to 1$ is $\frac{2\alpha+1}{3\alpha+2}$, and its limit when $\hat{\lambda} \to \infty$ is 1.

Hence, the threshold λ exists and is unique provided that

$$\frac{1+\alpha}{2}\frac{u(w_1)-u(w_2)}{u(w_1)} > \frac{2\alpha+1}{3\alpha+2}$$

It is easy to show that this inequality is equivalent to

$$\frac{u(w_1) - u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}.$$

Therefore, if $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$, then there exists a unique value for $\hat{\lambda}$, greater than 1, that satisfies (17). This, in turn, implies:

$$f(\lambda_1) \begin{cases} > 0 & \text{when } \lambda_1 < \hat{\lambda} \\ = 0 & \text{when } \lambda_1 = \hat{\lambda} \\ < 0 & \text{when } \lambda_1 > \hat{\lambda} \end{cases}$$
$$e_1 - e_2 \begin{cases} > 0 & \text{when } \lambda_1 < \hat{\lambda} \\ = 0 & \text{when } \lambda_1 = \hat{\lambda} \\ < 0 & \text{when } \lambda_1 > \hat{\lambda} \end{cases}$$

The condition $e_1 \ge e_2$ is only satisfied when $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 \le \hat{\lambda}$. And $e_1 > e_2$ is only satisfied when $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 < \hat{\lambda}$.

Therefore the solution

$$e_1 = \frac{\alpha}{2c} \lambda_1^{\alpha - 1} (\widetilde{v}_1)^{1 - \alpha} (v_2)^{\alpha}$$
$$e_2 = \frac{\alpha}{2c} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{\alpha + 1}$$

only applies when $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 < \hat{\lambda}$. (2) case 2: $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_2 \ge e_1$, which corresponds to proposition 3 (ii).

Player 1
$$max \left(1 - \frac{1}{2}\frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right)\widetilde{v}_1 - ce_1$$

Player 2 $max \left(1 - \frac{1}{2}\frac{e_1^{\alpha}}{e_2^{\alpha}}\right)v_2 - ce_2$

F.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}} \widetilde{v}_1 - c = 0$$
$$[e_2] \quad \frac{\alpha}{2} \frac{e_1^{\alpha}}{e_2^{\alpha+1}} v_2 - c = 0$$

S.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1}(-\alpha-1)\frac{e_2^{\alpha}}{e_1^{\alpha+2}}\widetilde{v}_1 < 0$$

$$[e_2] \quad \frac{\alpha}{2}(-\alpha - 1)\frac{e_1^{\alpha}}{e_2^{\alpha + 2}}v_2 < 0$$

Solve F.O.C , we get

$$e_{1} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha+1}{2\alpha+1}} (v_{2})^{\frac{\alpha}{2\alpha+1}}$$
$$e_{2} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (v_{2})^{\frac{\alpha+1}{2\alpha+1}}$$
$$\frac{e_{2}}{e_{1}} = \lambda_{1}^{\frac{1}{2\alpha+1}} (\widetilde{v}_{1})^{-\frac{1}{2\alpha+1}} (v_{2})^{\frac{1}{2\alpha+1}}$$

Check the conditions $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_2 \ge e_1$: (1) $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$

$$\lambda_1 e_1^{\alpha} \ge e_2^{\alpha} \iff \frac{\lambda_1 e_1^{\alpha}}{e_2^{\alpha}} \ge 1$$
$$\iff \lambda_1^{\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_1)^{\frac{\alpha}{2\alpha+1}} (v_2)^{-\frac{\alpha}{2\alpha+1}} \ge 1$$

Since $\lambda_1 > 1$ and $\tilde{v}_1 > v_2$, the inequality is always satisfied. Therefore $\lambda_1 e_1^{\alpha} > e_2^{\alpha}$ always holds when $\lambda_1 > 1$.

(2) $e_2 \ge e_1$ We have already seen in case (1) that $e_2 \ge e_1$ is satisfied when either $\frac{u(w_1)-u(w_2)}{u(w_2)} \le \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \ge \hat{\lambda}$.

Therefore the solution

$$e_1 = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_1)^{\frac{\alpha+1}{2\alpha+1}} (v_2)^{\frac{\alpha}{2\alpha+1}}$$
$$e_2 = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_1)^{\frac{\alpha}{2\alpha+1}} (v_2)^{\frac{\alpha+1}{2\alpha+1}}$$

only applies when either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \geq \hat{\lambda}$ (3) case 3: $\lambda_1 e_1^{\alpha} \leq e_2^{\alpha}$ and $e_2 \geq e_1$

Player 1 max $\frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e^{\alpha}} \widetilde{v}_1 - ce_1$

Player 2
$$max \quad [1 - \frac{1}{2}(\frac{e_1}{e_2})^{\alpha}]v_2 - ce_2$$

F.o.c

$$\begin{bmatrix} e_1 \end{bmatrix} \quad \frac{\alpha \lambda_1}{2} \frac{e_1^{\alpha-1}}{e_2^{\alpha}} \widetilde{v}_1 - c = 0$$
$$\begin{bmatrix} e_2 \end{bmatrix} \quad \frac{\alpha}{2} \frac{e_1^{\alpha}}{e_2^{\alpha+1}} v_2 - c = 0$$

divide the two F.O.C , we get

$$\frac{e_2}{e_1} = \frac{v_2}{\lambda_1 \widetilde{v}_1} < 1$$

which contradicts the condition that $e_2 \ge e_1$

Therefore, the equilibrium in this semi-final:

(1) When $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 < \hat{\lambda}$, which corresponds to Proposition 3 (i)

$$e_{1}^{s} = \frac{\alpha}{2c} \lambda_{1}^{\alpha-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{1-\alpha} \\ \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \right)^{\alpha} [u(w_{1}) - u(w_{2})] \\ e_{2}^{s} = \frac{\alpha}{2c} \lambda_{1}^{\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \\ \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \right)^{1+\alpha} [u(w_{1}) - u(w_{2})]$$

where $e_1^s > e_2^s$.

$$p_{21}^{s} = \frac{1}{2} \left(\frac{e_{2}^{s}}{e_{1}^{s}}\right)^{\alpha}$$

= $\frac{1}{2} \left(\frac{\lambda_{1}v_{2}}{\widetilde{v}_{1}}\right)^{\alpha}$
= $\frac{1}{2} \lambda_{1}^{\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\right)^{\alpha}$

$$p_{12}^{s} = 1 - p_{21}^{s}$$
$$= 1 - \frac{1}{2}\lambda_{1}^{\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\right)^{\alpha}$$

$$\begin{split} \widetilde{p}_{12}^{s} &= 1 - \frac{1}{2} \frac{\left(e_{2}^{s}\right)^{\alpha}}{\lambda_{1} \left(e_{1}^{s}\right)^{\alpha}} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-1} \left[\lambda_{1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right) \right]^{\alpha} \\ &= 1 - \frac{1}{2} \lambda_{1}^{\alpha-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right)^{\alpha} \end{split}$$

(2) When either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \geq \hat{\lambda}$, which corresponds to Proposition 3 (ii).

$$e_1^s = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha+1}{2\alpha+1}} \\ \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{\frac{\alpha}{2\alpha+1}} \left[u(w_1) - u(w_2) \right]$$

$$e_2^s = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \\ \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{\frac{\alpha+1}{2\alpha+1}} [u(w_1) - u(w_2)]$$

where $\lambda_1 (e_1^s)^{\alpha} > (e_2^s)^{\alpha}$ and $e_1^s \leqslant e_2^s$.

$$p_{12}^{s} = \frac{1}{2} \left(\frac{e_{1}^{s}}{e_{2}^{s}} \right)^{\alpha}$$

$$= \frac{1}{2} \left[\lambda_{1}^{-\frac{1}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{1}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \right)^{-\frac{1}{2\alpha+1}} \right]^{\alpha}$$

$$= \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \right)^{-\frac{\alpha}{2\alpha+1}}$$

$$p_{21}^{s} = 1 - p_{12}^{s}$$
$$= 1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{-\frac{\alpha}{2\alpha+1}}$$

$$\begin{split} \widetilde{p}_{12}^{s} &= 1 - \frac{1}{2} \frac{\left(e_{2}^{s}\right)^{\alpha}}{\lambda_{1}\left(e_{1}^{s}\right)^{\alpha}} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-1} \lambda_{1}^{\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{\frac{\alpha}{2\alpha+1}} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{\frac{\alpha}{2\alpha+1}} \end{split}$$

3. equilibrium efforts compared to benchmark

(1) Proposition 3 (i): when $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 < \hat{\lambda}$

We show that $e_1^s > \overline{e}^s > e_2^s$:

$$e_1^s > \overline{e}^s \iff \frac{e_1^s}{\overline{e}^s} > 1$$
$$\iff \frac{\frac{\alpha}{2c}\lambda_1^{\alpha-1}\widetilde{v_1}^{1-\alpha}v_2^{\alpha}}{\frac{\alpha}{2c}\overline{v}} > 1$$
$$\iff \left(\frac{\widetilde{v}_1}{\lambda_1v_2}\right)^{1-\alpha} > 1$$

$$\frac{\frac{e_2^s}{\overline{e}^s} \iff \frac{\frac{e_2^s}{\overline{e}^s} > 1}{\bigotimes \frac{\frac{\alpha}{2c} \lambda_1^{\alpha} (\widetilde{v_1})^{-\alpha} (v_2)^{\alpha+1}}{\frac{\frac{\alpha}{2c} \overline{v}}{\widetilde{v}}} > 1$$
$$\iff \left(\frac{\lambda_1 v_2}{\widetilde{v_1}}\right)^{\alpha} > 1$$

Since $\frac{\tilde{v}_1}{\lambda_1 v_2} > 1$, we can get $e_1^s > \overline{e}^s$ and $e_2^s < \overline{e}^s$.

(2) Proposition 3 (ii): when either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \geq \hat{\lambda}$

We show that $e_1^s \leqslant e_2^s \leqslant \overline{e}^s$:

Since we already showed that $e_1^s \leq e_2^s$ is satisfied under this condition, we only have to show $e_2^s \leq \overline{e}^s$.

$$e_{2}^{s} \leqslant \overline{e}^{s} \iff \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \widetilde{v}_{1}^{\frac{\alpha}{2\alpha+1}} v_{2}^{\frac{\alpha+1}{2\alpha+1}} \leqslant \frac{\alpha}{2c} \overline{v}$$
$$\iff \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \widetilde{v}_{1}^{\frac{\alpha}{2\alpha+1}} v_{2}^{\frac{\alpha+1}{2\alpha+1}} \leqslant \frac{\alpha}{2c} v_{2}$$
$$\iff \widetilde{v}_{1}^{\frac{\alpha}{2\alpha+1}} \leqslant \lambda_{1}^{\frac{\alpha}{2\alpha+1}} v_{2}^{\frac{\alpha}{2\alpha+1}}$$
$$\iff \widetilde{v}_{1} \leqslant \lambda_{1} v_{2}$$

which always holds, thus $e_1^s \leq e_2^s \leq \overline{e}^s$ always holds.

4. perceived and true winning probabilities compared to benchmark

- (1) Proposition 3 (i): when $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ and $\lambda_1 < \hat{\lambda}$
 - We show that $p_{21}^s < \frac{1}{2}$ and $\tilde{p}_{12}^s > p_{12}^s > \frac{1}{2}$: $p_{21}^s < \frac{1}{2} \iff \frac{1}{2} \left(\frac{e_2^s}{e_1^s}\right)^{\alpha} < \frac{1}{2}$ $\iff e_2^s < e_1^s$ $p_{12}^s = 1 - p_{21}^s > \frac{1}{2}$ $\tilde{p}_{12}^s > p_{12}^s \iff 1 - \frac{1}{2} \left(\frac{e_2^s}{\lambda_1 e_1^s}\right)^{\alpha} > 1 - \frac{1}{2} \left(\frac{e_2^s}{e_1^s}\right)^{\alpha}$ $\iff \lambda_1 > 1$

(2) Proposition 3 (ii): when either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \geq \hat{\lambda}$ We show that $p_{12}^s \leq \frac{1}{2}, p_{21}^s \geq \frac{1}{2}$ and $\tilde{p}_{12}^s > \frac{1}{2}$:

$$p_{12}^{s} \leqslant \frac{1}{2} \Longleftrightarrow \frac{1}{2} \left(\frac{e_{1}^{s}}{e_{2}^{s}}\right)^{\alpha} \leqslant \frac{1}{2}$$
$$\iff e_{1}^{s} \leqslant e_{2}^{s}$$

$$p_{21}^s = 1 - p_{12}^s \geqslant \frac{1}{2}$$

$$\begin{split} \widetilde{p}_{12}^{s} &> \frac{1}{2} \Longleftrightarrow 1 - \frac{1}{2} \left(\frac{e_{2}^{s}}{\lambda_{1} e_{1}^{s}} \right)^{\alpha} > \frac{1}{2} \\ &\iff \frac{e_{2}^{s}}{\lambda_{1} e_{1}^{s}} < 1 \\ &\iff \frac{\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2} \right)^{\frac{\alpha+1}{2\alpha+1}}}{\lambda_{1}^{\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha+1}{2\alpha+1}} \left(v_{2} \right)^{\frac{\alpha}{2\alpha+1}}} < 1 \\ &\iff \lambda_{1}^{-\frac{2\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{-\frac{1}{2\alpha+1}} \left(v_{2} \right)^{\frac{1}{2\alpha+1}} < 1 \\ &\iff \lambda_{1}^{-\frac{2\alpha}{2\alpha+1}} \left(\frac{v_{2}}{\widetilde{v}_{1}} \right)^{\frac{1}{2\alpha+1}} < 1 \end{split}$$

5. Participation constraints

(1) When
$$\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$$
 and $\lambda_1 < \hat{\lambda}$
 $\widetilde{E}^s(U_{12}) = \widetilde{p}_{12}^s \widetilde{v}_1 - ce_1^s$
 $> p_{12}^s \widetilde{v}_1 - ce_1^s$
 $= \left(1 - \frac{1}{2}\lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{\alpha}\right) \widetilde{v}_1 - c\frac{\alpha}{2c}\lambda_1^{\alpha-1} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha}$
 $= \widetilde{v}_1 - \frac{1}{2}\lambda_1^{\alpha} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha} - \frac{\alpha}{2}\lambda_1^{\alpha-1} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha}$
 $> \widetilde{v}_1 - \frac{1}{2}\lambda_1^{\alpha} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha} - \frac{1}{2}\lambda_1^{\alpha-1} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha}$
 $> \widetilde{v}_1 - \frac{1}{2}\lambda_1^{\alpha} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha} - \frac{1}{2}\lambda_1^{\alpha} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha}$
 $= \widetilde{v}_1 - \lambda_1^{\alpha} (\widetilde{v}_1)^{1-\alpha} (v_2)^{\alpha}$
 $= (\widetilde{v}_1)^{1-\alpha} \left[(\widetilde{v}_1)^{\alpha} - \lambda_1^{\alpha} (v_2)^{\alpha} \right]$
 > 0

$$E^{s}(U_{21}) = p_{21}^{s} v_{2} - c e_{2}^{s}$$

$$= \frac{1}{2} \lambda_{1}^{\alpha} (\widetilde{v}_{1})^{-\alpha} (v_{2})^{\alpha} v_{2} - c \frac{\alpha}{2c} \lambda_{1}^{\alpha} (\widetilde{v}_{1})^{-\alpha} (v_{2})^{\alpha+1}$$

$$= \frac{1}{2} \lambda_{1}^{\alpha} (\widetilde{v}_{1})^{-\alpha} (v_{2})^{1+\alpha} - \frac{\alpha}{2} \lambda_{1}^{\alpha} (\widetilde{v}_{1})^{-\alpha} (v_{2})^{\alpha+1}$$

$$= \frac{1-\alpha}{2} \lambda_{1}^{\alpha} (\widetilde{v}_{1})^{-\alpha} (v_{2})^{1+\alpha}$$

$$\geq 0$$

(2) When either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \geq \hat{\lambda}$

$$E^s(U_{12}) = \widetilde{p}_{12}^s \widetilde{v}_1 - c e_1^s$$

Since $\widetilde{p}_{12}^s > \frac{1}{2}$, $\widetilde{v}_1 > \overline{v}$ and $e_1^s < \overline{e}^s$, we can get that $\widetilde{E}^s(U_{12}) > \overline{E}^s(U) \ge 0$.

$$\begin{split} E^{s}(U_{21}) &= p_{21}^{s} v_{2} - c e_{2}^{s} \\ &= \left(1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1}\right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2}\right)^{-\frac{\alpha}{2\alpha+1}}\right) v_{2} - c \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1}\right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2}\right)^{\frac{\alpha+1}{2\alpha+1}} \\ &= v_{2} - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1}\right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2}\right)^{\frac{\alpha+1}{2\alpha+1}} - \frac{\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1}\right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2}\right)^{\frac{\alpha+1}{2\alpha+1}} \\ &= v_{2} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1}\right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2}\right)^{\frac{\alpha+1}{2\alpha+1}} \\ &= \left(v_{2}\right)^{\frac{\alpha+1}{2\alpha+1}} \left(\left(v_{2}\right)^{\frac{\alpha}{2\alpha+1}} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1}\right)^{\frac{\alpha}{2\alpha+1}}\right) \\ &\geqslant 0 \end{split}$$

Derivation of equation (7)

The two FOCs in the semifinal are

$$\frac{\partial \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1} \widetilde{v}_1(\lambda_1) = c$$
$$\frac{\partial p_{21}^s(e_1, e_2)}{\partial e_2} v_2 = c$$

Total differentiation gives us

$$\begin{pmatrix} \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1^2} de_1 + \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial e_2} de_2 + \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial \lambda_1} d\lambda_1 \end{pmatrix} \widetilde{v}_1(\lambda_1)$$

$$+ \frac{\partial \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1} \frac{\partial \widetilde{v}_1(\lambda_1)}{\partial \lambda_1} d\lambda_1 = 0$$

$$\begin{pmatrix} \frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2 \partial e_1} de_1 + \frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2^2} de_2 \end{pmatrix} v_2 = 0$$

or

$$\begin{pmatrix} \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1^2} \frac{de_1}{d\lambda_1} + \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial e_2} \frac{de_2}{d\lambda_1} + \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial \lambda_1} \end{pmatrix} \widetilde{v}_1(\lambda_1) + \frac{\partial \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1} \frac{\partial \widetilde{v}_1(\lambda_1)}{\partial \lambda_1} = 0$$

$$\begin{pmatrix} \frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2 \partial e_1} \frac{de_1}{d\lambda_1} + \frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2^2} \frac{de_2}{d\lambda_1} \end{pmatrix} v_2 = 0$$
Solving the second equation for $\frac{de_2}{de_2}$ we obtain

Solving the second equation for $\frac{de_2}{d\lambda_1}$ we obtain $\frac{\partial^2 n^3}{\partial x_1}$

$$\frac{de_2}{d\lambda_1} = -\frac{\frac{\partial^2 p_{21}^2(e_1, e_2)}{\partial e_2 \partial e_1}}{\frac{\partial^2 p_{21}^2(e_1, e_2)}{\partial e_2^2}} \frac{de_1}{d\lambda_1}$$

Replacing in the first equation we obtain

$$\left(\frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1^2} \frac{de_1}{d\lambda_1} - \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial e_2} \frac{\frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2 \partial e_1}}{\partial e_2^2} \frac{de_1}{d\lambda_1} + \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial \lambda_1}\right) \widetilde{v}_1(\lambda_1) + \frac{\partial \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1} \frac{\partial \widetilde{v}_1(\lambda_1)}{\partial \lambda_1} = 0$$

Let's solve this equation for $\frac{de_1}{d\lambda_1}$

$$\frac{de_1}{d\lambda_1} \left(\frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1^2} - \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial e_2} \frac{\frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2 \partial e_1}}{\partial e_2^2} \right) \widetilde{v}_1(\lambda_1) \\
= -\frac{\partial \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1} \frac{\partial \widetilde{v}_1(\lambda_1)}{\partial \lambda_1} - \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial \lambda_1} \widetilde{v}_1(\lambda_1)$$

or

$$\begin{split} \frac{de_1}{d\lambda_1} &= -\frac{\frac{\partial^2 \widetilde{p}_{12}^s(e_1,e_2,\lambda_1)}{\partial e_1 \partial \lambda_1} \widetilde{v}_1(\lambda_1) + \frac{\partial \widetilde{p}_{12}^s(e_1,e_2,\lambda_1)}{\partial e_1} \frac{\partial \widetilde{v}_1(\lambda_1)}{\partial \lambda_1}}{\left(\frac{\partial^2 \widetilde{p}_{12}^s(e_1,e_2,\lambda_1)}{\partial e_1^2} - \frac{\partial^2 \widetilde{p}_{12}^s(e_1,e_2,\lambda_1)}{\partial e_1 \partial e_2} \frac{\partial^2 \widetilde{p}_{21}^s(e_1,e_2)}{\frac{\partial^2 p_{21}^s(e_1,e_2)}{\partial e_2^2}}\right) \widetilde{v}_1(\lambda_1)} \\ &= -\frac{\frac{\partial mg \widetilde{p}_{12}^s}{\partial \lambda_1} \widetilde{v}_1(\lambda_1) + mg \widetilde{p}_{12}^s \frac{\partial \widetilde{v}_1(\lambda_1)}{\partial \lambda_1}}{\left(\frac{\partial mg \widetilde{p}_{12}^s}{\partial e_1} - \frac{\partial mg \widetilde{p}_{12}^s}{\partial e_2} \frac{\frac{\partial mg p_{21}^s}{\partial e_1}}{\partial e_2}}\right) \widetilde{v}_1(\lambda_1) \end{split}$$

The denominator has to be negative, that is,

$$D^{s} = \frac{\partial mg\widetilde{p}_{12}^{s}}{\partial e_{1}} - \frac{\partial mg\widetilde{p}_{12}^{s}}{\partial e_{2}} \frac{\frac{\partial mgp_{21}^{s}}{\partial e_{1}}}{\frac{\partial mgp_{21}^{s}}{\partial e_{2}}} < 0$$

Note that in a semifinal where the overconfident player exerts more effort than the rational player we have $\lambda_1 e_1^{\alpha} > e_2^{\alpha}$ and $e_2 < e_1$. This implies that in the semifinal we have

$$mg\widetilde{p}_{12}^{s} = \frac{\partial \widetilde{p}_{12}^{s}(e_1, e_2, \lambda_1)}{\partial e_1} = \frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}}$$
$$mgp_{21}^{s} = \frac{\partial p_{21}^{s}(e_1, e_2)}{\partial e_2} = \frac{\alpha}{2} \frac{e_2^{\alpha-1}}{e_1^{\alpha}}$$

This implies

$$\begin{split} \frac{\partial mg\widetilde{p}_{12}^s}{\partial e_1} &= \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1^2} = -(\alpha+1)\frac{\alpha}{2\lambda_1}\frac{e_2^{\alpha}}{e_1^{\alpha+2}}\\ &\frac{\partial mg\widetilde{p}_{12}^s}{\partial e_2} = \frac{\partial^2 \widetilde{p}_{12}^s(e_1, e_2, \lambda_1)}{\partial e_1 \partial e_2} = \frac{\alpha^2}{2\lambda_1}\frac{e_2^{\alpha-1}}{e_1^{\alpha+1}}\\ &\frac{\partial mgp_{21}^s}{\partial e_2} = \frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2^2} = (\alpha-1)\frac{\alpha}{2}\frac{e_2^{\alpha-2}}{e_1^{\alpha}}\\ &\frac{\partial mgp_{21}^s}{\partial e_1} = \frac{\partial^2 p_{21}^s(e_1, e_2)}{\partial e_2 \partial e_1} = -\frac{\alpha^2}{2}\frac{e_2^{\alpha-1}}{e_1^{\alpha+1}} \end{split}$$

Hence, we have

$$\begin{split} D^{s} &= \frac{\partial mg \widetilde{p}_{12}^{s}}{\partial e_{1}} - \frac{\partial mg \widetilde{p}_{12}^{s}}{\partial e_{2}} \frac{\frac{\partial mg \widetilde{p}_{21}^{s}}{\partial e_{2}}}{\frac{\partial mg \widetilde{p}_{21}^{s}}{\partial e_{2}}} \\ &= -(\alpha+1) \frac{\alpha}{2\lambda_{1}} \frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} - \frac{\alpha^{2}}{2\lambda_{1}} \frac{e_{2}^{\alpha-1}}{e_{1}^{\alpha+1}} \frac{-\frac{\alpha^{2}}{2} \frac{e_{2}^{\alpha-1}}{e_{1}^{\alpha+1}}}{(\alpha-1)\frac{\alpha}{2} \frac{e_{2}^{\alpha-2}}{e_{1}^{\alpha}}} \\ &= -(\alpha+1) \frac{\alpha}{2\lambda_{1}} \frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} + \frac{\alpha^{3}}{2\lambda_{1}(\alpha-1)} \frac{e_{2}^{2}}{e_{1}^{\alpha+1}} \frac{\frac{e_{2}^{\alpha-1}}{e_{1}^{\alpha+1}}}{\frac{e_{2}^{\alpha-2}}{e_{1}^{\alpha}}} \\ &= -(\alpha+1) \frac{\alpha}{2\lambda_{1}} \frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} + \frac{\alpha^{3}}{2\lambda_{1}(\alpha-1)} \frac{e_{2}^{\alpha}}{e_{1}^{\alpha+1}} \frac{e_{2}^{\alpha-2}}{e_{1}^{\alpha}} \\ &= \frac{\alpha}{2\lambda_{1}} \left[-(\alpha+1) + \frac{\alpha^{2}}{(\alpha-1)} \right] \frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} \\ &= -\frac{\alpha}{2\lambda_{1}} \frac{1}{1-\alpha} \frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} \\ &= 0 \end{split}$$

In a semifinal where the overconfident player exerts less effort than the rational player we have $\lambda_1 e_1^{\alpha} > e_2^{\alpha}$ and $e_2 > e_1$. This implies that in the semifinal we have

$$mg\widetilde{p}_{12}^{s} = \frac{\partial\widetilde{p}_{12}^{s}(e_1, e_2, \lambda_1)}{\partial e_1} = \frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}}$$

$$\begin{split} mgp_{21}^{s} &= \frac{\partial p_{21}^{s}(e_{1}, e_{2})}{\partial e_{2}} = \frac{\alpha}{2} \frac{e_{1}^{\alpha}}{e_{2}^{\alpha+1}} \\ \frac{\partial mg\widetilde{p}_{12}^{s}}{\partial e_{1}} &= \frac{\partial^{2}\widetilde{p}_{12}^{s}(e_{1}, e_{2}, \lambda_{1})}{\partial e_{1}^{2}} = -(\alpha+1)\frac{\alpha}{2\lambda_{1}} \frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} \\ \frac{\partial mg\widetilde{p}_{12}^{s}}{\partial e_{2}} &= \frac{\partial^{2}\widetilde{p}_{12}^{s}(e_{1}, e_{2}, \lambda_{1})}{\partial e_{1}\partial e_{2}} = \frac{\alpha^{2}}{2\lambda_{1}} \frac{e_{2}^{\alpha-1}}{e_{1}^{\alpha+1}} \\ \frac{\partial mgp_{21}^{s}}{\partial e_{2}} &= \frac{\partial^{2}p_{21}^{s}(e_{1}, e_{2})}{\partial e_{2}^{2}} = -(\alpha+1)\frac{\alpha}{2} \frac{e_{1}^{\alpha}}{e_{2}^{\alpha+2}} \\ \frac{\partial mgp_{21}^{s}}{\partial e_{1}} &= \frac{\partial^{2}p_{21}^{s}(e_{1}, e_{2})}{\partial e_{2}\partial e_{1}} = \frac{\alpha^{2}}{2} \frac{e_{1}^{\alpha-1}}{e_{2}^{\alpha+1}} \end{split}$$

Hence, we have

$$\begin{split} D^{s} &= \frac{\partial mg\widetilde{p}_{12}^{s}}{\partial e_{1}} - \frac{\partial mg\widetilde{p}_{12}^{s}}{\partial e_{2}} \frac{\partial mg\widetilde{p}_{21}^{s}}{\partial e_{2}} \\ &= -(\alpha+1)\frac{\alpha}{2\lambda_{1}}\frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} - \frac{\alpha^{2}}{2\lambda_{1}}\frac{e_{2}^{\alpha-1}}{e_{1}^{\alpha+1}} \frac{\frac{\alpha^{2}}{2}\frac{e_{1}^{\alpha-1}}{e_{2}^{\alpha+1}}}{-(\alpha+1)\frac{\alpha}{2}\frac{e_{1}^{\alpha}}{e_{2}^{\alpha+2}}} \\ &= -(\alpha+1)\frac{\alpha}{2\lambda_{1}}\frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} + \frac{\alpha^{3}}{2\lambda_{1}(\alpha+1)}\frac{e_{2}^{\alpha-1}}{e_{1}^{\alpha+1}}\frac{\frac{e_{1}^{\alpha}}{e_{2}^{\alpha+2}}}{\frac{e_{1}^{\alpha}}{e_{2}^{\alpha+2}}} \\ &= -(\alpha+1)\frac{\alpha}{2\lambda_{1}}\frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} + \frac{\alpha^{3}}{2\lambda_{1}(\alpha+1)}\frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} \\ &= \frac{\alpha}{2\lambda_{1}}\left[-(\alpha+1)+\frac{\alpha^{2}}{(\alpha+1)}\right]\frac{e_{2}^{\alpha}}{e_{1}^{\alpha+2}} \end{split}$$

This is negative as long as

$$\alpha + 1 > \frac{\alpha^2}{\alpha + 1}$$

or

$$(\alpha + 1)^2 > \alpha$$

which is true.

Proof of Proposition 4

1. Continuation values:

Rational player 3:

$$v_{3} = p_{12}^{s} E^{f}(U_{31}) + p_{21}^{s} E^{f}(U_{32})$$

= $p_{12}^{s} \left[u(w_{1}) - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})] \right] + p_{21}^{s} \left[\frac{1-\alpha}{2} u(w_{1}) + \frac{1+\alpha}{2} u(w_{2}) \right]$

Rational player 4:

$$v_{4} = p_{12}^{s} E^{f}(U_{41}) + p_{21}^{s} E^{f}(U_{42})$$

= $p_{12}^{s} \left[u(w_{1}) - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})] \right] + p_{21}^{s} \left[\frac{1-\alpha}{2} u(w_{1}) + \frac{1+\alpha}{2} u(w_{2}) \right]$

Note that since $E^{f}(U_{31}) = E^{f}(U_{41}) > \overline{E}^{f}(U) = E^{f}(U_{32}) = E^{f}(U_{42})$, we have $v_{3} = v_{4} > \overline{v}$.

2. The equilibrium

(1) When
$$\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$$
 and $\lambda_1 < \hat{\lambda}$

$$e_3^s = e_4^s = \frac{\alpha}{2c} v_3$$

= $\frac{\alpha}{2c} \left[p_{12}^s \left[u(w_1) - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha}{2\alpha+1}} [u(w_1) - u(w_2)] \right] + p_{21}^s \left[\frac{1-\alpha}{2} u(w_1) + \frac{1+\alpha}{2} u(w_2) \right] \right]$

where

$$p_{12}^{s} = 1 - \frac{1}{2}\lambda_{1}^{\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}}\right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\right)^{\alpha}.$$
$$p_{34}^{s} = p_{43}^{s} = \frac{1}{2}$$

(2) When either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \ge \hat{\lambda}$

$$e_3^s = e_4^s = \frac{\alpha}{2c} v_3$$

= $\frac{\alpha}{2c} \left[p_{12}^s \left[u(w_1) - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha}{2\alpha+1}} [u(w_1) - u(w_2)] \right] + p_{21}^s \left[\frac{1-\alpha}{2} u(w_1) + \frac{1+\alpha}{2} u(w_2) \right] \right]$

where

$$p_{12}^{s} = \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\frac{\alpha}{2\alpha+1}} \left[\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\right]^{-\frac{\alpha}{2\alpha+1}} p_{34}^{s} = p_{43}^{s} = \frac{1}{2}$$

We show that $e_3^s = e_4^s > \overline{e}^s$ holds in both (1) and (2):

$$e_3^s = e_4^s > \overline{e}^s \Longleftrightarrow \frac{\alpha}{2c} v_3 > \frac{\alpha}{2c} \overline{v}$$
$$\iff v_3 > \overline{v}$$

3. Participation constraint

$$E^{s}(U_{34}) = p_{34}^{s}v_{3} - ce_{3}^{s} = \frac{1}{2}v_{3} - c\frac{\alpha}{2c}v_{3}$$
$$= \frac{1-\alpha}{2}v_{3}$$
$$\geqslant 0$$

$$E^s(U_{43}) = E^s(U_{34}) \ge 0$$

Proof of Proposition 5

1. When
$$\frac{u(w_1) - u(w_2)}{u(w_2)} > \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$$
 and $\lambda_1 < \hat{\lambda}$
(1) P_1
 $P_1 = p_{13}^f p_{12}^s$
 $= \frac{1}{2} \lambda_1^{-\frac{\alpha}{2\alpha+1}} \left[1 - \frac{1}{2} \lambda_1^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{\alpha} \right]$
Let

$$f(\lambda_1) = P_1 - \frac{1}{4}.$$

We can get

$$f(\lambda_1 = 1) = \frac{1}{2} \times \frac{1}{2} - \frac{1}{4} = 0$$
$$f(\lambda_1 = \hat{\lambda}) = \frac{1}{2}\hat{\lambda}^{-\frac{\alpha}{2\alpha+1}} \times \frac{1}{2} - \frac{1}{4} < 0$$

 $f(\lambda_1)$ can also be written as the following:

$$f(\lambda_1) = \frac{1}{2}\lambda_1^{-\frac{\alpha}{2\alpha+1}} -\frac{1}{4}\lambda_1^{\alpha-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\right)^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} - \frac{1}{4}\lambda_1^{\alpha-\frac{\alpha}{2\alpha+1}} - \frac{1}{4}\lambda_1^{\alpha-\frac{\alpha}{$$

Taking derivative of $f(\lambda_1)$ we obtain

$$f'(\lambda_1) = -\frac{1}{2} \frac{\alpha}{2\alpha+1} \lambda_1^{-\frac{\alpha}{2\alpha+1}-1} - \frac{1}{4} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{\alpha} \\ \left[\left(\alpha - \frac{\alpha}{2\alpha+1} \right) \lambda_1^{\alpha - \frac{\alpha}{2\alpha+1}-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} + \lambda_1^{\alpha - \frac{\alpha}{2\alpha+1}} (-\alpha) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha-1} \\ \left(-\frac{\alpha+1}{2} \right) \left(-\frac{\alpha+1}{2\alpha+1} \right) \lambda_1^{-\frac{\alpha+1}{2\alpha+1}-1} \right]$$

$$f'(\lambda_{1}=1) = -\frac{1}{2}\frac{\alpha}{2\alpha+1} - \frac{1}{4}\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{\alpha} \left[\left(\alpha - \frac{\alpha}{2\alpha+1}\right)\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{-\alpha} - \alpha\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{-\alpha-1}\frac{\alpha+1}{2}\frac{\alpha+1}{2\alpha+1}\right]$$

$$= -\frac{1}{2}\frac{\alpha}{2\alpha+1} - \frac{1}{4}\left[\alpha - \frac{\alpha}{2\alpha+1} - \alpha\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{-1}\frac{\alpha+1}{2}\frac{\alpha+1}{2\alpha+1}\right]$$

$$= -\frac{1}{2}\frac{\alpha}{2\alpha+1} - \frac{1}{4}\left[\alpha\left(\frac{2\alpha}{2\alpha+1} - \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{-1}\frac{\alpha+1}{2}\frac{\alpha+1}{2\alpha+1}\right)\right]$$

$$= -\frac{1}{2}\frac{\alpha}{2\alpha+1} - \frac{1}{4}\left[\frac{\alpha}{2\alpha-1}\left(2\alpha - \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{-1}\frac{\alpha+1}{2}(1+\alpha)\right)\right]$$

$$= -\frac{1}{2}\frac{\alpha}{2\alpha+1}\left[1 + \frac{1}{2}\left(2\alpha - \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{-1}\frac{\alpha+1}{2}(1+\alpha)\right)\right]$$

 $f'(\lambda_1 = 1)$ and $1 + \frac{1}{2} \left(2\alpha - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} \frac{\alpha + 1}{2} (1 + \alpha) \right)$ has the opposite sign.

When $1 + \frac{1}{2} \left(2\alpha - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} \frac{\alpha + 1}{2} (1 + \alpha) \right) < 0, \ f'(\lambda_1 = 1) > 0.$ And since

$$\begin{aligned} 1 + \frac{1}{2} \left(2\alpha - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{-1} \frac{\alpha+1}{2} (1+\alpha) \right) < 0 \\ \iff 2\alpha - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{-1} \frac{\alpha+1}{2} (1+\alpha) < -2 \\ \iff 2\alpha + 2 < \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{-1} \frac{\alpha+1}{2} (1+\alpha) \\ \iff 2 < \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right)^{-1} \frac{\alpha+1}{2} \\ \iff 4 \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right) < 1+\alpha \\ \iff 4 + \frac{4u(w_2)}{u(w_1) - u(w_2)} < 3 (1+\alpha) \\ \iff \frac{4u(w_2)}{u(w_1) - u(w_2)} < 3 (1+\alpha) - 4 \\ \iff \frac{4u(w_2)}{u(w_1) - u(w_2)} < 3\alpha - 1 \end{aligned}$$

 $f'(\lambda_1 = 1) > 0$ is only satisfied when $\alpha > \frac{1}{3}$ and $\frac{u(w_1) - u(w_2)}{u(w_2)} > \frac{4}{3\alpha - 1}$.

We show $\frac{4}{3\alpha-1} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$:

$$\frac{4}{3\alpha - 1} > \frac{2(2\alpha + 1)}{\alpha(3\alpha + 1)} \iff 4\alpha (3\alpha + 1) > 2(2\alpha + 1) (3\alpha - 1)$$
$$\iff 12\alpha^2 + 4\alpha > 12\alpha^2 + 2\alpha - 2$$
$$\iff 2\alpha + 2 > 0$$

Thus we know that under the conditions $\alpha > \frac{1}{3}$ and $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{4}{3\alpha-1}$, $f(\lambda_1)$ is positive when λ_1 is small and close to 1. Therefore there exist parameter configurations where the overconfident player's equilibrium winning probability P_1 is higher than the benchmark.

(2) P_2

We show that $P_2 < \frac{1}{4}$:

$$P_{2} = p_{23}^{f} p_{21}^{s}$$
$$= \frac{1}{2} p_{21}^{s}$$
$$< \frac{1}{2} \times \frac{1}{2} = \frac{1}{4}$$

(3) P_3 and P_4

We show that $P_3 = P_4 > \frac{1}{4}$:

$$P_{3} = P_{4} = p_{12}^{s} p_{31}^{f} p_{34}^{s} + p_{21}^{s} p_{32}^{f} p_{34}^{s}$$

$$= p_{12}^{s} p_{31}^{f} \frac{1}{2} + p_{21}^{s} \frac{1}{2} \frac{1}{2}$$

$$= p_{12}^{s} p_{31}^{f} \frac{1}{2} + (1 - p_{12}^{s}) \frac{1}{2} \frac{1}{2}$$

$$= p_{12}^{s} \left(p_{31}^{f} \frac{1}{2} - \frac{1}{4} \right) + \frac{1}{4}$$

$$> p_{12}^{s} \left(\frac{1}{2} \frac{1}{2} - \frac{1}{4} \right) + \frac{1}{4} = \frac{1}{4}$$

(4) compare P_1 and P_3

$$P_{1} - P_{3} = p_{13}^{f} p_{12}^{s} - p_{12}^{s} p_{31}^{f} p_{34}^{s} - p_{21}^{s} p_{32}^{f} p_{34}^{s}$$

$$= p_{13}^{f} p_{12}^{s} - p_{12}^{s} p_{31}^{f} p_{34}^{s} - (1 - p_{12}^{s}) p_{32}^{f} p_{34}^{s}$$

$$= p_{13}^{f} p_{12}^{s} - \frac{1}{2} p_{12}^{s} p_{31}^{f} - \frac{1}{2} \frac{1}{2} (1 - p_{12}^{s})$$

$$= p_{13}^{f} p_{12}^{s} - \frac{1}{2} p_{12}^{s} \left(1 - p_{13}^{f} \right) - \frac{1}{4} (1 - p_{12}^{s})$$

$$= \frac{3}{2} p_{13}^{f} p_{12}^{s} - \frac{1}{4} p_{12}^{s} - \frac{1}{4}$$

The sign of $P_1 - P_3$ is the same as the sign of $6p_{13}^f p_{12}^s - p_{12}^s - 1$ Let $f(\lambda_1) = 6p_{13}^f p_{12}^s - p_{12}^s - 1$

$$f(\lambda_1 = 1) = 6 \times \frac{1}{2} \times \frac{1}{2} - \frac{1}{2} - 1 = 0$$

$$\begin{split} f(\lambda_1 = \hat{\lambda}) &= 6 \times \frac{1}{2} \hat{\lambda}^{-\frac{\alpha}{2\alpha+1}} \times \frac{1}{2} - \frac{1}{2} - 1 < 0 \\ f(\lambda_1) &= 3\lambda_1^{-\frac{\alpha}{2\alpha+1}} \left[1 - \frac{1}{2} \lambda_1^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \right] \\ &- \left[1 - \frac{1}{2} \lambda_1^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \right] - 1 \\ &= 3\lambda_1^{-\frac{\alpha}{2\alpha+1}} - \frac{1}{2} \lambda_1^{\alpha} \left(3\lambda_1^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \\ &\left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} - 2 \end{split}$$

$$\begin{aligned} f'(\lambda_1) &= -3\frac{\alpha}{2\alpha+1}\lambda_1^{-\frac{\alpha}{2\alpha+1}-1} - \frac{1}{2}\left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\right)^{\alpha} \\ & \left[\left(3\left(-\frac{\alpha}{2\alpha+1} + \alpha\right)\lambda_1^{-\frac{\alpha}{2\alpha+1} + \alpha - 1} - \alpha\lambda_1^{\alpha - 1}\right)\left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} \right. \\ & \left. + \lambda_1^{\alpha}\left(3\lambda_1^{-\frac{\alpha}{2\alpha+1}} - 1\right)\left(-\alpha\right)\left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha - 1} \right. \\ & \left. \frac{1+\alpha}{2}\frac{\alpha+1}{2\alpha+1}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}-1}\right] \end{aligned}$$

$$\begin{aligned} f'(\lambda_1 = 1) &= -3\frac{\alpha}{2\alpha + 1} - \frac{1}{2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \\ & \left[\left(3 \left(-\frac{\alpha}{2\alpha + 1} + \alpha \right) - \alpha \right) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-\alpha} \right. \\ & - 2\alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-\alpha - 1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ &= -3\frac{\alpha}{2\alpha + 1} - \frac{1}{2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \\ & \left[\left(2\alpha - 3\frac{\alpha}{2\alpha + 1} \right) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-\alpha} \right] \\ & - 2\alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-\alpha - 1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ &= -3\frac{\alpha}{2\alpha + 1} - \frac{1}{2} \left(2\alpha - 3\frac{\alpha}{2\alpha + 1} \right) \\ & + \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ &= -3\frac{\alpha}{2\alpha + 1} - \alpha + \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \\ &= -\frac{3}{2}\frac{\alpha}{2\alpha + 1} - \alpha + \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ &= \alpha \left(-\frac{3}{2}\frac{1}{2\alpha + 1} - 1 + \frac{1 + \alpha}{2}\frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} \right) \\ & \Leftrightarrow \frac{1 + \alpha}{2}\frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} > \\ & \Leftrightarrow \frac{\frac{1 + \alpha}{2}\frac{\alpha + 1}{2\alpha + 1}}{\frac{3}{2}\frac{1}{2\alpha + 1}} \right) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \\ & \Leftrightarrow \frac{\frac{(\alpha + 1)^2}{3}}{\frac{3}{2}(2\alpha + 1)} > \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \\ & \Leftrightarrow \frac{(\alpha + 1)^2}{(4\alpha + 5)} > \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \\ & \leftrightarrow \frac{(\alpha + 1)^2}{(4\alpha + 5)} > \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \\ & \leftrightarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \right) \\ & \leftrightarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \right) \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\ & \leftarrow (\alpha + 1)^2 + \alpha - 1 + \frac{u(w_2)}{2} \\$$

$$\iff \frac{(\alpha+1)}{(4\alpha+5)} + \frac{\alpha-1}{2} > \frac{u(w_2)}{u(w_1) - u(w_2)}$$

$$\iff \frac{2(\alpha+1)^2 + (\alpha-1)(4\alpha+5)}{2(4\alpha+5)} > \frac{u(w_2)}{u(w_1) - u(w_2)}$$

$$\iff \frac{6\alpha^2 + 5\alpha - 3}{2(4\alpha+5)} > \frac{u(w_2)}{u(w_1) - u(w_2)}$$
This is satisfied when $\alpha > \frac{-5+\sqrt{97}}{12}$ and $\frac{u(w_1) - u(w_2)}{u(w_2)} > \frac{2(4\alpha+5)}{6\alpha^2 + 5\alpha - 3}.$

We show $\frac{2(4\alpha+5)}{6\alpha^2+5\alpha-3} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$:

$$\frac{2(4\alpha+5)}{6\alpha^2+5\alpha-3} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)} \iff \frac{4\alpha+5}{6\alpha^2+5\alpha-3} > \frac{2\alpha+1}{\alpha(3\alpha+1)}$$
$$\iff (4\alpha+5)\alpha(3\alpha+1) > (2\alpha+1)\left(6\alpha^2+5\alpha-3\right)$$
$$\iff 12\alpha^3+19\alpha^2+5\alpha > 12\alpha^3+16\alpha^2-\alpha-3$$
$$\iff 3\alpha^2+6\alpha+3 > 0$$

Thus we know that under the conditions $\alpha > \frac{-5+\sqrt{97}}{12}$ and $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(4\alpha+5)}{6\alpha^2+5\alpha-3}$, $f(\lambda_1)$ is positive when λ_1 is small and close to 1. Therefore there exist parameter configurations where the overconfident player's equilibrium winning probability P_1 is higher than that of the rational player in the other semi-final P_3 .

- 2. When either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leqslant \frac{2(1+2\alpha)}{\alpha(1+3\alpha)}$ or $\lambda_1 \geqslant \hat{\lambda}$
 - (1) P_1

Since player 3 and player 4 are identical, the equilibrium winning probability of player 1 is

$$P_1 = p_{13}^f p_{12}^s \\ < \frac{1}{2} \times \frac{1}{2} = \frac{1}{4}$$

 $(2) P_2$

$$P_{2} = p_{23}^{f} p_{21}^{s}$$
$$= \frac{1}{2} p_{21}^{s}$$
$$> \frac{1}{2} \times \frac{1}{2} = \frac{1}{4}$$

(3) P_3 and P_4

 $P_3 = P_4 > \frac{1}{4}$ still holds.

Proof of Proposition 6

The perceived winning probabilities of the players are:

$$\widetilde{p}_{13}^{f} = \begin{cases} 1 - \frac{1}{2} \frac{e_3^{\alpha}}{\lambda_1 e_1^{\alpha}} & \text{if } \lambda_1 e_1^{\alpha} \geqslant e_3^{\alpha} \\ \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_3^{\alpha}} & \text{if } \lambda_1 e_1^{\alpha} \leqslant e_3^{\alpha} \end{cases}$$
$$\widetilde{p}_{31}^{f} = \begin{cases} 1 - \frac{1}{2} \frac{e_1^{\alpha}}{\lambda_3 e_3^{\alpha}} & \text{if } \lambda_3 e_3^{\alpha} \geqslant e_1^{\alpha} \\ \frac{1}{2} \frac{\lambda_3 e_3^{\alpha}}{e_1^{\alpha}} & \text{if } \lambda_3 e_3^{\alpha} \leqslant e_1^{\alpha} \end{cases}$$
$$\max \quad \widetilde{E}^{f}(U_{12}) = \widetilde{p}_{12}^{f}[u(w_1) - u(w_2)] + u(w_2) - c_1^{\alpha} = c_1^{\alpha} \end{cases}$$

Overconfident player 1 max $\widetilde{E}^f(U_{13}) = \widetilde{p}_{13}^f[u(w_1) - u(w_2)] + u(w_2) - ce_1$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_3^{\alpha}}{\lambda_1 e_1^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_1 & \text{if } \lambda_1 e_1^{\alpha} \ge e_3^{\alpha} \\ \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_3^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_1 & \text{if } \lambda_1 e_1^{\alpha} \le e_3^{\alpha} \end{cases}$$

Overconfident player 3 max $\widetilde{E}^{f}(U_{31}) = \widetilde{p}_{31}^{f}[u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{3}$ $= \begin{cases} \left(1 - \frac{1}{2}\frac{e_{1}^{\alpha}}{\lambda_{3}e_{3}^{\alpha}}\right) [u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{3} & \text{if} \quad \lambda_{3}e_{3}^{\alpha} \geqslant e_{1}^{\alpha} \\ \frac{1}{2}\frac{\lambda_{3}e_{3}^{\alpha}}{e_{1}^{\alpha}} [u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{3} & \text{if} \quad \lambda_{3}e_{3}^{\alpha} \leqslant e_{1}^{\alpha} \end{cases}$

There are 4 cases.

 $\begin{cases} \lambda_1 e_1^{\alpha} \geqslant e_3^{\alpha} & and \quad \lambda_3 e_3^{\alpha} \geqslant e_1^{\alpha} \\ \lambda_1 e_1^{\alpha} \geqslant e_3^{\alpha} & and \quad \lambda_3 e_3^{\alpha} \leqslant e_1^{\alpha} \\ \lambda_1 e_1^{\alpha} \leqslant e_3^{\alpha} & and \quad \lambda_3 e_3^{\alpha} \geqslant e_1^{\alpha} \\ \lambda_1 e_1^{\alpha} \leqslant e_3^{\alpha} & and \quad \lambda_3 e_3^{\alpha} \leqslant e_1^{\alpha} \end{cases}$

Since $\lambda_1 > \lambda_3 > 1$, the fourth case is impossible.

1. equilibrium efforts

 $\begin{array}{ll} (1) \ \text{case 1:} \ \lambda_{1}e_{1}^{\alpha} \geqslant e_{3}^{\alpha} & and \quad \lambda_{3}e_{3}^{\alpha} \geqslant e_{1}^{\alpha} \\ \\ \text{Player 1} & max \quad \left(1 - \frac{1}{2}\frac{e_{3}^{\alpha}}{\lambda_{1}e_{1}^{\alpha}}\right) [u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{1} \\ \\ \text{Player 3} & max \quad \left(1 - \frac{1}{2}\frac{e_{1}^{\alpha}}{\lambda_{3}e_{3}^{\alpha}}\right) [u(w_{1}) - u(w_{2})] + u(w_{2}) - ce_{3} \\ \\ \text{F.o.c} \\ \\ \left[e_{1}\right] & \frac{\alpha}{2\lambda_{1}}\frac{e_{3}^{\alpha}}{e_{1}^{\alpha+1}} [u(w_{1}) - u(w_{2})] - c = 0 \\ \\ \left[e_{3}\right] & \frac{\alpha}{2\lambda_{3}}\frac{e_{1}^{\alpha}}{e_{3}^{\alpha+1}} [u(w_{1}) - u(w_{2})] - c = 0 \\ \\ \\ \text{S.o.c} \\ \\ \left[e_{1}\right] & \frac{\alpha}{2\lambda_{1}} (-\alpha - 1)\frac{e_{3}^{\alpha}}{e_{1}^{\alpha+2}} [u(w_{1}) - u(w_{2})] < 0 \\ \\ \left[e_{3}\right] & \frac{\alpha}{2\lambda_{3}} (-\alpha - 1)\frac{e_{1}^{\alpha}}{e_{3}^{\alpha+2}} [u(w_{1}) - u(w_{2})] < 0 \\ \\ \\ \\ \text{Solve F.O.C , we get} \end{array}$

$$e_{1} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})]$$
$$e_{3} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})]$$

Check the condition $\lambda_3 e_3^{\alpha} \ge e_1^{\alpha}$:

$$\lambda_{3}e_{3}^{\alpha} \ge e_{1}^{\alpha} \Longleftrightarrow \frac{\lambda_{3}e_{3}^{\alpha}}{e_{1}^{\alpha}} \ge 1$$
$$\iff \lambda_{1}^{\frac{\alpha}{2\alpha+1}}\lambda_{3}^{\frac{\alpha+1}{2\alpha+1}} \ge 1$$

Check the condition $\lambda_1 e_1^{\alpha} \ge e_3^{\alpha}$:

$$\begin{split} \lambda_1 e_1^{\alpha} \geqslant e_3^{\alpha} & \Longleftrightarrow \frac{\lambda_1 e_1^{\alpha}}{e_3^{\alpha}} \geqslant 1\\ & \Longleftrightarrow \lambda_1^{\frac{\alpha+1}{2\alpha+1}} \lambda_3^{\frac{\alpha}{2\alpha+1}} \geqslant 1 \end{split}$$

which is always satisfied.

(2) case 2: $\lambda_1 e_1^{\alpha} \ge e_3^{\alpha}$ and $\lambda_3 e_3^{\alpha} \le e_1^{\alpha}$ Player 1 max $\left(1 - \frac{1}{2} \frac{e_3^{\alpha}}{\lambda_1 e_1^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_1$ Player 3 max $\frac{1}{2} \frac{\lambda_3 e_3^{\alpha}}{e_1^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_3$

F.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1} \frac{e_3^{\alpha}}{e_1^{\alpha+1}} [u(w_1) - u(w_2)] - c = 0$$

$$[e_3] \quad \frac{\alpha}{2} \lambda_3 \frac{e_3^{\alpha-1}}{e_1^{\alpha}} [u(w_1) - u(w_2)] - c = 0$$

divide the two F.O.C , we get

$$\frac{e_3}{e_1} = \lambda_1 \lambda_3 > 1$$

which contradicts the condition that $\lambda_3 e_3^{\alpha} \leq e_1^{\alpha}$ (3) case 3: $\lambda_1 e_1^{\alpha} \leq e_3^{\alpha}$ and $\lambda_3 e_3^{\alpha} \geq e_1^{\alpha}$

Player 1
$$max \quad \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_3^{\alpha}} [u(w_1) - u(w_2)] + u(w_2) - ce_1$$

Player 3 $max \quad \left(1 - \frac{1}{2} \frac{e_1^{\alpha}}{\lambda_3 e_3^{\alpha}}\right) [u(w_1) - u(w_2)] + u(w_2) - ce_3$

F.o.c

$$[e_1] \quad \frac{\alpha}{2}\lambda_1 \frac{e_1^{\alpha-1}}{e_3^{\alpha}} [u(w_1) - u(w_2)] - c = 0$$
$$[e_3] \quad \frac{\alpha}{2\lambda_3} \frac{e_1^{\alpha}}{e_3^{\alpha+1}} [u(w_1) - u(w_2)] - c = 0$$

divide the two F.O.C, we get

$$\frac{e_3}{e_1} = \frac{1}{\lambda_1 \lambda_3} < 1$$

which contradicts the condition that $\lambda_1 e_1^{\alpha} \leq e_3^{\alpha}$.

Thus the unique equilibrium is

$$e_1^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \lambda_3^{-\frac{\alpha}{2\alpha+1}} [u(w_1) - u(w_2)]$$

$$e_3^f = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}} \lambda_3^{-\frac{\alpha+1}{2\alpha+1}} [u(w_1) - u(w_2)]$$

Since $\lambda_1 > \lambda_3 > 1$, we can get $e_1^f < e_3^f < \overline{e}^f$.

2. winning probabilities

The true winning probabilities are

$$\begin{split} p_{13}^{f} &= \frac{1}{2} \left(\frac{e_{1}^{f}}{e_{3}^{f}} \right)^{\alpha} \\ &= \frac{1}{2} \left(\lambda_{1}^{-\frac{1}{2\alpha+1}} \lambda_{3}^{\frac{1}{2\alpha+1}} \right)^{\alpha} \\ &= \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{3}^{\frac{\alpha}{2\alpha+1}} \\ &< \frac{1}{2} \\ p_{31}^{f} &= 1 - p_{13}^{f} = 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{3}^{\frac{\alpha}{2\alpha+1}} > \frac{1}{2} \end{split}$$

The perceived winning probabilities are

$$\begin{split} \widetilde{p}_{13}^{f} &= 1 - \frac{1}{2} \frac{(e_{3}^{f})^{\alpha}}{\lambda_{1}(e_{1}^{f})^{\alpha}} \\ &= 1 - \frac{1}{2\lambda_{1}} \left(\lambda_{1}^{\frac{1}{2\alpha+1}} \lambda_{3}^{-\frac{1}{2\alpha+1}} \right)^{\alpha} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} \end{split}$$

$$\begin{split} \widetilde{p}_{31}^{f} &= 1 - \frac{1}{2} \frac{(e_{1}^{f})^{\alpha}}{\lambda_{3}(e_{3}^{f})^{\alpha}} \\ &= 1 - \frac{1}{2\lambda_{3}} \left(\lambda_{1}^{-\frac{1}{2\alpha+1}} \lambda_{3}^{\frac{1}{2\alpha+1}} \right)^{\alpha} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \end{split}$$

~

Thus we have

$$\widetilde{p}_{13}^f > \widetilde{p}_{31}^f > p_{31}^f > \frac{1}{2} > p_{13}^f$$

3. expected utilities of final

$$\begin{split} \widetilde{E}^{f}(U_{13}) &= \widetilde{p}_{13}^{f} u(w_{1}) + (1 - \widetilde{p}_{13}^{f}) u(w_{2}) - c e_{1}^{f} \\ &= \left(1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}}\right) u(w_{1}) + \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} u(w_{2}) \\ &- \frac{\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})] \\ &= u(w_{1}) - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})] \end{split}$$

$$\widetilde{E}^{f}(U_{31}) = \widetilde{p}_{31}^{f}u(w_{1}) + (1 - \widetilde{p}_{31}^{f})u(w_{2}) - ce_{3}^{f}$$

$$= \left(1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}}\right)u(w_{1}) + \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}}u(w_{2})$$

$$- \frac{\alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}}[u(w_{1}) - u(w_{2})]$$

$$= u(w_{1}) - \frac{1 + \alpha}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}}[u(w_{1}) - u(w_{2})]$$

Since $\alpha < 1$ and $\lambda_1 > \lambda_3 > 1$, $\widetilde{E}^f(U_{13}) > \widetilde{E}^f(U_{31}) > \overline{E}^f(U)$. The participation constraints are also satisfied.

Proof of Proposition 7

1. Continuation values

Using Proposition 2, we can get the continuation values of each player. Overconfident player 1:

$$\widetilde{v}_1 = p_{34}^s \widetilde{E}^f(U_{13}) + p_{43}^s \widetilde{E}^f(U_{14})$$

Since player 3 and 4 are identical, $\widetilde{E}^{f}(U_{13}) = \widetilde{E}^{f}(U_{14})$,

$$\widetilde{v}_1 = u(w_1) - \frac{1+\alpha}{2}\lambda_1^{-\frac{\alpha+1}{2\alpha+1}}[u(w_1) - u(w_2)]$$

We show that $\tilde{v}_1 > \overline{v}$:

$$\begin{split} \widetilde{v}_1 &> \overline{v} \Longleftrightarrow u(w_1) - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} [u(w_1) - u(w_2)] > u(w_1) - \frac{1+\alpha}{2} [u(w_1) - u(w_2)] \\ &\iff \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} < 1 \end{split}$$

Overconfident player 2:

$$\widetilde{v}_2 = p_{34}^s \widetilde{E}^f(U_{23}) + p_{43}^s \widetilde{E}^f(U_{24})$$

Since player 3 and 4 are identical, $\widetilde{E}^{f}(U_{23}) = \widetilde{E}^{f}(U_{24})$,

$$\widetilde{v}_2 = u(w_1) - \frac{1+\alpha}{2}\lambda_2^{-\frac{\alpha+1}{2\alpha+1}}[u(w_1) - u(w_2)] > \overline{v}$$

We can easily get $\tilde{v}_1 > \tilde{v}_2$:

$$\widetilde{v}_1 > \widetilde{v}_2 \Longleftrightarrow \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} < \lambda_2^{-\frac{\alpha+1}{2\alpha+1}}$$

Thus we have $\widetilde{v}_1 > \widetilde{v}_2 > \overline{v}$.

2. The equilibrium

Player 1 max
$$E^{s}(U_{12}) = \widetilde{p}_{12}^{s}\widetilde{v}_{1} - ce_{1}$$

$$= \begin{cases} \left(1 - \frac{1}{2}\frac{e_{2}^{\alpha}}{\lambda_{1}e_{1}^{\alpha}}\right)\widetilde{v}_{1} - ce_{1} & \text{if } \lambda_{1}e_{1}^{\alpha} \ge e_{2}^{\alpha} \\ \frac{1}{2}\frac{\lambda_{1}e_{1}^{\alpha}}{e_{2}^{\alpha}}\widetilde{v}_{1} - ce_{1} & \text{if } \lambda_{1}e_{1}^{\alpha} \le e_{2}^{\alpha} \end{cases}$$

 \sim

Player 2 max
$$\widetilde{E}^s(U_{21}) = \widetilde{p}_{21}^s \widetilde{v}_2 - ce_2$$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_1^{\alpha}}{\lambda_2 e_2^{\alpha}}\right) \widetilde{v}_2 - ce_2 & \text{if } \lambda_2 e_2^{\alpha} \ge e_1^{\alpha} \\ \frac{1}{2} \frac{\lambda_2 e_2^{\alpha}}{e_1^{\alpha}} \widetilde{v}_2 - ce_2 & \text{if } \lambda_2 e_2^{\alpha} \leqslant e_1^{\alpha} \end{cases}$$

There are 4 cases.

$$\begin{cases} \lambda_1 e_1^{\alpha} \geqslant e_2^{\alpha} \quad and \quad \lambda_2 e_2^{\alpha} \leqslant e_1^{\alpha} \\ \lambda_1 e_1^{\alpha} \geqslant e_2^{\alpha} \quad and \quad \lambda_2 e_2^{\alpha} \geqslant e_1^{\alpha} \\ \lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha} \quad and \quad \lambda_2 e_2^{\alpha} \geqslant e_1^{\alpha} \\ \lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha} \quad and \quad \lambda_2 e_2^{\alpha} \leqslant e_1^{\alpha} \end{cases}$$

Since $\lambda_1 > \lambda_2 > 1$, the fourth case is impossible.

- (1) case 1: $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $\lambda_2 e_2^{\alpha} \le e_1^{\alpha}$, which corresponds to (i) in proposition 7.
 - Player 1 $max \left(1 \frac{1}{2}\frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right)\widetilde{v}_1 ce_1$ Player 2 $max \quad \frac{1}{2}\frac{\lambda_2 e_2^{\alpha}}{e_1^{\alpha}}\widetilde{v}_2 - ce_2$

F.o.c

$$\begin{bmatrix} e_1 \end{bmatrix} \quad \frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}} \widetilde{v}_1 - c = 0$$
$$\begin{bmatrix} e_2 \end{bmatrix} \quad \frac{\alpha\lambda_2}{2} \frac{e_2^{\alpha-1}}{e_1^{\alpha}} \widetilde{v}_2 - c = 0$$

S.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1}(-\alpha-1)\frac{e_2^{\alpha}}{e_1^{\alpha+2}}\widetilde{v}_1 < 0$$
$$[e_2] \quad \frac{\alpha\lambda_2}{2}(\alpha-1)\frac{e_2^{\alpha-2}}{e_1^{\alpha}}\widetilde{v}_2 < 0$$

Solve the two F.O.C , we get

$$e_1 = \frac{\alpha}{2c} \lambda_1^{\alpha - 1} \lambda_2^{\alpha} (\widetilde{v}_1)^{1 - \alpha} (\widetilde{v}_2)^{\alpha}$$

$$e_2 = \frac{\alpha}{2c} \lambda_1^{\alpha} \lambda_2^{\alpha+1} (\widetilde{v}_1)^{-\alpha} (\widetilde{v}_2)^{\alpha+1}$$

$$\frac{e_2}{e_1} = \frac{\lambda_2 \widetilde{v}_2}{\lambda_1^{-1} \widetilde{v}_1}$$

Check the conditions $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $\lambda_2 e_2^{\alpha} \le e_1^{\alpha}$:

(1) $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ When $\lambda_2 e_2^{\alpha} \le e_1^{\alpha}$ is satisfied, $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ is satisfied.

(2)
$$\lambda_2 e_2^{\alpha} \leqslant e_1^{\alpha}$$

$$\begin{split} \lambda_{2}e_{2}^{\alpha} &\leqslant e_{1}^{\alpha} \\ &\longleftrightarrow \lambda_{2} \left(\frac{e_{2}}{e_{1}}\right)^{\alpha} \leqslant 1 \\ &\Leftrightarrow \lambda_{2} \left(\frac{\lambda_{1}\lambda_{2}\tilde{v}_{2}}{\tilde{v}_{1}}\right)^{\alpha} \leqslant 1 \\ &\Leftrightarrow \lambda_{1}^{\alpha}\lambda_{2}^{\alpha+1}(\tilde{v}_{1})^{-\alpha}(\tilde{v}_{2})^{\alpha} \leqslant 1 \\ &\Leftrightarrow \lambda_{1}^{\alpha}\lambda_{2}^{\alpha+1}(\tilde{v}_{2})^{\alpha} \leqslant \lambda_{1}^{-\alpha}(\tilde{v}_{1})^{\alpha} \\ &\Leftrightarrow \lambda_{1}\lambda_{2}^{\frac{\alpha+1}{\alpha}}\tilde{v}_{2} \leqslant \tilde{v}_{1} \\ &\Leftrightarrow \lambda_{1}\lambda_{2}^{\frac{\alpha+1}{\alpha}} \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right) \leqslant \left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right) \\ &\Leftrightarrow \left(\lambda_{1}\lambda_{2}^{\frac{\alpha+1}{\alpha}} - 1\right) \frac{u(w_{1})}{u(w_{1})-u(w_{2})} \leqslant \frac{1+\alpha}{2} \left(\lambda_{1}\lambda_{2}^{\frac{\alpha+1}{\alpha}-\frac{\alpha+1}{2\alpha+1}} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right) \\ &\Leftrightarrow \frac{u(w_{1})}{u(w_{1})-u(w_{2})} \frac{2}{1+\alpha} \leqslant \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} \end{split}$$

Therefore the solution

only applies

$$e_1^s = \frac{\alpha}{2c} \lambda_1^{\alpha-1} \lambda_2^{\alpha} (\widetilde{v}_1)^{1-\alpha} (\widetilde{v}_2)^{\alpha}$$
$$e_2^s = \frac{\alpha}{2c} \lambda_1^{\alpha} \lambda_2^{\alpha+1} (\widetilde{v}_1)^{-\alpha} (\widetilde{v}_2)^{\alpha+1}$$
$$\text{when } \frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1+\alpha} \leqslant \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}}.$$

Therefore, in proposition 7 (i), $e_1^s > e_2^s$ is always satisfied since $\lambda_2 e_2^{\alpha} \leq e_1^{\alpha}$ and $\lambda_2 > 1$.

(2) case 2: $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $\lambda_2 e_2^{\alpha} \ge e_1^{\alpha}$, which corresponds to (ii) and (iii) in proposition 7.

Player 1 $max \left(1 - \frac{1}{2}\frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right)\widetilde{v}_1 - ce_1$ Player 2 $max \left(1 - \frac{1}{2}\frac{e_1^{\alpha}}{\lambda_2 e_2^{\alpha}}\right)\widetilde{v}_2 - ce_2$

F.o.c

$$\begin{bmatrix} e_1 \end{bmatrix} \quad \frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}} \widetilde{v}_1 - c = 0$$
$$\begin{bmatrix} e_2 \end{bmatrix} \quad \frac{\alpha}{2\lambda_2} \frac{e_1^{\alpha}}{e_2^{\alpha+1}} \widetilde{v}_2 - c = 0$$

S.o.c

 $[e_1] \quad \frac{\alpha}{2\lambda_1}(-\alpha-1)\frac{e_2^{\alpha}}{e_1^{\alpha+2}}\widetilde{v}_1 < 0$

$$[e_2] \quad \frac{\alpha}{2\lambda_2}(-\alpha-1)\frac{e_1^{\alpha}}{e_2^{\alpha+2}}\widetilde{v}_2 < 0$$

Solve F.O.C , we get

$$e_{1} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha}{2\alpha+1}}$$
$$e_{2} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha+1}{2\alpha+1}}$$
$$\frac{e_{2}}{e_{1}} = \left(\frac{\lambda_{2}^{-1} \widetilde{v}_{2}}{\lambda_{1}^{-1} \widetilde{v}_{1}}\right)^{\frac{1}{2\alpha+1}}$$

Check the conditions $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $\lambda_2 e_2^{\alpha} \ge e_1^{\alpha}$: (1) $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$

$$\lambda_{1}e_{1}^{\alpha} \geq e_{2}^{\alpha} \iff \frac{\lambda_{1}e_{1}^{\alpha}}{e_{2}^{\alpha}} \geq 1$$
$$\iff \lambda_{1}^{\frac{\alpha+1}{2\alpha+1}}\lambda_{2}^{\frac{\alpha}{2\alpha+1}}(\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}}(\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \geq 1$$
$$\iff \lambda_{1}^{\frac{\alpha+1}{2\alpha+1}}(\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} \geq \lambda_{2}^{-\frac{\alpha}{2\alpha+1}}(\widetilde{v}_{2})^{\frac{\alpha}{2\alpha+1}}$$

this is always satisfied. Therefore $\lambda_1 e_1^\alpha \ge e_2^\alpha$ always holds. (2) $\lambda_2 e_2^\alpha \ge e_1^\alpha$

$$\begin{split} \lambda_{2}e_{2}^{\alpha} &\geq e_{1}^{\alpha} \\ \Leftrightarrow \lambda_{2} \left(\frac{e_{2}}{e_{1}}\right)^{\alpha} \geq 1 \\ \Leftrightarrow \lambda_{2} \left(\frac{\lambda_{2}^{-1}\tilde{v}_{2}}{\lambda_{1}^{-1}\tilde{v}_{1}}\right)^{\frac{\alpha}{2\alpha+1}} \geq 1 \\ \Leftrightarrow \lambda_{1}^{\frac{\alpha}{2\alpha+1}} \lambda_{2}^{\frac{\alpha+1}{2\alpha+1}} (\tilde{v}_{1})^{-\frac{\alpha}{2\alpha+1}} (\tilde{v}_{2})^{\frac{\alpha}{2\alpha+1}} \geq 1 \\ \Leftrightarrow \lambda_{1}^{\frac{\alpha}{2\alpha+1}} \lambda_{2}^{\frac{\alpha+1}{2\alpha+1}} (\tilde{v}_{2})^{\frac{\alpha}{2\alpha+1}} \geq (\tilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} \\ \Leftrightarrow \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \tilde{v}_{2}^{\alpha} \geq \tilde{v}_{1} \\ \Leftrightarrow \lambda_{1} \lambda_{2}^{\frac{\alpha+1}{\alpha}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right) \geq \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right) \\ \Leftrightarrow \left(\lambda_{1} \lambda_{2}^{\frac{\alpha+1}{\alpha}} - 1\right) \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \geq \frac{1+\alpha}{2} \left(\lambda_{1} \lambda_{2}^{\frac{\alpha+1}{\alpha} - \frac{\alpha+1}{2\alpha+1}} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right) \\ \Leftrightarrow \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1+\alpha} \geq \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} \end{split}$$

Thus $\lambda_2 e_2^{\alpha} \ge e_1^{\alpha}$ is satisfied when $\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} \ge \frac{\lambda_2^{\frac{(\alpha+1)}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}}$. Therefore the solution

$$e_1^s = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \lambda_2^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_1)^{\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_2)^{\frac{\alpha}{2\alpha+1}}$$

$$e_2^s = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha}{2\alpha+1}} \lambda_2^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_1)^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_2)^{\frac{\alpha+1}{2\alpha+1}}$$
only applies when
$$\frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1+\alpha} \ge \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}}.$$

When the above condition is satisfied,

$$e_1^s - e_2^s \begin{cases} > 0 & \text{when } \lambda_1^{-1} \widetilde{v}_1 > \lambda_2^{-1} \widetilde{v}_2 \\ = 0 & \text{when } \lambda_1^{-1} \widetilde{v}_1 = \lambda_2^{-1} \widetilde{v}_2 \\ < 0 & \text{when } \lambda_1^{-1} \widetilde{v}_1 < \lambda_2^{-1} \widetilde{v}_2 \end{cases}$$

$$\begin{split} \lambda_{1}^{-1} \widetilde{v}_{1} &\gtrless \lambda_{2}^{-1} \widetilde{v}_{2} \\ \iff \lambda_{1}^{-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) \gtrless \lambda_{2}^{-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right) \\ \iff \frac{1 + \alpha}{2} \left(\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1} - 1} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1} - 1} \right) \end{Bmatrix} \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \left(\lambda_{2}^{-1} - \lambda_{1}^{-1} \right) \\ \iff \frac{\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1} - 1} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1} - 1}}{\lambda_{2}^{-1} - \lambda_{1}^{-1}} \end{Bmatrix} \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \\ \iff \frac{\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1} - 1} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1} - 1}}{\lambda_{1} - \lambda_{2}} \end{Bmatrix} \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \\ \iff \frac{\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1} - 1} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1} - 1}}{\lambda_{1} - \lambda_{2}} \end{Bmatrix} \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \\ \iff \frac{1}{\lambda_{1} - \lambda_{2}} \left(\frac{\lambda_{1}}{\lambda_{2}^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_{2}}{\lambda_{1}^{\frac{\alpha+1}{2\alpha+1}}} \right) \end{Bmatrix} \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \end{split}$$

Therefore,

$$e_{1}^{s} - e_{2}^{s} \begin{cases} > 0 \quad \text{when } \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} < \frac{1}{\lambda_{1} - \lambda_{2}} \left(\frac{\lambda_{1}}{\lambda_{2}^{\alpha + 1}} - \frac{\lambda_{2}}{\lambda_{1}^{2\alpha + 1}} \right) \\ = 0 \quad \text{when } \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} = \frac{1}{\lambda_{1} - \lambda_{2}} \left(\frac{\lambda_{1}}{\frac{\alpha + 1}{\lambda_{2}}} - \frac{\lambda_{2}}{\lambda_{1}^{\alpha + 1}} \right) \\ < 0 \quad \text{when } \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} > \frac{1}{\lambda_{1} - \lambda_{2}} \left(\frac{\lambda_{1}}{\frac{\alpha + 1}{\lambda_{2}}} - \frac{\lambda_{2}}{\lambda_{1}^{\alpha + 1}} \right) \\ < 0 \quad \text{when } \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} > \frac{1}{\lambda_{1} - \lambda_{2}} \left(\frac{\lambda_{1}}{\frac{\alpha + 1}{\lambda_{2}}} - \frac{\lambda_{2}}{\lambda_{1}^{\alpha + 1}} \right) \end{cases}$$

Combining the condition $\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} \ge \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}}$ we can get

$$e_{1}^{s} - e_{2}^{s} \begin{cases} > 0 \quad \text{when} \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} \leqslant \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} < \frac{1}{\lambda_{1} - \lambda_{2}} \left(\frac{\lambda_{1}}{\lambda_{2}^{\frac{\alpha+1}{\alpha+1}}} - \frac{\lambda_{2}}{\lambda_{1}^{\frac{\alpha+1}{2\alpha+1}}} \right) \\ \leqslant 0 \quad \text{when} \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \geqslant \max \left[\frac{\frac{\lambda_{1}}{\alpha+1} - \frac{\lambda_{2}}{\alpha+1}}{\lambda_{2}^{\frac{\alpha+1}{\alpha+1}}} , \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} \right] \end{cases}$$

where the first line corresponds to proposition 7 (ii) and the second line corresponds to proposition 7 (iii).

(3) case 3: $\lambda_1 e_1^{\alpha} \leq e_2^{\alpha}$ and $\lambda_2 e_2^{\alpha} \geq e_1^{\alpha}$ Player 1 $max \quad \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_2^{\alpha}} \widetilde{v}_1 - ce_1$ Player 2 $max \quad \left(1 - \frac{e_1^{\alpha}}{\lambda_2 e_2^{\alpha}}\right) \widetilde{v}_2 - ce_2$ F.o.c $[e_1] \quad \frac{\alpha \lambda_1}{2} \frac{e_1^{\alpha-1}}{e_2^{\alpha}} \widetilde{v}_1 - c = 0$ $[e_2] \quad \frac{\alpha}{2} \frac{e_1^{\alpha}}{\lambda_2 e_2^{\alpha+1}} \widetilde{v}_2 - c = 0$

divide the two F.O.C , we get

$$\frac{e_2}{e_1} = \frac{\widetilde{v}_2}{\lambda_1 \lambda_2 \widetilde{v}_1} < 1$$

which contradicts the condition that $\lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha}$

Therefore, the equilibrium in this semi-final is given by:

(1) Proposition 7 (i): when
$$\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} \leq \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}}$$

 $e_1^s = \frac{\alpha}{2c} \lambda_1^{\alpha-1} \lambda_2^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{1-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_2^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\alpha} [u(w_1) - u(w_2)]$

$$e_{2}^{s} = \frac{\alpha}{2c} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \\ \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{1+\alpha} [u(w_{1}) - u(w_{2})]$$

We show that $e_1^s > \overline{e}^s$:

$$e_1^s = \frac{\alpha}{2c} \left(\lambda_1^{-1} \widetilde{v}_1\right)^{1-\alpha} \lambda_2^{\alpha} (\widetilde{v}_2)^{\alpha}$$

> $\frac{\alpha}{2c} (\lambda_2 \widetilde{v}_2)^{1-\alpha} \lambda_2^{\alpha} (\widetilde{v}_2)^{\alpha}$
= $\frac{\alpha}{2c} \lambda_2 \widetilde{v}_2$
> $\frac{\alpha}{2c} \overline{v} = \overline{e}^s$

The equilibrium winning probabilities are

. . .

$$p_{21}^{s} = \frac{1}{2} \left(\frac{e_{2}^{s}}{e_{1}^{s}}\right)^{\alpha}$$

= $\frac{1}{2} \left(\frac{\lambda_{2} \widetilde{v}_{2}}{\lambda_{1}^{-1} \widetilde{v}_{1}}\right)^{\alpha}$
= $\frac{1}{2} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\alpha}$

$$p_{12}^{s} = 1 - p_{21}^{s}$$
$$= 1 - \frac{1}{2}\lambda_{1}^{\alpha}\lambda_{2}^{\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\alpha}$$

$$\widetilde{p}_{12}^{s} = 1 - \frac{1}{2} \frac{(e_2^{s})^{\alpha}}{\lambda_1 (e_1^{s})^{\alpha}} \\ = 1 - \frac{1}{2} \lambda_1^{\alpha - 1} \lambda_2^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_2^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{\alpha}$$

$$\begin{split} \widetilde{p}_{21}^{s} &= \frac{1}{2} \frac{\lambda_{2} \left(e_{2}^{s}\right)^{\alpha}}{\left(e_{1}^{s}\right)^{\alpha}} \\ &= \frac{1}{2} \lambda_{2} \left(\frac{\lambda_{2} \widetilde{v}_{2}}{\lambda_{1}^{-1} \widetilde{v}_{1}}\right)^{\alpha} \\ &= \frac{1}{2} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\alpha} \end{split}$$

Since $e_1^s > e_2^s$, we can get $p_{12}^s > \frac{1}{2} > p_{21}^s$. Since $\lambda_2 (e_2^s)^{\alpha} \leq (e_1^s)^{\alpha}$, we can get $\widetilde{p}_{21}^s \leq \frac{1}{2}$.

Thus in Proposition 7 (i) we have

$$\widetilde{p}_{12}^s > p_{12}^s > \frac{1}{2} \geqslant \widetilde{p}_{21}^s > p_{21}^s$$

(2) Proposition 7 (ii) and (iii): when $\frac{u(w_1)}{u(w_1)-u(w_2)}\frac{2}{1+\alpha} \ge \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}}-\lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}}-\lambda_1^{-1}}$ The equilibrium efforts are

$$e_1^s = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \lambda_2^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha+1}{2\alpha+1}} \\ \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_2^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \left[u(w_1) - u(w_2) \right]$$

$$e_{2}^{s} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \\ \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})]$$

and the efforts satisfy

$$e_{1}^{s} \begin{cases} > e_{2}^{s} > \overline{e}^{s} \quad \text{when} \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} \leqslant \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1+\alpha} < \frac{1}{\lambda_{1} - \lambda_{2}} \left(\frac{\lambda_{1}}{\lambda_{2}^{\frac{\alpha+1}{\alpha+1}}} - \frac{\lambda_{2}}{\lambda_{1}^{\frac{\alpha+1}{2\alpha+1}}} \right) \\ \leqslant e_{2}^{s} \quad \text{when} \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1+\alpha} \geqslant \max \left[\frac{\frac{\lambda_{1}}{\frac{\lambda_{2}^{\alpha+1}}{\lambda_{2}^{\frac{\alpha+1}{\alpha+1}}}}{\lambda_{1} - \lambda_{2}}, \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} \right] \end{cases}$$

$$(1) \text{ Proposition 7 (ii): when } \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}} \leqslant \frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1+\alpha} < \frac{1}{\lambda_1 - \lambda_2} \left(\frac{\lambda_1}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{\lambda_1^{\frac{\alpha+1}{2\alpha+1}}} \right)$$

We first show that $e_1^s > e_2^s > \overline{e}^s$ is satisfied under (1). Since $e_2^s = \frac{\alpha}{2c} \left(\lambda_1^{-1} \widetilde{v}_1\right)^{\frac{\alpha}{2\alpha+1}} \left(\lambda_2^{-1} \widetilde{v}_2\right)^{\frac{\alpha+1}{2\alpha+1}}$ and $e_1^s > e_2^s$, if both $\lambda_1^{-1} \widetilde{v}_1 > \overline{v}$ and $\lambda_2^{-1} \widetilde{v}_2 > \overline{v}$ are satisfied then we can get $e_1^s > e_2^s > \overline{e}^s$.

We show that under the condition of $\frac{u(w_1)}{u(w_1)-u(w_2)}\frac{2}{1+\alpha} < \frac{1}{\lambda_1-\lambda_2}\left(\frac{\lambda_1}{\lambda_2^{\alpha+1}} - \frac{\lambda_2}{\lambda_1^{2\alpha+1}}\right)$, both $\lambda_1^{-1}\widetilde{v}_1 > \overline{v}$ and $\lambda_2^{-1}\widetilde{v}_2 > \overline{v}$ are satisfied:

Let

$$f(\lambda_1) = \lambda_1^{-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right)$$

$$f'(\lambda_1) = -\lambda_1^{-2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right) + \lambda_1^{-1} \frac{1+\alpha}{2} \frac{\alpha+1}{2\alpha+1} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}-1}$$
$$= \lambda_1^{-2} \left[-\left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right) + \frac{1+\alpha}{2} \frac{\alpha+1}{2\alpha+1} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right]$$
$$= \lambda_1^{-2} \left[\frac{1+\alpha}{2} \left(\frac{\alpha+1}{2\alpha+1} + 1 \right) \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} - \frac{u(w_1)}{u(w_1) - u(w_2)} \right]$$

Let $g(\lambda_1) = \frac{1+\alpha}{2} \left(\frac{\alpha+1}{2\alpha+1} + 1\right) \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} - \frac{u(w_1)}{u(w_1) - u(w_2)}$, we can easily get $g'(\lambda_1) < 0$.

$$g(\lambda_1 = 1) = \frac{1+\alpha}{2} \left(\frac{\alpha+1}{2\alpha+1} + 1\right) - \frac{u(w_1)}{u(w_1) - u(w_2)}$$
$$g(\lambda_1 \to \infty) = -\frac{u(w_1)}{u(w_1) - u(w_2)} < 0$$

(a) When
$$\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$$

If $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$, then $g(\lambda_1 = 1) < 0$, $g(\lambda_1) \leq 0$ is always satisfied and thus $f'(\lambda_1) < 0$ always holds. Therefore $f(\lambda_1) < f(\lambda_2) < f(1) = \overline{v}$. This contradicts the condition $\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} < \frac{1}{\lambda_1-\lambda_2} \left(\frac{\lambda_1}{\lambda_2^{\frac{\alpha+1}{\alpha+1}}} - \frac{\lambda_2}{\lambda_1^{\frac{\alpha+1}{2\alpha+1}}}\right)$ since we showed earlier that this condition is equivalent to $f(\lambda_1) > f(\lambda_2)$ and $e_1^s > e_2^s$.
b) When
$$\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$$

If $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$, then $g(\lambda_1 = 1) > 0$ and thus
 $g(\lambda_1)$ and $f'(\lambda_1) \begin{cases} > 0 \quad \text{when } \lambda_1 < \left(\frac{\frac{1+\alpha}{2}\left(\frac{\alpha+1}{2\alpha+1}+1\right)}{\frac{u(w_1)}{u(w_1)-u(w_2)}}\right)^{\frac{2\alpha+1}{\alpha+1}} \\ < 0 \quad \text{when } \lambda_1 > \left(\frac{\frac{1+\alpha}{2}\left(\frac{\alpha+1}{2\alpha+1}+1\right)}{\frac{u(w_1)}{u(w_1)-u(w_2)}}\right)^{\frac{2\alpha+1}{\alpha+1}} \\ f(\lambda_1) \begin{cases} > \overline{v} \quad \text{when } \lambda_1 < \hat{\lambda} \\ = \overline{v} \quad \text{when } \lambda_1 > \hat{\lambda} \end{cases}$

where $\hat{\lambda}$ is as derived in the proposition 3.

(

When $f(\lambda_1) > f(\lambda_2)$ is satisfied, $f(\lambda_1) > f(\lambda_2) > f(1) = f(\hat{\lambda}) = \overline{v}$ must be true. If $f(\lambda_2) < \overline{v}$, then $\lambda_2 > \hat{\lambda}$. And since $f'(\lambda_1) < 0$ for $\forall \lambda_1 > \hat{\lambda}$, $f(\lambda_1) < f(\lambda_2)$ when $f(\lambda_2) < \overline{v}$. This contradicts $f(\lambda_1) > f(\lambda_2)$. Thus when the condition $\frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1 + \alpha} < \frac{1}{\lambda_1 - \lambda_2} \left(\frac{\lambda_1}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{\lambda_1^{\frac{\alpha+1}{2\alpha+1}}} \right)$ is satisfied, $e_1 > e_2 > \overline{e}^s$ is always true.

The true equilibrium winning probabilities under (1) are

$$p_{21}^{s} = \frac{1}{2} \left(\frac{e_{2}^{s}}{e_{1}^{s}}\right)^{\alpha}$$

$$= \frac{1}{2} \left(\frac{\lambda_{2}^{-1} \widetilde{v}_{2}}{\lambda_{1}^{-1} \widetilde{v}_{1}}\right)^{\frac{\alpha}{2\alpha+1}}$$

$$= \frac{1}{2} \lambda_{1}^{\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\frac{\alpha}{2\alpha+1}}$$

$$\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\frac{\alpha}{2\alpha+1}}$$

$$p_{12}^{s} = 1 - p_{21}^{s}$$

$$= 1 - \frac{1}{2}\lambda_{1}^{\frac{\alpha}{2\alpha+1}}\lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\frac{\alpha}{2\alpha+1}}$$

$$\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\frac{\alpha}{2\alpha+1}}$$

Since $e_1^s > e_2^s$, we can get $p_{12}^s > \frac{1}{2} > p_{21}^s$. Thus we have $\widetilde{p}_{12}^s > p_{12}^s > \frac{1}{2} > p_{21}^s$.

(2) Proposition 7 (iii): when $\frac{u(w_1)}{u(w_1)-u(w_2)}\frac{2}{1+\alpha} \ge \max\left[\frac{\frac{\lambda_1}{\alpha_2^{\alpha+1}} - \frac{\lambda_2}{\alpha_1^{\alpha+1}}}{\lambda_1^{-\alpha+1}}, \frac{\lambda_2^{(\alpha+1)^2}}{\lambda_2^{(\alpha+1)} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\alpha-1} - \lambda_1^{-1}}\right]$ The true equilibrium winning probabilities are

$$\begin{split} p_{12}^{s} &= \frac{1}{2} \left(\frac{e_{1}^{s}}{e_{2}^{s}} \right)^{\alpha} \\ &= \frac{1}{2} \left(\frac{\lambda_{1}^{-1} \widetilde{v}_{1}}{\lambda_{2}^{-1} \widetilde{v}_{2}} \right)^{\frac{\alpha}{2\alpha+1}} \\ &= \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \\ &\qquad \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\frac{\alpha}{2\alpha+1}} \end{split}$$

$$p_{21}^{s} = 1 - p_{12}^{s}$$

$$= 1 - \frac{1}{2}\lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\lambda_{2}^{\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{\frac{\alpha}{2\alpha+1}}$$

$$\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\frac{\alpha}{2\alpha+1}}$$

Since $e_1^s \leqslant e_2^s$, we can get $p_{21}^s \ge \frac{1}{2} \ge p_{12}^s$. The perceived equilibrium winning probabilities under (2) are:

$$\begin{split} \widetilde{p}_{12}^{s} &= 1 - \frac{1}{2} \frac{\left(e_{2}^{s}\right)^{\alpha}}{\lambda_{1} \left(e_{1}^{s}\right)^{\alpha}} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-1} \left(\lambda_{1}^{\frac{1}{2\alpha+1}} \lambda_{2}^{-\frac{1}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\frac{1}{2\alpha+1}} \\ &\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{1}{2\alpha+1}} \right)^{\alpha} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\frac{\alpha}{2\alpha+1}} \\ &\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \end{split}$$

$$\begin{split} \widetilde{p}_{21}^{s} &= 1 - \frac{1}{2} \frac{\left(e_{1}^{s}\right)^{\alpha}}{\lambda_{2} \left(e_{2}^{s}\right)^{\alpha}} \\ &= 1 - \frac{1}{2} \lambda_{2}^{-1} \left(\lambda_{1}^{-\frac{1}{2\alpha+1}} \lambda_{2}^{\frac{1}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{1}{2\alpha+1}} \right) \\ &\quad \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\frac{1}{2\alpha+1}} \right)^{\alpha} \\ &= 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \\ &\quad \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\frac{\alpha}{2\alpha+1}} \end{split}$$

We show that $\widetilde{p}_{12}^s > \widetilde{p}_{21}^s$:

$$\begin{split} \widetilde{p}_{12}^{s} &> \widetilde{p}_{21}^{s} \\ &\iff 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{2} \right)^{\frac{\alpha}{2\alpha+1}} > 1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{2} \right)^{-\frac{\alpha}{2\alpha+1}} \\ &\iff \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{2} \right)^{-\frac{\alpha}{2\alpha+1}} > \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{2} \right)^{\frac{\alpha}{2\alpha+1}} \\ &\iff \lambda_{1}^{\frac{1}{2\alpha+1}} \lambda_{2}^{-\frac{1}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{2\alpha}{2\alpha+1}} \left(\widetilde{v}_{2} \right)^{-\frac{2\alpha}{2\alpha+1}} > 1 \\ &\iff \frac{\lambda_{1}^{\frac{1}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{2\alpha}{2\alpha+1}}}{\lambda_{2}^{\frac{1}{2\alpha+1}} \left(\widetilde{v}_{2} \right)^{\frac{2\alpha}{2\alpha+1}}} > 1 \end{split}$$

Thus in Proposition 7 (ii) we have $\tilde{p}_{12}^s > p_{12}^s > \frac{1}{2} > p_{21}^s$ and in Proposition 7 (iii) we have $\tilde{p}_{12}^s > \tilde{p}_{21}^s > p_{21}^s \geqslant \frac{1}{2} \geqslant p_{12}^s$.

3. Participation constraint

$$\begin{array}{l} \text{(1) When } \frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1+\alpha} \leqslant \frac{\lambda_2^{\frac{(\alpha+1)^2}{2(\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}} \\ \widetilde{E}^s(U_{12}) = \widetilde{p}_{12}^s \widetilde{v}_1 - c e_1^s \\ > p_{12}^s \widetilde{v}_1 - c e_1^s \\ = \left(1 - \frac{1}{2}\lambda_1^{\alpha}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{-\alpha} \left(\widetilde{v}_2\right)^{\alpha}\right) \widetilde{v}_1 - c\frac{\alpha}{2c}\lambda_1^{\alpha-1}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{1-\alpha} \left(\widetilde{v}_2\right)^{\alpha} \\ = \widetilde{v}_1 - \frac{1}{2}\lambda_1^{\alpha}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{1-\alpha} \left(\widetilde{v}_2\right)^{\alpha} - \frac{\alpha}{2}\lambda_1^{\alpha-1}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{1-\alpha} \left(\widetilde{v}_2\right)^{\alpha} \\ > \widetilde{v}_1 - \frac{1}{2}\lambda_1^{\alpha}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{1-\alpha} \left(\widetilde{v}_2\right)^{\alpha} - \frac{\alpha}{2}\lambda_1^{\alpha}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{1-\alpha} \left(\widetilde{v}_2\right)^{\alpha} \\ = \widetilde{v}_1 - \frac{1+\alpha}{2}\lambda_1^{\alpha}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{1-\alpha} \left(\widetilde{v}_2\right)^{\alpha} \\ = \widetilde{v}_1 - \frac{1+\alpha}{2}\lambda_1^{\alpha}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{1-\alpha} \left(\widetilde{v}_2\right)^{\alpha} \\ = \widetilde{v}_1 \left[1 - \frac{1+\alpha}{2}\lambda_1^{\alpha}\lambda_2^{\alpha} \left(\widetilde{v}_1\right)^{-\alpha} \left(\widetilde{v}_2\right)^{\alpha}\right] \\ = \widetilde{v}_1 \left[1 - \frac{1+\alpha}{2}\left(\frac{e_2^s}{e_1^s}\right)^{\alpha}\right] \\ > 0 \end{array}$$

$$\begin{aligned} \widetilde{E}^{s}(U_{21}) &= \widetilde{p}_{21}^{s} \widetilde{v}_{2} - c e_{2}^{s} \\ &= \frac{1}{2} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \left(\widetilde{v}_{1} \right)^{-\alpha} \left(\widetilde{v}_{2} \right)^{\alpha} \widetilde{v}_{2} - c \frac{\alpha}{2c} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \left(\widetilde{v}_{1} \right)^{-\alpha} \left(\widetilde{v}_{2} \right)^{\alpha+1} \\ &= \frac{1}{2} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \left(\widetilde{v}_{1} \right)^{-\alpha} \left(\widetilde{v}_{2} \right)^{1+\alpha} - \frac{\alpha}{2} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \left(\widetilde{v}_{1} \right)^{-\alpha} \left(\widetilde{v}_{2} \right)^{\alpha+1} \\ &= \frac{1-\alpha}{2} \lambda_{1}^{\alpha} \lambda_{2}^{\alpha+1} \left(\widetilde{v}_{1} \right)^{-\alpha} \left(\widetilde{v}_{2} \right)^{1+\alpha} \\ &\geqslant 0 \end{aligned}$$

(2) When
$$\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} \ge \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}}$$

$$\begin{split} \widetilde{E}^{s}(U_{12}) &= \widetilde{p}_{12}^{s} \widetilde{v}_{1} - c e_{1}^{s} \\ &= \left(1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha}{2\alpha+1}} \right) \widetilde{v}_{1} - c \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha}{2\alpha+1}} \\ &= \widetilde{v}_{1} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha}{2\alpha+1}} \\ &= (\widetilde{v}_{1})^{\frac{\alpha+1}{2\alpha+1}} \left[(\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha}{2\alpha+1}} \right] \\ &> 0 \end{split}$$

$$\begin{split} \widetilde{E}^{s}(U_{21}) &= \widetilde{p}_{21}^{s} \widetilde{v}_{1} - ce_{2}^{s} \\ &= \left(1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right) \widetilde{v}_{2} - c \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha+1}{2\alpha+1}} \\ &= \widetilde{v}_{2} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{\frac{\alpha+1}{2\alpha+1}} \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right] \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right] \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right] \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right] \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right] \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right] \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right] \\ &= \widetilde{v}_{2} \left[1 - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{2})^{-\frac{\alpha}{2\alpha+1}} \right]$$

Lemma 4

$$\lim_{\lambda_{1}\to\infty} \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} = \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}$$
$$\lim_{\lambda_{1}\to\lambda_{2}} \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)}} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha}} - \lambda_{1}^{-1}} = \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}$$
$$\lim_{\lambda_{1}\to\infty} \frac{\frac{\lambda_{1}}{\alpha^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_{2}}{\lambda_{1}^{\frac{\alpha+1}{2\alpha+1}}}}{\lambda_{1} - \lambda_{2}} = \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}$$
$$\lim_{\lambda_{1}\to\lambda_{2}} \frac{\frac{\lambda_{1}}{\alpha^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_{2}}{\alpha^{\frac{\alpha+1}{2\alpha+1}}}}{\lambda_{1} - \lambda_{2}} = \frac{3\alpha+2}{2\alpha+1}\lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}}$$

Proof of Lemma 4

$$\begin{split} \lim_{\lambda_1 \to \infty} \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}} &= \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}}}{\lambda_2^{\frac{\alpha+1}{\alpha}}} = \lambda_2^{-\frac{\alpha+1}{2\alpha+1}} \\ \lim_{\lambda_1 \to \lambda_2} \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}} \\ &= \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_2^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_2^{-1}} \\ &= \frac{\lambda_2^{-\frac{3\alpha+2}{2\alpha+1}} \left(\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)} + \frac{3\alpha+2}{2\alpha+1}} - 1\right)}{\lambda_2^{-1} \left(\lambda_2^{\frac{\alpha+1}{\alpha} + 1} - 1\right)} \\ &= \lambda_2^{-\frac{3\alpha+2}{2\alpha+1} + 1} \frac{\lambda_2^{\frac{2\alpha+1}{\alpha}} - 1}}{\lambda_2^{\frac{2\alpha+1}{\alpha}} - 1} \\ &= \lambda_2^{-\frac{\alpha+1}{2\alpha+1}} \end{split}$$

$$\lim_{\lambda_1 \to \infty} \frac{\frac{\lambda_1}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{\lambda_1^{\frac{\alpha+1}{2\alpha+1}}}}{\lambda_1 - \lambda_2} = \lim_{\lambda_1 \to \infty} \frac{\frac{1}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{\lambda_1^{\frac{\alpha+1}{2\alpha+1}+1}}}{1 - \frac{\lambda_2}{\lambda_1^{\frac{\alpha+1}{2\alpha+1}+1}}}{1 - \frac{\lambda_2}{\lambda_1}}$$
$$= \frac{\lambda_2^{-\frac{\alpha+1}{2\alpha+1}} - 0}{1 - 0}$$
$$= \lambda_2^{-\frac{\alpha+1}{2\alpha+1}}$$

Let $t = \lambda_1 - \lambda_2$,

$$\lim_{\lambda_1 \to \lambda_2} \frac{\frac{\lambda_1}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{\lambda_1^{\frac{\alpha+1}{2\alpha+1}}}}{\lambda_1 - \lambda_2} = \lim_{t \to 0} \frac{\frac{\frac{\lambda_2 + t}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{(\lambda_2 + t)^{\frac{\alpha+1}{2\alpha+1}}}}{t}}{t}$$
$$= \lim_{t \to 0} \frac{\frac{\partial \left(\frac{\lambda_2 + t}{\frac{\alpha+1}{\lambda_2^{\frac{\alpha+1}{2\alpha+1}}} - \frac{\lambda_2}{(\lambda_2 + t)^{\frac{\alpha+1}{2\alpha+1}}}\right)}{\frac{\partial t}{\frac{\partial t}{\frac{\partial t}{2}}}}{\frac{\partial t}{\partial t}}$$
$$= \lim_{t \to 0} \frac{\lambda_2^{-\frac{\alpha+1}{2\alpha+1}} - \lambda_2 \left(-\frac{\alpha+1}{2\alpha+1}\right) (\lambda_2 + t)^{-\frac{\alpha+1}{2\alpha+1} - 1}}{1}}{1}$$
$$= \left(\frac{3\alpha + 2}{2\alpha + 1}\right) \lambda_2^{-\frac{\alpha+1}{2\alpha+1}}$$

Lemma 5

Lemma 5 In the semifinal between two rational players of a two-stage elimination contest where the overconfident players 1 and 2 are seeded in one semifinal, the rational players

3 and 4 are seeded in the other semifinal, and $\lambda_1 > \lambda_2 > 1 = \lambda_3 = \lambda_4$, the equilibrium efforts and winning probabilities satisfy $e_3^s = e_4^s > \overline{e}^s$ and $p_{34}^s = p_{43}^s = 1/2$.

Proof of Lemma 5

1. Continuation values

Rational player 3:

$$\begin{aligned} v_{3} &= p_{12}^{s} E^{f}(U_{31}) + p_{21}^{s} E^{f}(U_{32}) \\ &= \left[p_{12}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \right) + p_{21}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \right) \right] \\ &\times \left[u(w_{1}) - u(w_{2}) \right] \\ &= \left[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(\lambda_{2}^{-\frac{\alpha}{2\alpha+1}} + p_{12}^{s} \left(\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} - \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \right) \right) \right] \left[u(w_{1}) - u(w_{2}) \right] \\ &> \overline{v} \end{aligned}$$

where p_{12}^s is as derived in the proof of proposition 7.

Rational player 4:

Since player 3 and player 4 are identical,

$$v_4 = v_3 > \overline{v}$$

2. The equilibrium

$$(1) \text{ When } \frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} \leqslant \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha+1}{\alpha}} - \lambda_1^{-1}} \\ e_3^s = e_4^s = \frac{\alpha}{2c} v_3 \\ = \frac{\alpha}{2c} \left[\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \left(\lambda_2^{-\frac{\alpha}{2\alpha+1}} + \left(1 - \frac{1}{2} \lambda_1^{\alpha} \lambda_2^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \right) \right. \\ \left. \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_2^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\alpha} \right) \left(\lambda_1^{-\frac{\alpha}{2\alpha+1}} - \lambda_2^{-\frac{\alpha}{2\alpha+1}} \right) \right) \right] [u(w_1) - u(w_2)] \\ p_{34}^s = p_{43}^s = \frac{1}{2} \\ (2) \text{ When } \frac{\lambda_2^{\frac{(\alpha+1)^2}{\alpha(2\alpha+1)}} - \lambda_1^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_2^{\frac{\alpha}{\alpha} - \lambda_1^{-1}}} \leqslant \frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1+\alpha} < \frac{\lambda_2^{\frac{\alpha}{2\alpha+1}} - \frac{\lambda_2^{-\frac{\alpha}{2\alpha+1}}}{\lambda_1 - \lambda_2}}{\lambda_1 - \lambda_2} \end{aligned}$$

$$\begin{split} e_{3}^{s} &= e_{4}^{s} = \frac{\alpha}{2c} v_{3} \\ &= \frac{\alpha}{2c} \bigg[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(\lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \right. \\ &+ \left(1 - \frac{1}{2} \lambda_{1}^{\frac{2\alpha}{2\alpha+1}} \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\frac{\alpha}{2\alpha+1}} \right)^{-\frac{\alpha}{2\alpha+1}} \\ &\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \right)^{\frac{\alpha}{2\alpha+1}} \right) \left(\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} - \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \right) \right) \bigg] [u(w_{1}) - u(w_{2})] \\ &p_{34}^{s} = p_{3}^{s} = \frac{1}{2} \\ \end{split}$$
(3) When $\frac{u(w_{1})}{u(w_{1}) - u(w_{2})^{\frac{1}{1+\alpha}}} \ge \max \left[\frac{\frac{\lambda_{1}^{\frac{\alpha}{1+1}} - \frac{\lambda_{1}^{\frac{\alpha}{2\alpha+1}}}{\lambda_{2}^{\frac{1}{\alpha+1}}}}{\lambda_{1}^{-\lambda_{2}^{\frac{\alpha}{2\alpha+1}}}}, \frac{\lambda_{2}^{\frac{(\alpha+1)^{2}}{\alpha(2\alpha+1)} - \lambda_{1}^{-\frac{3\alpha+2}{2\alpha+1}}}{\lambda_{2}^{\frac{\alpha+1}{\alpha} - \lambda_{1}^{-1}}} \right] \\ e_{3}^{s} = e_{4}^{s} = \frac{\alpha}{2c} v_{3} \\ &= \frac{\alpha}{2c} \bigg[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \bigg(\lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \\ &+ \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{2}^{\frac{\alpha}{2\alpha+1}} \bigg(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \bigg)^{\frac{\alpha}{2\alpha+1}} \\ &\left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{2}^{-\frac{\alpha+1}{2\alpha+1}} \bigg)^{-\frac{\alpha}{2\alpha+1}} \bigg(\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} - \lambda_{2}^{-\frac{\alpha}{2\alpha+1}} \bigg) \bigg] \bigg] [u(w_{1}) - u(w_{2})] \\ &p_{34}^{s} = p_{43}^{s} = \frac{1}{2} \end{split}$

Since $v_3 = v_4 > \overline{v}$, $e_3^s = e_4^s > \overline{e}^s$ is always satisfied.

3. Participation constraint

$$E^{s}(U_{34}) = p_{34}^{s}v_{3} - ce_{3}$$
$$= \frac{1-\alpha}{2}v_{3}$$
$$\geqslant 0$$

$$E^{s}(U_{43}) = E^{s}(U_{34}) \ge 0$$

Proof of Proposition 8

1. Continuation values of each player

Overconfident player 1:

$$\begin{split} \widetilde{v}_{1} &= p_{34}^{s} \widetilde{E}^{f}(U_{13}) + p_{43}^{s} \widetilde{E}^{f}(U_{14}) \\ &= p_{34}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}} \lambda_{3}^{-\frac{\alpha}{2\alpha + 1}} \right) \left[u(w_{1}) - u(w_{2}) \right] \\ &+ p_{43}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}} \right) \left[u(w_{1}) - u(w_{2}) \right] \\ &= \left[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{34}^{s} + p_{34}^{s} \lambda_{3}^{-\frac{\alpha}{2\alpha + 1}} \right) \lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}} \right] \left[u(w_{1}) - u(w_{2}) \right] \end{split}$$

Rational player 2:

$$v_{2} = p_{34}^{s} E^{f}(U_{23}) + p_{43}^{s} E^{f}(U_{24})$$

$$= p_{34}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}}\right) [u(w_{1}) - u(w_{2})]$$

$$+ p_{43}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\right) [u(w_{1}) - u(w_{2})]$$

$$= \left[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{34}^{s} + p_{34}^{s} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}}\right)\right] [u(w_{1}) - u(w_{2})]$$

Overconfident player 3:

$$\begin{split} \widetilde{v}_{3} &= p_{12}^{s} \widetilde{E}^{f}(U_{31}) + p_{21}^{s} \widetilde{E}^{f}(U_{32}) \\ &= p_{12}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right) \left[u(w_{1}) - u(w_{2}) \right] \\ &+ p_{21}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right) \left[u(w_{1}) - u(w_{2}) \right] \\ &= \left[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{12}^{s} + p_{12}^{s} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \right) \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right] \left[u(w_{1}) - u(w_{2}) \right] \end{split}$$

Rational player 4:

$$v_{4} = p_{12}^{s} E^{f}(U_{41}) + p_{21}^{s} E^{f}(U_{42})$$

$$= p_{12}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\right) [u(w_{1}) - u(w_{2})]$$

$$+ p_{21}^{s} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2}\right) [u(w_{1}) - u(w_{2})]$$

$$= \left[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{12}^{s} + p_{12}^{s} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}}\right)\right] [u(w_{1}) - u(w_{2})]$$

2. The equilibrium of the semifinal between player 1 and player 2

Player 1 max $\widetilde{E}^s(U_{12}) = \widetilde{p}_{12}^s \widetilde{v}_1 - ce_1$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right) \widetilde{v}_1 - c e_1 & \text{if} \quad \lambda_1 e_1^{\alpha} \ge e_2^{\alpha} \\ \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_2^{\alpha}} \widetilde{v}_1 - c e_1 & \text{if} \quad \lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha} \end{cases}$$

Player 2 max
$$E^{s}(U_{21}) = p_{21}^{s}v_{2} - ce_{2}$$

$$= \begin{cases} \left(1 - \frac{1}{2}\frac{e_{1}^{\alpha}}{e_{2}^{\alpha}}\right)v_{2} - ce_{2} & \text{if } e_{2} \ge e_{1} \\ \frac{1}{2}\frac{e_{2}^{\alpha}}{e_{1}^{\alpha}}v_{2} - ce_{2} & \text{if } e_{2} \le e_{1} \end{cases}$$

There are 4 cases.

 $\begin{cases} \lambda_1 e_1^{\alpha} \geqslant e_2^{\alpha} & and \quad e_2 \leqslant e_1 \\ \lambda_1 e_1^{\alpha} \geqslant e_2^{\alpha} & and \quad e_2 \geqslant e_1 \\ \lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha} & and \quad e_2 \geqslant e_1 \\ \lambda_1 e_1^{\alpha} \leqslant e_2^{\alpha} & and \quad e_2 \leqslant e_1 \end{cases}$

Since $\lambda_1 > 1$, the fourth case is impossible.

(1) case 1: $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_2 \le e_1$ Player 1 $max \left(1 - \frac{1}{2}\frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right)\widetilde{v}_1 - ce_1$ Player 2 $max \quad \frac{1}{2}\left(\frac{e_2}{e_1}\right)^{\alpha}v_2 - ce_2$ F.o.c $[e_1] \quad \frac{\alpha}{2\lambda_1}\frac{e_2^{\alpha}}{e_1^{\alpha+1}}\widetilde{v}_1 - c = 0$ $[e_2] \quad \frac{\alpha}{2}\frac{e_2^{\alpha-1}}{e_1^{\alpha}}v_2 - c = 0$

S.o.c

- $[e_1] \quad \frac{\alpha}{2\lambda_1}(-\alpha-1)\frac{e_2^{\alpha}}{e_1^{\alpha+2}}\widetilde{v}_1 < 0$
- $[e_2] \quad \frac{\alpha}{2}(\alpha 1)\frac{e_2^{\alpha 2}}{e_1^{\alpha}}v_2 < 0$

Solve the two F.O.C , we get

$$e_{1} = \frac{\alpha}{2c} \lambda_{1}^{\alpha-1} (\widetilde{v}_{1})^{1-\alpha} (v_{2})^{\alpha}$$

$$e_{2} = \frac{\alpha}{2c} \lambda_{1}^{\alpha} (\widetilde{v}_{1})^{-\alpha} (v_{2})^{\alpha+1}$$

$$\frac{e_{2}}{e_{1}} = \lambda_{1} \frac{v_{2}}{\widetilde{v}_{1}}$$

$$p_{21}^{s} = \frac{1}{2} \left(\frac{e_{2}}{e_{1}}\right)^{\alpha}$$

$$= \frac{1}{2} \left(\frac{\lambda_{1} v_{2}}{\widetilde{v}_{1}}\right)^{\alpha}$$

Check the conditions $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_1 \ge e_2$:

As long as $e_1 \ge e_2$ is satisfied, $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ is satisfied. So we only have to check $e_1 \ge e_2$.

$$e_{2} \leqslant e_{1} \iff \frac{e_{2}}{e_{1}} \leqslant 1$$

$$\iff \frac{1}{2} \left(\frac{e_{2}}{e_{1}}\right)^{\alpha} \leqslant \frac{1}{2}$$

$$\iff \frac{e_{2}}{e_{1}} \leqslant 1$$

$$\iff \frac{v_{2}}{\lambda_{1}^{-1} \widetilde{v}_{1}} \leqslant 1$$

$$\iff \frac{\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{34}^{s} + p_{34}^{s} \lambda_{3}^{-\frac{\alpha}{2\alpha + 1}}\right)}{\lambda_{1}^{-1} \left[\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{34}^{s} + p_{34}^{s} \lambda_{3}^{-\frac{\alpha}{2\alpha + 1}}\right) \lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}}\right]} \leqslant 1$$

Let

$$f(p_{34}^s) = \lambda_1^{-1} \left[\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \left(1 - p_{34}^s + p_{34}^s \lambda_3^{-\frac{\alpha}{2\alpha+1}} \right) \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right] \\ - \left[\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \left(1 - p_{34}^s + p_{34}^s \lambda_3^{-\frac{\alpha}{2\alpha+1}} \right) \right]$$

Rearrange the terms we can get

$$\begin{split} f(p_{34}^s) &= \lambda_1^{-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right) - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \\ &+ \frac{1 + \alpha}{2} \left(\lambda_3^{-\frac{\alpha}{2\alpha + 1}} - 1 \right) \left(1 - \lambda_1^{-1 - \frac{\alpha + 1}{2\alpha + 1}} \right) p_{34}^s \end{split}$$

Since $\left(\lambda_3^{-\frac{\alpha}{2\alpha+1}}-1\right)\left(1-\lambda_1^{-1-\frac{\alpha+1}{2\alpha+1}}\right)<0$ and $p_{34}^s\in[0,1], f(p_{34}^s)$ reaches minimum at $p_{34}^s=1$. Thus, $e_2\leqslant e_1$ is always satisfied as long as $f(p_{34}^s=1)\geqslant 0$.

$$f(p_{34}^s = 1) = \lambda_1^{-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right) - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{1 + \alpha}{2} \left(\lambda_3^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_1^{-1 - \frac{\alpha+1}{2\alpha+1}} \right)$$

$$\begin{split} \lambda_{1}^{-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) - \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right) \\ &+ \frac{1 + \alpha}{2} \left(\lambda_{3}^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_{1}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) \geq 0 \\ \Leftrightarrow \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) - \lambda_{1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right) \\ &+ \lambda_{1} \frac{1 + \alpha}{2} \left(\lambda_{3}^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_{1}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) \geq 0 \\ \Leftrightarrow \frac{1 + \alpha}{2} \left[\lambda_{1} \left(1 + \left(\lambda_{3}^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_{1}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) \right) - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right] \geq (\lambda_{1} - 1) \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \\ \Leftrightarrow \frac{1}{\lambda_{1} - 1} \left[\lambda_{1} \left(\lambda_{1}^{-1 - \frac{\alpha+1}{2\alpha+1}} + \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} - \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} \lambda_{1}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right] \geq \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \\ \Leftrightarrow \frac{1}{\lambda_{1} - 1} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} \left(\lambda_{1} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} \right) \geq \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \\ \text{To ensure } e_{1} > e_{2} \text{ and } p_{12}^{s} > p_{21}^{s}, \text{ we need } \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \\ \text{Since } \frac{\lambda_{1} - \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}}{\lambda_{1} - 1} \text{ decreasing in } \lambda_{1} \text{ for } \lambda_{1} > 1 \text{ and its limit when } \lambda_{1} \to 1 \text{ is } \frac{3\alpha+2}{2\alpha+1}, \end{split}$$

we can get
$$\frac{1}{\lambda_1 - 1} \lambda_3^{-\frac{\alpha}{2\alpha + 1}} \left(\lambda_1 - \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right) < \frac{3\alpha + 2}{2\alpha + 1}$$
. Hence, to satisfy the inequal-
ity $\frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1 + \alpha} < \frac{1}{\lambda_1 - 1} \lambda_3^{-\frac{\alpha}{2\alpha + 1}} \left(\lambda_1 - \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right)$, we also need $\frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1 + \alpha} < \frac{3\alpha + 2}{2\alpha + 1}$, which is equivalent to $\frac{u(w_1) - u(w_2)}{u(w_2)} > \frac{2(2\alpha + 1)}{\alpha(3\alpha + 1)}$.
(2) case 2: $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_2 \ge e_1$

Player 1
$$max \left(1 - \frac{1}{2}\frac{e_2^{\alpha}}{\lambda_1 e_1^{\alpha}}\right)\widetilde{v}_1 - ce_1$$

Player 2 $max \left(1 - \frac{1}{2}\frac{e_1^{\alpha}}{e_2^{\alpha}}\right)v_2 - ce_2$

F.o.c

$$\begin{bmatrix} e_1 \end{bmatrix} \quad \frac{\alpha}{2\lambda_1} \frac{e_2^{\alpha}}{e_1^{\alpha+1}} \widetilde{v}_1 - c = 0$$
$$\begin{bmatrix} e_2 \end{bmatrix} \quad \frac{\alpha}{2} \frac{e_1^{\alpha}}{e_2^{\alpha+1}} v_2 - c = 0$$

S.o.c

$$[e_1] \quad \frac{\alpha}{2\lambda_1}(-\alpha-1)\frac{e_2^{\alpha}}{e_1^{\alpha+2}}\widetilde{v}_1 < 0$$
$$[e_2] \quad \frac{\alpha}{2}(-\alpha-1)\frac{e_1^{\alpha}}{e_2^{\alpha+2}}v_2 < 0$$

Solve F.O.C , we get

$$e_1 = \frac{\alpha}{2c} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_1)^{\frac{\alpha+1}{2\alpha+1}} (v_2)^{\frac{\alpha}{2\alpha+1}}$$

$$e_{2} = \frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{1})^{\frac{\alpha}{2\alpha+1}} (v_{2})^{\frac{\alpha+1}{2\alpha+1}}$$
$$\frac{e_{2}}{e_{1}} = \lambda_{1}^{\frac{1}{2\alpha+1}} (\widetilde{v}_{1})^{-\frac{1}{2\alpha+1}} (v_{2})^{\frac{1}{2\alpha+1}}$$

Check the conditions $\lambda_1 e_1^{\alpha} \ge e_2^{\alpha}$ and $e_2 \ge e_1$: $(1) \ \lambda_1 e_1^{\alpha} \geqslant e_2^{\alpha}$

$$\lambda_1 e_1^{\alpha} \ge e_2^{\alpha} \iff \frac{\lambda_1 e_1^{\alpha}}{e_2^{\alpha}} \ge 1$$
$$\iff \lambda_1^{\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_1)^{\frac{\alpha}{2\alpha+1}} (v_2)^{-\frac{\alpha}{2\alpha+1}} \ge 1$$

Since $\lambda_1 > 1$ and $\tilde{v}_1 > v_2$, $\lambda_1 e_1^{\alpha} > e_2^{\alpha}$ is always satisfied. (2) $e_2 \ge e_1$ $e_2 \ge e_1$ is always satisfied as long as $f(p_{34}^s = 0) \le 0$.

$$f(p_{34}^s = 0) = \lambda_1^{-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right) - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)$$

$$f(p_{34}^s = 0) \leqslant 0 \iff \frac{u(w_1)}{u(w_1) - u(w_2)} \left(\lambda_1^{-1} - 1\right) + \frac{1 + \alpha}{2} \left(1 - \lambda_1^{-1 - \frac{\alpha + 1}{2\alpha + 1}}\right) \leqslant 0$$
$$\iff \frac{1 + \alpha}{2} \left(1 - \lambda_1^{-1 - \frac{\alpha + 1}{2\alpha + 1}}\right) \leqslant \frac{u(w_1)}{u(w_1) - u(w_2)} \left(1 - \lambda_1^{-1}\right)$$
$$\iff \frac{1 - \lambda_1^{-1 - \frac{\alpha + 1}{2\alpha + 1}}}{1 - \lambda_1^{-1}} \leqslant \frac{u(w_1)}{u(w_1) - u(w_2)} \frac{2}{1 + \alpha}$$

(3) case 3: $\lambda_1 e_1^{\alpha} \leq e_2^{\alpha}$ and $e_2 \geq e_1$

Player 1
$$max \quad \frac{1}{2} \frac{\lambda_1 e_1^{\alpha}}{e_2^{\alpha}} \widetilde{v}_1 - ce_1$$

Player 2 $max \quad \left(1 - \frac{1}{2} \left(\frac{e_1}{e_2}\right)^{\alpha}\right) v_2 - ce_2$

F.o.c

$$\begin{bmatrix} e_1 \end{bmatrix} \quad \frac{\alpha \lambda_1}{2} \frac{e_1^{\alpha - 1}}{e_2^{\alpha}} \widetilde{v}_1 - c = 0$$
$$\begin{bmatrix} e_2 \end{bmatrix} \quad \frac{\alpha}{2} \frac{e_1^{\alpha}}{e_2^{\alpha + 1}} v_2 - c = 0$$

divide the two F.O.C , we get

$$\frac{e_2}{e_1} = \frac{v_2}{\lambda_1 \widetilde{v}_1} < 1$$

which contradicts the condition that $e_2 \ge e_1$

3. The equilibrium between player 3 and player 4

Player 3 max $\widetilde{E}^s(U_{34}) = \widetilde{p}^s_{34}\widetilde{v}_3 - ce_3$

$$= \begin{cases} \left(1 - \frac{1}{2} \frac{e_4^{\alpha}}{\lambda_3 e_3^{\alpha}}\right) \widetilde{v}_3 - ce_3 & \text{if} \quad \lambda_3 e_3^{\alpha} \geqslant e_4^{\alpha} \\ \frac{1}{2} \frac{\lambda_3 e_3^{\alpha}}{e_4^{\alpha}} \widetilde{v}_3 - ce_3 & \text{if} \quad \lambda_3 e_3^{\alpha} \leqslant e_4^{\alpha} \end{cases}$$

Player 4 max $E^{s}(U_{43}) = p_{43}^{s}v_{4} - ce_{4}$ $= \begin{cases} \left(1 - \frac{1}{2}\frac{e_{3}^{\alpha}}{e_{4}^{\alpha}}\right)v_{4} - ce_{4} & \text{if } e_{4} \ge e_{3} \\ \frac{1}{2}\frac{e_{4}^{\alpha}}{e_{3}^{\alpha}}v_{4} - ce_{4} & \text{if } e_{4} \le e_{3} \end{cases}$

There are 4 cases.

 $\begin{cases} \lambda_3 e_3^{\alpha} \geqslant e_4^{\alpha} & and \quad e_4 \leqslant e_3\\ \lambda_3 e_3^{\alpha} \geqslant e_4^{\alpha} & and \quad e_4 \geqslant e_3\\ \lambda_3 e_3^{\alpha} \leqslant e_4^{\alpha} & and \quad e_4 \geqslant e_3\\ \lambda_3 e_3^{\alpha} \leqslant e_4^{\alpha} & and \quad e_4 \leqslant e_3 \end{cases}$

Since $\lambda_3 > 1$, the fourth case is impossible.

- (1) case 1: $\lambda_3 e_3^{\alpha} \ge e_4^{\alpha}$ and $e_4 \le e_3$ Player 1 $max \left(1 - \frac{1}{2}\frac{e_4^{\alpha}}{\lambda_3 e_3^{\alpha}}\right)\widetilde{v}_3 - ce_3$ Player 2 $max \quad \frac{1}{2}(\frac{e_4}{e_3})^{\alpha}v_4 - ce_4$ F.o.c
 - $[e_3] \quad \frac{\alpha}{2\lambda_3} \frac{e_4^{\alpha}}{e_3^{\alpha+1}} \widetilde{v}_3 c = 0$
 - $[e_4] \quad \frac{\alpha}{2} \frac{e_4^{\alpha-1}}{e_3^{\alpha}} v_4 c = 0$

S.o.c

$$\begin{split} & [e_3] \quad \frac{\alpha}{2\lambda_3}(-\alpha-1)\frac{e_4^{\alpha}}{e_3^{\alpha+2}}\widetilde{v}_3 < 0 \\ & [e_4] \quad \frac{\alpha}{2}(\alpha-1)\frac{e_4^{\alpha-2}}{e_3^{\alpha}}v_4 < 0 \end{split}$$

Solve the two F.O.C , we get

$$e_{3} = \frac{\alpha}{2c} \lambda_{3}^{\alpha-1} (\widetilde{v}_{3})^{1-\alpha} (v_{4})^{\alpha}$$
$$e_{4} = \frac{\alpha}{2c} \lambda_{3}^{\alpha} (\widetilde{v}_{3})^{-\alpha} (v_{4})^{\alpha+1}$$
$$\frac{e_{4}}{e_{3}} = \lambda_{3} \frac{v_{4}}{\widetilde{v}_{3}}$$

$$p_{43}^{s} = \frac{1}{2} \left(\frac{e_4}{e_3}\right)^{\alpha}$$
$$= \frac{1}{2} \left(\frac{\lambda_3 v_4}{\widetilde{v}_3}\right)^{\alpha}$$

Check the conditions $\lambda_3 e_3^{\alpha} \ge e_4^{\alpha}$ and $e_3 \ge e_4$:

As long as $e_3 \ge e_4$ is satisfied, $\lambda_3 e_3^{\alpha} \ge e_4^{\alpha}$ is satisfied. So we only have to check $e_3 \ge e_4$.

$$\begin{split} e_{4} &\leqslant e_{3} \Longleftrightarrow \frac{e_{4}}{e_{3}} \leqslant 1 \\ &\iff \frac{1}{2} \left(\frac{e_{4}}{e_{3}}\right)^{\alpha} \leqslant \frac{1}{2} \\ &\iff p_{43}^{s} \leqslant \frac{1}{2} \\ &\iff \frac{v_{4}}{\lambda_{3}^{-1} \widetilde{v}_{3}} \leqslant 1 \\ &\iff \frac{\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{12}^{s} + p_{12}^{s} \lambda_{3}^{-\frac{\alpha}{2\alpha + 1}}\right) \\ &\iff \frac{\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \left(1 - p_{12}^{s} + p_{12}^{s} \lambda_{1}^{-\frac{\alpha}{2\alpha + 1}}\right) \lambda_{3}^{-\frac{\alpha + 1}{2\alpha + 1}} \right] \leqslant 1 \end{split}$$

Let

$$f(p_{12}^s) = \lambda_3^{-1} \left[\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \left(1 - p_{12}^s + p_{12}^s \lambda_1^{-\frac{\alpha}{2\alpha+1}} \right) \lambda_3^{-\frac{\alpha+1}{2\alpha+1}} \right] \\ - \left[\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \left(1 - p_{12}^s + p_{12}^s \lambda_1^{-\frac{\alpha}{2\alpha+1}} \right) \right]$$

Rearrange the terms we can get

$$f(p_{12}^s) = \lambda_3^{-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_3^{-\frac{\alpha+1}{2\alpha+1}} \right) - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{1 + \alpha}{2} \left(\lambda_1^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_3^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) p_{12}^s$$

Since $\left(\lambda_1^{-\frac{\alpha}{2\alpha+1}}-1\right)\left(1-\lambda_3^{-1-\frac{\alpha+1}{2\alpha+1}}\right)<0$ and $p_{12}^s\in[0,1], f(p_{12}^s)$ reaches minimum at $p_{12}^s=1$. Thus, $e_4\leqslant e_3$ is always satisfied as long as $f(p_{12}^s=1)\geqslant 0$.

$$f(p_{12}^s = 1) = \lambda_3^{-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \lambda_3^{-\frac{\alpha+1}{2\alpha+1}} \right) - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1+\alpha}{2} \right) + \frac{1+\alpha}{2} \left(\lambda_1^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_3^{-1-\frac{\alpha+1}{2\alpha+1}} \right)$$

$$\begin{split} \lambda_{3}^{-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right) &- \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right) \\ &+ \frac{1 + \alpha}{2} \left(\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_{3}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) \geqslant 0 \\ \Leftrightarrow \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right) - \lambda_{3} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right) \\ &+ \lambda_{3} \frac{1 + \alpha}{2} \left(\lambda_{1}^{-\frac{\alpha}{2\alpha+1}} - 1 \right) \left(1 - \lambda_{3}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) \geqslant 0 \\ \Leftrightarrow \frac{1 + \alpha}{2} \left[\lambda_{3} \left(1 - \left(1 - \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \right) \left(1 - \lambda_{3}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) \right) - \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right] \geqslant (\lambda_{3} - 1) \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \\ \Leftrightarrow \frac{1}{\lambda_{3} - 1} \left[\lambda_{3} \left(\lambda_{3}^{-1 - \frac{\alpha+1}{2\alpha+1}} + \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} - \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \lambda_{3}^{-1 - \frac{\alpha+1}{2\alpha+1}} \right) - \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right] \geqslant \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \\ \Leftrightarrow \frac{1}{\lambda_{3} - 1} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\lambda_{3} - \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} \right) \geqslant \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha} \end{aligned}$$

To ensure $e_3 > e_4$ and $p_{34}^s > p_{43}^s$, we need $\frac{u(w_1)}{u(w_1)-u(w_2)} \frac{2}{1+\alpha} < \frac{1}{\lambda_3-1} \lambda_1^{-\frac{\alpha}{2\alpha+1}} \left(\lambda_3 - \lambda_3^{-\frac{\alpha+1}{2\alpha+1}}\right)$. Similar to the equilibrium in the semifinal between player 1 and player 2, we also need $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$.

(2) case 2:
$$\lambda_3 e_3^{\alpha} \ge e_4^{\alpha}$$
 and $e_4 \ge e_3$

Player 1 $max \left(1 - \frac{1}{2}\frac{e_4^{\alpha}}{\lambda_3 e_3^{\alpha}}\right)\widetilde{v}_3 - ce_3$ Player 2 $max \left(1 - \frac{1}{2}\frac{e_3^{\alpha}}{e_4^{\alpha}}\right)v_4 - ce_4$

F.o.c

$$[e_3] \quad \frac{\alpha}{2\lambda_3} \frac{e_4^{\alpha}}{e_3^{\alpha+1}} \widetilde{v}_3 - c = 0$$
$$[e_4] \quad \frac{\alpha}{2} \frac{e_3^{\alpha}}{e_4^{\alpha+1}} v_4 - c = 0$$

S.o.c

$$[e_3] \quad \frac{\alpha}{2\lambda_3}(-\alpha-1)\frac{e_4^{\alpha}}{e_3^{\alpha+2}}\widetilde{v}_3 < 0$$
$$[e_4] \quad \frac{\alpha}{2}(-\alpha-1)\frac{e_3^{\alpha}}{e_3^{\alpha+2}}v_4 < 0$$

$$\begin{bmatrix} e_4 \end{bmatrix} \quad \frac{1}{2} (-\alpha - 1) \frac{1}{e_4^{\alpha+2}} v_4 < \frac{1}{2} v_4 < \frac{1}{2}$$

Solve F.O.C , we get

$$e_{3} = \frac{\alpha}{2c} \lambda_{3}^{-\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_{3})^{\frac{\alpha+1}{2\alpha+1}} (v_{4})^{\frac{\alpha}{2\alpha+1}}$$
$$e_{4} = \frac{\alpha}{2c} \lambda_{3}^{-\frac{\alpha}{2\alpha+1}} (\widetilde{v}_{3})^{\frac{\alpha}{2\alpha+1}} (v_{4})^{\frac{\alpha+1}{2\alpha+1}}$$
$$\frac{e_{4}}{e_{3}} = \lambda_{3}^{\frac{1}{2\alpha+1}} (\widetilde{v}_{3})^{-\frac{1}{2\alpha+1}} (v_{4})^{\frac{1}{2\alpha+1}}$$

Check the conditions $\lambda_3 e_3^{\alpha} \ge e_4^{\alpha}$ and $e_4 \ge e_3$:

(1) $\lambda_3 e_3^{\alpha} \ge e_4^{\alpha}$

$$\begin{split} \lambda_3 e_3^{\alpha} \geqslant e_4^{\alpha} & \Longleftrightarrow \frac{\lambda_3 e_3^{\alpha}}{e_4^{\alpha}} \geqslant 1 \\ & \Longleftrightarrow \lambda_3^{\frac{\alpha+1}{2\alpha+1}} (\widetilde{v}_3)^{\frac{\alpha}{2\alpha+1}} (v_4)^{-\frac{\alpha}{2\alpha+1}} \geqslant 1 \end{split}$$

Since $\lambda_3 > 1$ and $\tilde{v}_3 > v_4$, $\lambda_3 e_3^{\alpha} > e_4^{\alpha}$ is always satisfied. (2) $e_4 \ge e_3$

 $e_2 \ge e_1$ is always satisfied as long as $f(p_{12}^s = 0) \le 0$.

$$f(p_{12}^s = 0) = \lambda_3^{-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_3^{-\frac{\alpha+1}{2\alpha+1}} \right) - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)$$

$$f(p_{12}^{s} = 0) \leqslant 0 \iff \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \left(\lambda_{3}^{-1} - 1\right) + \frac{1 + \alpha}{2} \left(1 - \lambda_{3}^{-1 - \frac{\alpha + 1}{2\alpha + 1}}\right) \leqslant 0$$
$$\iff \frac{1 + \alpha}{2} \left(1 - \lambda_{3}^{-1 - \frac{\alpha + 1}{2\alpha + 1}}\right) \leqslant \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \left(1 - \lambda_{3}^{-1}\right)$$
$$\iff \frac{1 - \lambda_{3}^{-1 - \frac{\alpha + 1}{2\alpha + 1}}}{1 - \lambda_{3}^{-1}} \leqslant \frac{u(w_{1})}{u(w_{1}) - u(w_{2})} \frac{2}{1 + \alpha}$$

(3) case 3: $\lambda_3 e_3^{\alpha} \leqslant e_4^{\alpha}$ and $e_4 \geqslant e_3$

Player 1
$$max \quad \frac{1}{2} \frac{\lambda_3 e_3^{\alpha}}{e_4^{\alpha}} \widetilde{v}_3 - ce_3$$

Player 2 $max \quad \left(1 - \frac{1}{2} \left(\frac{e_3}{e_4}\right)^{\alpha}\right) v_4 - ce_4$
F.o.c

$$\alpha \lambda_2 e_2^{\alpha-1} \sim$$

$$[e_3] \quad \frac{\alpha\lambda_3}{2} \frac{e_3^{\alpha-1}}{e_4^{\alpha}} \widetilde{v}_3 - c = 0$$

$$[e_4] \quad \frac{\alpha}{2} \frac{e_3}{e_4^{\alpha+1}} v_4 - c = 0$$

divide the two F.O.C , we get

$$\frac{e_4}{e_3} = \frac{v_4}{\lambda_3 \widetilde{v}_3} < 1$$

which contradicts the condition that $e_4 \ge e_3$.

Proof of Proposition 9

1. Equilibrium expected utility of overconfident player 1 in the semifinal with rational player 2.

The equilibrium continuation value of overconfident player 1 is

$$v_{1} = p_{34}^{f} E^{f}(U_{13}) + p_{43}^{f} E^{f}(U_{14})$$

= $E^{f}(U_{13})$
= $\frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} [u(w_{1}) - u(w_{2})] - \frac{\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} [u(w_{1}) - u(w_{2})] + u(w_{2})$
= $\left(\frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} - \frac{\alpha}{2} \lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}} + \frac{u(w_{2})}{u(w_{1}) - u(w_{2})}\right) [u(w_{1}) - u(w_{2})]$

and his equilibrium expected utility in the semifinal with rational player 2 when $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$ and $\lambda < \hat{\lambda}$ is

$$\begin{split} E^{s}(U_{12}) &= p_{12}^{s} v_{1} - c e_{1}^{s} \\ &= \left[1 - \frac{1}{2} \lambda_{1}^{\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{-\alpha} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right)^{\alpha} \right] \\ &\quad \left(\frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha + 1}} - \frac{\alpha}{2} \lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}} + \frac{u(w_{2})}{u(w_{1}) - u(w_{2})} \right) [u(w_{1}) - u(w_{2})] \\ &\quad - \frac{\alpha}{2} \lambda_{1}^{\alpha - 1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \lambda_{1}^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{1 - \alpha} \\ &\quad \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1 + \alpha}{2} \right)^{\alpha} [u(w_{1}) - u(w_{2})] \end{split}$$

The benchmark equilibrium expected utility of the semifinal is

$$\overline{E}^{s}(U) = \frac{1-\alpha}{2}\overline{v} = \frac{1-\alpha}{2}\left(\frac{u(w_{1})}{u(w_{1})-u(w_{2})} - \frac{1+\alpha}{2}\right)\left[u(w_{1})-u(w_{2})\right]$$

Let

$$\begin{split} f(\lambda_1) &= \frac{E^s(U_{12}) - \overline{E}^s(U)}{[u(w_1) - u(w_2)]} \\ &= \left[1 - \frac{1}{2} \lambda_1^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \right] \\ &\left(\frac{1}{2} \lambda_1^{-\frac{\alpha}{2\alpha + 1}} - \frac{\alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} + \frac{u(w_2)}{u(w_1) - u(w_2)} \right) \\ &- \frac{\alpha}{2} \lambda_1^{\alpha - 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{1 - \alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \\ &- \frac{1 - \alpha}{2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \end{split}$$

We can easily get that $f(\lambda_1 = 1) = 0$.

$$\begin{split} f'(\lambda_1) &= \left[1 - \frac{1}{2} \lambda_1^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \right] \\ &\quad \left(-\frac{1}{2} \frac{\alpha}{2\alpha + 1} \lambda_1^{-\frac{\alpha}{2\alpha+1} - 1} + \frac{\alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \lambda_1^{-\frac{\alpha+1}{2\alpha+1} - 1} \right) \\ &\quad + \left[-\frac{1}{2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \left(\alpha \lambda_1^{\alpha-1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \right. \\ &\quad + \lambda_1^{\alpha}(-\alpha) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha-1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \lambda_1^{-\frac{\alpha+1}{2\alpha+1} - 1} \right) \right] \\ &\quad \left(\frac{1}{2} \lambda_1^{-\frac{\alpha}{2\alpha+1}} - \frac{\alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} + \frac{u(w_2)}{u(w_1) - u(w_2)} \right) \\ &\quad - \frac{\alpha}{2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha} \left[(\alpha - 1) \lambda_1^{\alpha-2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{1-\alpha} \\ &\quad + \lambda_1^{\alpha-1}(1 - \alpha) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha+1}{2\alpha+1}} \right)^{-\alpha} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \lambda_1^{-\frac{\alpha+1}{2\alpha+1} - 1} \right] \end{split}$$

$$\begin{split} f'(\lambda_1 = 1) &= \frac{1}{2} \frac{\alpha}{2\alpha + 1} + \\ & \left[-\frac{\alpha}{2} + \frac{\alpha}{2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{-1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \\ & - \frac{\alpha}{2} (\alpha - 1) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) - \frac{\alpha}{2} (1 - \alpha) \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \\ & = \frac{1}{2} \frac{\alpha}{2\alpha + 1} - \frac{\alpha}{2} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{\alpha}{2} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \\ & - \frac{\alpha}{2} (\alpha - 1) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) - \frac{\alpha}{2} (1 - \alpha) \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \\ & = \frac{\alpha}{2} \left[\frac{1}{2\alpha + 1} - \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \alpha \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \alpha \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha^2}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \alpha \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha^2}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha^2}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha^2}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha^2}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ & = \frac{\alpha^2}{2} \left[\frac{1}{2\alpha + 1} - \alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right) \right] \\ & = \frac{\alpha^2}{2} \left[\frac{1}{2\alpha + 1} - \frac{1 + \alpha}{2} \left(1 + \frac{\alpha + 1}{2\alpha + 1} \right) - 1 - \frac{u(w_2)}{u(w_1) - u(w_2)} \right] \\ & = \frac{\alpha^2}{2} \left[\frac{\alpha(3\alpha + 1) + 1}{2(2\alpha + 1)} - \frac{u(w_2)}{u(w_1) - u(w_2)} \right] \\ & = \frac{\alpha^2(3\alpha + 1) + 1}{2(2\alpha + 1)} - \frac{u(w_2)}{u(w_1) - u(w_2)} \right] \\ & = \frac{\alpha^2(3\alpha + 1) + 1}{2(2\alpha + 1)} - \frac{u(w_2)}{u(w_1) - u(w_2)} \\ & = \frac{\alpha^2(3\alpha + 1) + 1}{2(2\alpha + 1)} - \frac{u(w_2)}{u(w_1) - u(w_2)} \right] \\ & = \frac{\alpha^2(3\alpha + 1) + 1}{2(2\alpha + 1)} - \frac{1}{2(2\alpha + 1)} - \frac{1}{2(2\alpha + 1)} - \frac{1}{2(2\alpha$$

2. Equilibrium expected utility of rational player 2 in the semifinal with overconfident player 1.

(1) When
$$\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$$
 and $\lambda_1 < \hat{\lambda}$

$$E^s(U_{21}) = p_{21}^s v_2 - ce_2^s$$

$$= \frac{1}{2} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{\alpha} v_2 - c\frac{\alpha}{2c} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{\alpha+1}$$

$$= \frac{1}{2} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{1+\alpha} - \frac{\alpha}{2} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{\alpha+1}$$

$$= \frac{1-\alpha}{2} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{1+\alpha}$$

$$= \frac{1-\alpha}{2} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (\overline{v})^{1+\alpha}$$

$$= \frac{1-\alpha}{2} \left(\frac{\overline{v}}{\lambda_1^{-1} \widetilde{v}_1} \right)^{\alpha} \overline{v}$$

$$< \frac{1-\alpha}{2} \overline{v} = \overline{E}^s(U)$$

(2) When either $\frac{u(w_1)-u(w_2)}{u(w_2)} \leq \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$ or $\lambda_1 \geq \hat{\lambda}$

$$\begin{split} E^{s}(U_{21}) &= p_{21}^{s} v_{2} - ce_{2}^{s} \\ &= \left[1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2} \right)^{-\frac{\alpha}{2\alpha+1}} \right] v_{2} - c\frac{\alpha}{2c} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2} \right)^{\frac{\alpha+1}{2\alpha+1}} \\ &= \left[1 - \frac{1}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2} \right)^{-\frac{\alpha}{2\alpha+1}} \right] v_{2} - \frac{\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2} \right)^{\frac{\alpha+1}{2\alpha+1}} \\ &= v_{2} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(v_{2} \right)^{\frac{\alpha+1}{2\alpha+1}} \\ &= \overline{v} - \frac{1+\alpha}{2} \lambda_{1}^{-\frac{\alpha}{2\alpha+1}} \left(\widetilde{v}_{1} \right)^{\frac{\alpha}{2\alpha+1}} \left(\overline{v} \right)^{\frac{\alpha+1}{2\alpha+1}} \\ &= \left[1 - \frac{1+\alpha}{2} \left(\frac{\lambda_{1}^{-1} \widetilde{v}_{1}}{\overline{v}} \right)^{\frac{\alpha}{2\alpha+1}} \right] \overline{v} \\ &> \frac{1-\alpha}{2} \overline{v} = \overline{E}^{s}(U) \end{split}$$

3. Equilibrium expected utility of rational player 3 (4) in the semifinal with rational player 4 (3).

$$E^{s}(U_{34}) = E^{s}(U_{43}) = \frac{1-\alpha}{2}v_{3} = \frac{1-\alpha}{2}v_{4} > \frac{1-\alpha}{2}\overline{v} = \overline{E}^{s}(U).$$

Proof of Proposition 10

Part (i) follows directly from Propositions 1 and 2. Let's then prove part (ii). We know that the equilibrium efforts in the semifinal between the two rational players are higher than the benchmark, thus the equilibrium aggregate effort in the semifinals stage is higher than that of the benchmark if the equilibrium total effort of the semifinal with an overconfident and a rational player is higher than that of the benchmark. We also know from Proposition 3 that if $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$ and $\lambda_1 < \hat{\lambda}$, then total effort in the

semifinal with an overconfident and a rational player is given by

$$e_1^s + e_2^s = \frac{\alpha}{2c} \lambda_1^{\alpha - 1} (\widetilde{v}_1)^{1 - \alpha} (v_2)^{\alpha} + \frac{\alpha}{2c} \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (v_2)^{\alpha + 1}$$
$$= \frac{\alpha}{2c} \overline{v} \left[\lambda_1^{\alpha - 1} (\widetilde{v}_1)^{1 - \alpha} (\overline{v})^{\alpha - 1} + \lambda_1^{\alpha} (\widetilde{v}_1)^{-\alpha} (\overline{v})^{\alpha} \right]$$

Hence, we have

$$\frac{e_1^s + e_2^s}{2\overline{e}^s} = \frac{\frac{\alpha}{2c}\overline{v} \left[\lambda_1^{\alpha - 1}(\widetilde{v}_1)^{1 - \alpha}(\overline{v})^{\alpha - 1} + \lambda_1^{\alpha}(\widetilde{v}_1)^{-\alpha}(\overline{v})^{\alpha}\right]}{2\frac{\alpha}{2c}\overline{v}}$$
$$= \frac{1}{2} \left[\lambda_1^{\alpha - 1}(\widetilde{v}_1)^{1 - \alpha}(\overline{v})^{\alpha - 1} + \lambda_1^{\alpha}(\widetilde{v}_1)^{-\alpha}(\overline{v})^{\alpha}\right]$$

Let $f(\lambda) = \lambda_1^{\alpha-1}(\widetilde{v}_1)^{1-\alpha}(\overline{v})^{\alpha-1} + \lambda_1^{\alpha}(\widetilde{v}_1)^{-\alpha}(\overline{v})^{\alpha}$,

$$f(\lambda) = \lambda_1^{\alpha - 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{1 - \alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha - 1} + \lambda_1^{\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \lambda_1^{-\frac{\alpha + 1}{2\alpha + 1}} \right)^{-\alpha} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2} \right)^{\alpha}$$

We can easily get that $f(\lambda_1 = 1) = 2$.

$$f'(\lambda_{1}) = \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{\alpha-1} \left[(\alpha-1)\lambda_{1}^{\alpha-2} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{1-\alpha} + \lambda_{1}^{\alpha-1}(1-\alpha) \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} \frac{1+\alpha}{2} \frac{\alpha+1}{2\alpha+1}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}-1} \right] + \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\right)^{\alpha} \left[\alpha\lambda_{1}^{\alpha-1} \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha} + \lambda_{1}^{\alpha}(-\alpha) \left(\frac{u(w_{1})}{u(w_{1}) - u(w_{2})} - \frac{1+\alpha}{2}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}}\right)^{-\alpha-1} \frac{1+\alpha}{2} \frac{\alpha+1}{2\alpha+1}\lambda_{1}^{-\frac{\alpha+1}{2\alpha+1}-1} \right]$$

$$\begin{aligned} f'(\lambda_1 = 1) &= \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{\alpha - 1} \left[(\alpha - 1) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{1 - \alpha} \right. \\ &+ (1 - \alpha) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-\alpha} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ &+ \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{\alpha} \left[\alpha \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-\alpha} \right. \\ &+ (-\alpha) \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-\alpha - 1} \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \right] \\ &= (\alpha - 1) + (1 - \alpha) \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-1} + \alpha \\ &- \alpha \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-1} \\ &= (2\alpha - 1) + (1 - 2\alpha) \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-1} \\ &= (1 - 2\alpha) \left[-1 + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-1} \right] \\ &- 1 + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-1} \right] \\ -1 + \frac{1 + \alpha}{2} \frac{\alpha + 1}{2\alpha + 1} \left(\frac{u(w_1)}{u(w_1) - u(w_2)} - \frac{1 + \alpha}{2}\right)^{-1} > 0 \text{ is equivalent to } \frac{u(w_1) - u(w_2)}{u(w_2)} > \frac{2(2\alpha + 1)}{\alpha(3\alpha + 1)}. \\ \text{Thus we have} \\ \text{If } \alpha < \frac{1}{2}, \text{ then } f'(\lambda_1 = 1) > 0. \\ \text{If } \alpha > \frac{1}{2}, \text{ then } f'(\lambda_1 = 1) < 0. \\ \text{Therefore, when } \frac{u(w_1) - u(w_2)}{u(w_1) - u(w_2)} > \frac{2(2\alpha + 1)}{\alpha(\alpha + 1)} \text{ and } \alpha < \frac{1}{\alpha}, \text{ there exist } \lambda_1 \in (1, \hat{\lambda}) \text{ such then } \end{array}$$

Therefore, when $\frac{u(w_1)-u(w_2)}{u(w_2)} > \frac{2(2\alpha+1)}{\alpha(3\alpha+1)}$ and $\alpha < \frac{1}{2}$, there exist $\lambda_1 \in (1, \hat{\lambda})$ such that $e_1^s + e_2^s > 2\overline{e}^s$ which implies

$$\sum_{i=1}^{4} e_i^s > 4\overline{e}^s$$

Proof of Proposition 11

We know from Proposition 6 that the equilibrium effort in a final between two equally overconfident players is equal to

$$e^f = \frac{\alpha}{2c\lambda} \left[u(w_1) - u(w_2) \right].$$

The perceived expected utility of the final is then given by

$$\widetilde{E}(U^f) = \frac{1}{2}u(w_1) + \frac{1}{2}u(w_2) - c\frac{\alpha}{2c\lambda}\left[u(w_1) - u(w_2)\right]$$
$$= \frac{\lambda - \alpha}{2\lambda}u(w_1) + \frac{\lambda + \alpha}{2\lambda}u(w_2).$$

The equilibrium effort in a semifinal between two equally overconfident players is equal to

$$e^{s} = \frac{\alpha}{2c\lambda} \widetilde{E}(U^{f})$$
$$= \frac{\alpha}{2c\lambda} \left[\frac{\lambda - \alpha}{2\lambda} u(w_{1}) + \frac{\lambda + \alpha}{2\lambda} u(w_{2}) \right]$$

Hence, total effort amounts to

$$\epsilon = 2\frac{\alpha}{2c\lambda} \left[u(w_1) - u(w_2) \right] + 4\frac{\alpha}{2c\lambda} \left[\frac{\lambda - \alpha}{2\lambda} u(w_1) + \frac{\lambda + \alpha}{2\lambda} u(w_2) \right]$$
$$= \frac{\alpha}{c\lambda} \left[\left(2 - \frac{\alpha}{\lambda} \right) u(w_1) + \frac{\alpha}{\lambda} u(w_2) \right].$$

The problem of the contest designer is to maximize total effort subject to $w_1 + w_2 = W$. Since $\alpha \in (0, 1]$ and $\lambda \ge 1$ the optimal solution to this problem when players are risk neutral is to set $w_1 = W$ and $w_2 = 0$, that is, to allocate all the prize money to the winner of the final. When players are risk averse this is no longer the case. Substituting the constraint $w_2 = W - w_1$ into the objective function we have the unconstrained problem

$$max \quad \frac{\alpha}{c\lambda} \left[\left(2 - \frac{\alpha}{\lambda} \right) u(w_1) + \frac{\alpha}{\lambda} u(W - w_1) \right]$$

The first-order condition is

$$\left(2-\frac{\alpha}{\lambda}\right)u'(w_1)-\frac{\alpha}{\lambda}u'(W-w_1)=0.$$

Rearranging the first-order condition we have

$$\frac{u'(w_1)}{u'(W-w_1)} = \frac{\frac{\alpha}{\lambda}}{2-\frac{\alpha}{\lambda}}.$$
(18)

Hence, when players are risk averse, the optimal prize structure involves multiple prizes with the winner of the final receiving most of the prize money and a smaller part being assigned to the runner-up. Since the right-hand side of (18) is decreasing in λ , it follows that an increase in overconfidence raises the share of the prize money allocated to the winner of the final.